LOCALIZATION THEOREMS AND PERTURBATION ANALYSIS ON QUATERNIONIC EIGENVALUE PROBLEMS

Ph.D. Thesis

by ISTKHAR ALI



DISCIPLINE OF MATHEMATICS INDIAN INSTITUTE OF TECHNOLOGY INDORE

AUGUST 2016

LOCALIZATION THEOREMS AND PERTURBATION ANALYSIS ON QUATERNIONIC EIGENVALUE PROBLEMS

A THESIS

Submitted in partial fulfillment of the requirements for the award of the degree of DOCTOR OF PHILOSOPHY

> by ISTKHAR ALI



DISCIPLINE OF MATHEMATICS INDIAN INSTITUTE OF TECHNOLOGY INDORE

AUGUST 2016



INDIAN INSTITUTE OF TECHNOLOGY INDORE CANDIDATE'S DECLARATION

I hereby certify that the work which is being presented in the thesis entitled "LOCALIZATION THEOREMS AND PERTURBATION ANALYSIS ON QUATERNIONIC EIGENVALUE PROBLEMS" in the partial fulfillment of the requirements for the award of the degree of DOCTOR OF PHILOSOPHY and submitted in the DISCIPLINE OF MATHEMATICS, Indian Institute of Technology Indore, is an authentic record of my own work carried out during the time period from January 2011 to August 2016 under the supervision of Dr. Sk. Safique Ahmad, Assistant Professor, Indian Institute of Technology Indore.

The matter presented in this thesis has not been submitted by me for the award of any other degree of this or any other institute.

Signature of the student with date

(ISTKHAR ALI)

This is to certify that the above statement made by the candidate is correct to the best of my knowledge.

Signature of Thesis Supervisor with date

(Dr. SK. SAFIQUE AHMAD)

ISTKHAR ALI has successfully given his Ph.D. Oral Examination held on November 20, 2017.

Sign. of Chairperson (OEB)Sign. of External ExaminerSign. of Thesis SupervisorDate:Date:Date:

Sign. of PSPC Member #1Sign. of PSPC Member #2Sign. of Convener, DPGCDate:Date:Date:

Sign. of Head of Discipline

Date:

ACKNOWLEDGEMENTS

I consider myself as a fortuitous research scholar to have Dr. Sk. Safique Ahmad as my thesis supervisor. I am highly indebted to him for introducing me to the field of Quaternion Linear Algebra and enabling me to pursue research in it. Long fruitful discussions, criticism for progression, constant motivations at the moments when I desperately needed them, bringing me back to the track whenever I was lost, communicating research ideas over phone whenever physical meetings were not possible, and sharing laughter moments are some of the reasons that I am able to reach to a stage where I am concluding my Doctoral Thesis. With his professional advice and faith in me, I was able to conquer each and every challenge that came in the way.

I owe a debt of gratitude to my PSPC members Dr. Antony Vijesh and Dr. Rajneesh Misra for their time, effort, expertise, and critical comments at various stages in general and during my research progress seminars in particular. I am highly grateful to all the faculty members of Discipline of Mathematics of IIT Indore for being so generous, supportive and helpful in forming a rigorous and meaningful thesis. I especially want to acknowledge Dr. Md. Aquil Khan, Dr. Niraj Shukla and Dr. Swadesh Sahoo for all the support and encouragement that I received from them from time to time.

Thanks are also due to Prof. Dr. Ivan Slapniar from the University of Split, Croatia, who had visited IIT Indore, for their fruitful discussions and suggestions.

I would like to convey my gratitude to Dr. Shaikh Mobin for constantly motivating me to accomplish this journey as early as possible.

My heartfelt thanks also goes to all my colleagues at the Discipline of Mathematics who wholeheartedly and sincerely helped me whenever I needed them and for the discussions that widened the horizons of my research. My special thanks goes to Mr. Manas, Mr. Nitin, Mrs. Linia, Ms. Sarita, Dr. Navneet, Mr. Prince and Ms. Rupsha for all their support, encouragement, discussions and mood refreshing whenever I was down. I am fortunate for being surrounded by many wonderful friends namely Mr. Akbar, Mr. Sajad, Dr. Ajay, Dr. Sunil, Dr. Aditya, Mr. Shailendra, Mr. Sadaf and Mr. Nasir who always motivated and supported me in every manner they could. I am going to miss all the fun moments that we shared at the dining table, during tea time and during our walk after dinner. This journey would not have been easy if I was not accompanied by such wonderful and fun loving people.

Doctoral research is such a journey where one needs support not only in the campus but off the campus as well. There are many people who contributed, in one way or the other, in conquering this noble journey. My sincere thanks are due to Dr. Dhananjay, Mr. Suraj, Mr. Salahuddin, Mr. Safdar, Mr. Shahabuddin, Dr. Enamullah, Mr. Abdul Samad, Mr. Attaullah, Mr. Shameem, Mr. Suryakant, Mr. Jilani, Mr. Vikas, Mr. Sachin, and Dr. Kamal.

I extend my thanks to IIT Indore for providing me all the facilities that I needed and to Council of Scientific & Industrial Research (CSIR), Govt. of India, for providing me financial assistance initially through their Junior Research Fellowship and later through their Senior Research Fellowship under grant no. 09/1022 (0006) / 2011-EMR-I.

I am highly indebted to my family members especially to my dear father (Mr. Ashik Ali Ansari) and mother (Mrs. Nasirun Nisha); my uncle (Mr. Qasim Ali) and aunty (Mrs. Azizun Nisha); my lovely brothers (Mr. Julphekar Ali and Shamshad Ali); caring sisters (Mrs. Noor Jahan, Mrs. Najma, Mrs. Asma and their respective husbands Mr. Eliyash Ansari, Mr. Hasmullah Ansari, Mr. Habibullah Ansari). I feel shortage of words for thanking my family for all their love and having faith in me. It is all because of your hard work, inspiration, emotional and moral support that I could become what I am today. I hope you are as proud of me as I am proud of you.

I am deeply grateful to my dear Gulsheen Naz who was there for me in every up and down of this magnificent journey. My dear, I owe a lot to you.

Finally, I can't go away without thanking the most merciful and the most beneficent "Almighty Allah" without whose help I would not have been able to accomplish this dream.

Istkhar Ali

List of Publications

- 1. Ahmad S.S., Ali I. (2016), Bounds for eigenvalues of matrix polynomials over quaternion division algebra, Adv. Appl. Clifford Algebras, 26:1096-1125.
- 2. Ahmad S.S., Ali I., Localization theorems of matrices and bounds for the zeros of polynomials over a quaternion division algebra, Filomat, to Appear.
- Ahmad S.S., Ali I., Localization theorems for eigenvalues of quaternionic matrices, (arXiv:1502.08014v2 [math.RA]).
- 4. Ahmad S.S., Ali I., Location for zeros of polynomials and eigenvalues of matrix pencils over a quaternion division algebra, Communicated.
- 5. Ahmad S.S., Ali I., Inclusion regions on zeros of polynomials over a quaternion division algebra, Preprint.
- 6. Ahmad S.S., Ali I., Perturbation bounds of right eigenvalues of quaternionic matrices and their applications, Preprint.
- 7. Ahmad S.S., Ali I., Modified Ostrowski balls for matrices over a quaternion division algebra, Preprint.

ABSTRACT

Quaternions are extensively used in programming video games, computer graphics, control theory and quantum physics, etc. A solution of a quaternionic linear differential equation with constant coefficients can be presented in terms of right eigenvalues as well as their corresponding eigenvectors of the associated quaternionic matrix. The study of a quaternionic linear differential equation with constant coefficients is based on finding the zeros of its corresponding quaternionic polynomial. In contrast to the complex case, the location of left eigenvalues of a quaternionic matrix plays an important role in the characterization of zeros of quaternionic polynomials. The stability of linear difference/differential equations with quaternionic matrix coefficients is based on the location of right eigenvalues of their corresponding block companion matrices.

This thesis mainly deals with localization theorems for the left and right eigenvalues of a quaternionic matrix and their applications for finding bounds/location of zeros of quaternionic polynomials. Bounds for the left and right eigenvalues of quaternionic matrix polynomials are derived. In the proposed research work we also discuss about perturbation bounds for right eigenvalues/generalized right eigenvalues of a quaternionic matrix/quaternionic matrix pencil. The entire work of this thesis is divided into seven chapters and has been briefly described below:

Chapter 1 describes preliminaries and basic facts related to the development of our theory.

In Chapter 2, inclusion regions for eigenvalues of a quaternionic matrix are derived and bounds for the zeros of quaternionic polynomials are presented. In this chapter, we study Gerschgorin, Ostrowski, and Brauer type theorems for the left and right eigenvalues of a quaternionic matrix. Thereafter a sufficient condition for the stability of a continuoustime quaternionic system is given.

Chapter 3 presents inclusion regions of zeros of quaternionic polynomials.

Chapter 4 discusses basic properties of regular quaternionic matrix pencils, localization theorems of generalized right eigenvalues of quaternionic matrix pencils, and their applications.

Chapter 5 derives the definitions of the left and right eigenvalues of quaternionic matrix polynomials. Next, we present bounds of left and right eigenvalues of quaternionic matrix polynomials. A sufficient condition for the stability of a discrete-time quaternionic system is given. Furthermore, bounds for the absolute values of the left and right eigenvalues of quaternionic matrix polynomials are devised and illustrated for the matrix p-norm, where $p = 1, 2, \infty$, and F (Frobenius). The above results generalize bounds for the absolute values of the eigenvalues of complex matrix polynomials.

Chapter 6 gives the concept of perturbation bounds for right eigenvalues/generalized right eigenvalues of a quaternionic matrix/quaternionic matrix pencil. In particular, Bauer-Fike type theorems for right eigenvalues/generalized right eigenvalues of a diagonalizable quaternionic matrix/diagonalizable quaternionic matrix pencil are derived. Then, a relative perturbation bound for right eigenvalues of an invertible diagonalizable quaternionic matrix is given. Perturbation bounds of right eigenvalues of a quaternionic matrix and perturbation bounds for the zeros of quaternionic polynomials are presented.

Finally, in Chapter 7, we give conclusions of our research work and future prospect of this work.

TABLE OF CONTENTS

ACKNOWLEDGEMENTS	iii
ABSTRACT	vii
LIST OF FIGURES	xiii
LIST OF TABLES	XV
NOTATION	xvii
Chapter 1 INTRODUCTION	1
1.1 Introduction	1
1.2 Preliminaries	4
1.2.1 Quaternionic matrices	7
1.2.2 Complex adjoint matrix	10
1.2.3 The eigenvalue problem	11
1.2.4 Quaternionic polynomials and their companion matrices	15
1.2.5 The generalized eigenvalue problem	19
1.2.6 The polynomial eigenvalue problem	21
Chapter 2 LOCALIZATION THEOREMS OF MATRICES OVER A	L
QUATERNION DIVISION ALGEBRA	25
2.1 Introduction	25
2.2 Distribution for the left and right eigenvalues of quaternionic	
matrices	26
2.3 Bounds for the zeros of quaternionic polynomials	43
Chapter 3 LOCATION OF ZEROS OF POLYNOMIALS OVER A	
QUATERNION DIVISION ALGEBRA	55
3.1 Introduction	55

matrices	56
3.3 Inclusion regions for right eigenvalues of quaternionic matrices	61
3.4 Inclusion regions for zeros of quaternionic polynomials 6	62
3.4.1 Inclusion regions for zeros of complex polynomials	68
Chapter 4 LOCALIZATION THEOREMS FOR QUATERNIONIC	
MATRIX PENCILS 7	73
4.1 Introduction and preliminaries 7	73
4.2 Location of generalized right eigenvalues 7	74
Chapter 5 BOUNDS FOR EIGENVALUES OF MATRIX	
POLYNOMIALS OVER A QUATERNION DIVISION	
ALGEBRA 8	81
5.1 Introduction and preliminaries	81
5.2 Inclusion regions for the eigenvalues of quaternionic block matrices 8	83
5.3 Quaternionic matrix polynomials 8	87
5.3.1 Right eigenvalues of quaternionic matrix polynomials and	
their bounds 8	88
5.3.2 Left eigenvalues of quaternionic matrix polynomials and their	
bounds 10)5
5.4 Bounds for eigenvalues of complex matrix polynomials 11	10
Chapter 6 PERTURBATION BOUNDS FOR RIGHT EIGENVALUES	
OF QUATERNIONIC MATRICES AND THEIR	
APPLICATIONS 11	15
6.1 Introduction 11	15
6.2 Perturbation analysis on quaternionic matrices 11	16
6.2.1 Bauer-Fike type theorem for the right eigenvalues 11	19
6.2.2 Perturbation bounds for non-diagonalizable quaternionic	
matrices 12	24
6.3 Perturbation bounds for the zeros of quaternionic polynomials 12	28
6.4 Bauer-Fike type theorem for generalized right eigenvalues 12	29

Chapter 7 CONCLUSIONS AND SCOPE FOR FUTURE WORK	137
APPENDIX-A	139
APPENDIX-B	143
BIBLIOGRAPHY	145

LIST OF FIGURES

2.1	Location of complex right eigenvalues of A (Example 2.14) as +.	34
2.2	Location of complex right eigenvalues of A (Example 2.20) as $+$.	39
2.3	Location of complex right eigenvalues of A (Example 2.21) as $+$.	40
3.1	Location of zeros of $p(z)$.	72

LIST OF TABLES

2.1	The zeros of $p_l(z)$ and $p_r(z)$ and their absolute values.	53
2.2	Lower and upper bounds for the zeros of $p_l(z)$ and $p_r(z)$.	54
2.3	Lower and upper bounds for the zeros of $p_l(z)$ and $p_r(z)$.	54
5.1	Lower and upper bounds for right eigenvalues of $\mathbf{L}(\lambda)$.	106
5.2	Lower and upper bounds for eigenvalues of $\mathcal{P}(z)$.	112
5.3	Lower and upper bounds for eigenvalues of $\mathcal{P}(z)$.	113

NOTATION

Symbols

\mathbb{R}	the field of real numbers
\mathbb{C}	the field of complex numbers
\mathbb{C}_{∞}	the extended complex plane $\mathbb{C} \cup \{\infty\}$
\mathbb{C}^+	closed upper complex halfplane
\mathbb{R}^+	the set of positive real numbers
H	the set of real quaternions
$\Im(z)$	the imaginary part of $z \in \mathbb{C}$
$\overline{q} = q^H$	the conjugate of $q \in \mathbb{H}$
q	the modulus of $q \in \mathbb{H}$
$\operatorname{Re}(q)$	the real part of $q \in \mathbb{H}$
$\operatorname{Im}(q)$	the imaginary part of $q \in \mathbb{H}$
$S_{\mathbb{H}^{-}}$	the set $\{q \in \mathbb{H} : \operatorname{Re}(q) < 0\}$
$S_{\mathbb{H}}$	the unit ball $\{q \in \mathbb{H} : q < 1\}$
[q]	the equivalence class of $q \in \mathbb{H}$
\mathcal{K}^n	the collections of all n -column vectors with entries
	in \mathcal{K} , where $\mathcal{K} \in \{\mathbb{R}, \mathbb{C}, \mathbb{H}\}$
x^T	the transpose of $x \in \mathcal{K}^n$
\overline{x}	the conjugate of $x \in \mathcal{K}^n$
x^H	the transpose conjugate of $x \in \mathcal{K}^n$
$M_{m \times n}(\mathcal{K})$	the set of $m \times n$ matrices with entries in \mathcal{K} , abbreviated to $M_n(\mathcal{K})$
	if $m = n$
I_n	the $n \times n$ identity matrix
0_n	the $n \times n$ zero matrix
A^{-1}	the inverse of $A \in M_n(\mathcal{K})$
A^T	the transpose of $A \in M_{m \times n}(\mathcal{K})$

\overline{A}	the conjugate of $A \in M_{m \times n}(\mathcal{K})$
A^H	the conjugate transpose of $A \in M_{m \times n}(\mathcal{K})$
$A \otimes B$	the Kronecker product of $A \in M_n(\mathbb{H})$ with $B \in M_n(\mathbb{H})$
$\operatorname{diag}(A_1,\ldots,A_k)$	block diagonal matrix with the diagonal blocks $A_1, \ldots, A_k \in M_n(\mathcal{K})$,
	abbreviated to $\operatorname{diag}(A_j)_{j=1}^k$
$ A _p$	the norm of $A \in M_n(\mathcal{K})$, where $p = 1, 2, \infty$ and F (Frobenius)
$K_2(A)$	the condition number of $A \in M_n(\mathbb{H})$ with respect to the matrix
	2-norm
Ψ_A	the complex adjoint matrix of $A \in M_n(\mathbb{H})$
$\Lambda(A)$	the set of eigenvalues of $A \in M_n(\mathbb{C})$
$\Lambda_l(A)$	the set of left eigenvalues of $A \in M_n(\mathbb{H})$
$\Lambda_r(A)$	the set of right eigenvalues of $A \in M_n(\mathbb{H})$
$\Lambda_s(A)$	the set of standard right eigenvalues of $A \in M_n(\mathbb{H})$
$\rho_l(A)$	the left spectral radius of $A \in M_n(\mathbb{H})$
$\rho_r(A)$	the right spectral radius of $A \in M_n(\mathbb{H})$
$p_l(z)$	the simple quaternionic polynomial of the form $\sum_{j=0}^{m} q_j z^j$,
	where $z, q_j \in \mathbb{H} \ (0 \le j \le m)$
$p_r(z)$	the simple quaternionic polynomial of the form $\sum_{j=0}^{m} z^{j} q_{j}$,
	where $z, q_j \in \mathbb{H} \ (0 \le j \le m)$
$q_l(z)$	the simple monic reversal quaternionic polynomial of $p_l(z)$
$q_r(z)$	the simple monic reversal quaternionic polynomial of $p_r(z)$
$Z_{\mathbb{H}}(p(z))$	the set of zeros of a simple quaternionic polynomial $p(z)$
$Z_{\mathbb{C}}(p(z))$	the set of complex zeros of a simple quaternionic polynomial $p(z)$
C_{p_l}	the corresponding companion matrix of the simple monic
	polynomial $p_l(z)$
C_{p_r}	the corresponding companion matrix of the simple monic
	polynomial $p_r(z)$
C_{q_l}	the corresponding companion matrix of the simple monic reversal
	polynomial $q_l(z)$
C_{q_r}	the corresponding companion matrix of the simple monic reversal
	polynomial $q_r(z)$

CHAPTER 1

INTRODUCTION

1.1. Introduction

The set of quaternions was discovered by the Irish mathematician Sir William Rowan Hamilton in 1843. It is an extension of the complex field. The set of quaternions is an associative but non-commutative algebra of rank four over \mathbb{R} . Quaternions are extensively used in programming video games, control theory, computer graphics, controlling spacecrafts, signal processing, quantum physics etc. (see, for example, [1,8,24,26,40,58]). Quaternions are used for describing rotations and orientations of objects in 3-dimensional space. For instance, spacecraft altitude-control systems are commanded in terms of quaternions (see, for example, [1,26] and the references therein). A quaternion is expressed by 4 real numbers whereas a 3×3 matrix requires 9 numbers. The composition of two rotations requires 16 multiplications and 12 additions in quaternion representation, but 27 multiplications and 18 additions in matrix representation [39].

The applications of complex eigenvalue problems have been studied in [4,7,10,11,14, 16,51,54,57]. The quaternionic eigenvalue problems occur in many scientific applications (see, for example, [2,3,9,13,18,27,31,32,43,44,52,53,55,56,61] and the references therein). The location of right eigenvalues of a quaternionic matrix plays an important role to find the stability of the linear differential equation with quaternionic matrix coefficients

(1.1)
$$\frac{\mathrm{d}^{\mathrm{m}}}{\mathrm{d}t^{\mathrm{m}}} w(t) + A_{m-1} \frac{\mathrm{d}^{\mathrm{m}-1}}{\mathrm{d}t^{\mathrm{m}-1}} w(t) + A_{m-2} \frac{\mathrm{d}^{\mathrm{m}-2}}{\mathrm{d}t^{\mathrm{m}-2}} w(t) + \dots + A_0 w(t) = 0,$$

and the quaternionic matrix difference equation

(1.2)
$$w(t+m) + A_{m-1}w(t+m-1) + \dots + A_1w(t+1) + A_0w(t) = 0,$$

where $w(t) = (w_1(t), w_2(t), \dots, w_n(t))^T \in \mathbb{H}^n$ with $w_i : \mathbb{R} \to \mathbb{H}$ $(t \in \mathbb{R}, 1 \leq i \leq n)$ (\mathbb{H} and \mathbb{H}^n are defined in Preliminaries 1.2) (see, for example, [43, 44]). The location of left eigenvalues of a quaternionic matrix plays a crucial role to find bounds/location of zeros of quaternionic polynomials. The quaternionic eigenvalue problems and location of left and right eigenvalues of a quaternionic matrix and their applications have been studied extensively (see, for example, [3, 6, 9, 18, 22, 27, 31, 32, 43, 44, 52, 60–63] and the references therein). Computation of zeros of quaternionic polynomials and their bounds have been studied extensively covering the theory as well as the applications (see, for example, [12, 19, 20, 23, 30, 41, 42, 45, 50]). An application of quaternionic quadratic equations is also found in solving homogeneous quaternionic linear second order differential equations with quaternion constant coefficients [28, 29]. The solution of quaternionic differential equation

(1.3)
$$\frac{\mathrm{d}^2}{\mathrm{d}t^2} u(t) - q_1 \frac{\mathrm{d}}{\mathrm{d}t} u(t) - q_0 u(t) = 0,$$

where $u : \mathbb{R} \to \mathbb{H}, q_0, q_1 \in \mathbb{H}$ and $t \in \mathbb{R}$, can be transformed by substituting $u(t) = \exp[zt] (z \in \mathbb{H})$ to the following quaternionic quadratic equation:

(1.4)
$$z^2 - q_1 z - q_0 = 0.$$

Generalizing (1.3), we observe that the problem of finding the solution of *m*-order linear differential equations with quaternion constant coefficients

(1.5)
$$\frac{\mathrm{d}^{\mathrm{m}}}{\mathrm{d}t^{\mathrm{m}}} u(t) - \sum_{k=1}^{m-1} q_k \frac{\mathrm{d}^k}{\mathrm{d}t^k} u(t) - q_0 u(t) = 0,$$

where $q_j \in \mathbb{H}$, $(0 \le j \le m-1)$, can be transformed to the problem of finding the zeros of the corresponding *m*-order quaternionic polynomial,

(1.6)
$$P_l(z) = z^m - \sum_{j=0}^{m-1} q_j z^j.$$

To understand of the complexity of the quaternionic left and right eigenvalue problems is a challenge for mathematicians and physicists. Nowadays, the study of the eigenvalue problem for complex linear quaternionic operators [31] plays a fundamental role in solving quaternionic differential equations [28].

Many physical problems that are governed by differential operators are much simplified by applying the quaternionic matrix formalism and solving the corresponding quaternionic right eigenvalue problem. For example, the solutions of linear differential equations with quaternion constant coefficients

(1.7)
$$\frac{\mathrm{d}^{\mathrm{m}}}{\mathrm{d}t^{\mathrm{m}}} u(t) - q_{m-1} \frac{\mathrm{d}^{\mathrm{m}-1}}{\mathrm{d}t^{\mathrm{m}-1}} u(t) - q_{m-2} \frac{\mathrm{d}^{\mathrm{m}-2}}{\mathrm{d}t^{\mathrm{m}-2}} u(t) - \dots - q_0 u(t) = 0,$$

where $q_j \in \mathbb{H}$ $(0 \leq j \leq m-1)$, can be presented in terms of right eigenvalues and eigenvectors of the quaternionic matrix

$$\begin{bmatrix} q_{m-1} & q_{m-2} & \dots & q_0 \\ 1 & 0 & \dots & 0 \\ \vdots & \vdots & \ddots & \vdots \\ \vdots & \vdots & \ddots & \vdots \\ 0 & \dots & 1 & 0 \end{bmatrix}.$$

This thesis mainly deals with localization results of eigenvalues of a quaternionic matrix, bounds for eigenvalues of quaternionic matrix polynomials, bounds of zeros of quaternionic polynomials, and perturbation bounds for right eigenvalues of a quaternionic matrix/quaternionic matrix pencil. We developed localization theorems for the left and right eigenvalues of a quaternionic matrix which include the Gerschgorin, Ostrowski, and Brauer type theorems for the left and right eigenvalues of a quaternionic matrix. We proposed inclusion regions for right eigenvalues of special matrices, viz., central closed quaternionic matrices, quaternionic Hermitian matrices, and quaternionic η -Hermitian matrices, where $\eta \in {\mathbf{i}, \mathbf{j}, \mathbf{k}}$ and $\mathbf{i}, \mathbf{j}, \mathbf{k}$ are the standard quaternion imaginary units. A sufficient condition for the stability of a continuous-time quaternionic system is given. Bounds/location of zeros of quaternionic polynomials are proposed.

Thereafter inclusion regions for generalized right eigenvalues of a quaternionic matrix pencil and their applications are presented.

Next, the definitions of left and right eigenvalues of quaternionic matrix polynomials are proposed. We present bounds for the left and right eigenvalues of quaternionic matrix polynomials via localization theorems for left and right eigenvalues of a quaternionic matrix/quaternionic block matrix. Further, a sufficient condition for the stability of a discrete-time quaternionic system is given. Bounds for the absolute values of the left and right eigenvalues of quaternionic matrix polynomials are derived and illustrated for the quaternionic matrix p-norm, where $p = 1, 2, \infty$, and F (Frobenius). The above results generalize bounds for the absolute values of eigenvalues of complex matrix polynomials which give sharper bounds to the existing bounds for the case of 1, 2, and ∞ matrix norms. Finally, the concept of perturbation bounds for right eigenvalues/generalized right eigenvalues of a quaternionic matrix/quaternionic matrix pencil is developed. In particular, Bauer-Fike type theorems for right eigenvalues/generalized right eigenvalues of a diagonalizable quaternionic matrix/diagonalizable quaternionic matrix pencil are derived. Moreover, a relative perturbation bound for right eigenvalues of an invertible diagonalizable quaternionic matrix is discussed. A residual bound for right eigenvalues of a quaternionic Hermitian matrix is also considered. Perturbation bounds for zeros of quaternionic polynomials are presented.

This thesis is organized as follows. Chapter 2 derives location of the left and right eigenvalues of quaternionic matrices and bounds of zeros of quaternionic polynomials. Chapter 3 explores inclusion regions for zeros of quaternionic polynomials. Chapter 4 concerns localization theorems for generalized right eigenvalues of quaternionic matrix pencils and their applications. Chapter 5 is devoted to bounds of eigenvalues of quaternionic matrix polynomials. Perturbation analysis for quaternionic matrices and quaternionic polynomials are presented in Chapter 6. Finally, in Chapter 7, we give conclusions of this thesis and future prospect of our work.

1.2. Preliminaries

Throughout this thesis, \mathbb{R} and \mathbb{C} denote the fields of real and complex numbers, respectively. We define the closed upper complex halfplane as

$$\mathbb{C}^+ := \{ \alpha + \beta i : \alpha, \beta \in \mathbb{R}, \ \beta \ge 0 \}.$$

Define

$$\mathbb{R}^+ := \{ \alpha : \alpha \in \mathbb{R}, \alpha > 0 \}.$$

The set of real quaternions is defined by

$$\mathbb{H} := \{ q = a_0 + a_1 \mathbf{i} + a_2 \mathbf{j} + a_3 \mathbf{k} : a_0, a_1, a_2, a_3 \in \mathbb{R} \}$$

with $\mathbf{i}^2 = \mathbf{j}^2 = \mathbf{k}^2 = \mathbf{i}\mathbf{j}\mathbf{k} = -1$. This relation implies

$$\mathbf{ij} = -\mathbf{ji} = \mathbf{k}, \ \mathbf{jk} = -\mathbf{kj} = \mathbf{i}, \ \mathbf{ki} = -\mathbf{ik} = \mathbf{j}.$$

The conjugate and the modulus of $q = a_0 + a_1 \mathbf{i} + a_2 \mathbf{j} + a_3 \mathbf{k}$ are defined as

$$\overline{q} = q^H := a_0 - a_1 \mathbf{i} - a_2 \mathbf{j} - a_3 \mathbf{k} \text{ and } |q| := \sqrt{a_0^2 + a_1^2 + a_2^2 + a_3^2},$$

respectively.

 $\Im(a)$ denotes the imaginary part of $a \in \mathbb{C}$. The real and imaginary parts of a quaternion $q = a_0 + a_1 \mathbf{i} + a_2 \mathbf{j} + a_3 \mathbf{k}$ are defined as

$$\operatorname{Re}(q) := a_0 \text{ and } \operatorname{Im}(q) := a_1 \mathbf{i} + a_2 \mathbf{j} + a_3 \mathbf{k},$$

respectively.

The stability region of a continuous-time quaternionic system is defined as

(1.8)
$$S_{\mathbb{H}^-} := \{q \in \mathbb{H} : \operatorname{Re}(q) < 0\}.$$

Similarly, the stability region of a discrete-time quaternionic system is defined as

(1.9)
$$S_{\mathbb{H}} := \{q \in \mathbb{H} : |q| < 1\}.$$

Let $p, q \in \mathbb{H}$. Then p and q are said to be similar, denoted by $p \sim q$, if

(1.10)
$$p \sim q \Leftrightarrow \exists 0 \neq r \in \mathbb{H} \text{ such that } p = r^{-1}qr.$$

The set

(1.11)
$$[p] := \{ u \in \mathbb{H} : u = \rho^{-1} p \rho \text{ for all } 0 \neq \rho \in \mathbb{H} \}$$

is called an equivalence class of $p \in \mathbb{H}$.

It is known [61, Theorem 2.2] that

(1.12)
$$p \sim q \Leftrightarrow \operatorname{Re}(p) = \operatorname{Re}(q) \text{ and } |p| = |q|.$$

From (1.10), (1.11) and (1.12), [p] can be written as

(1.13)
$$[p] := \{ x \in \mathbb{H} : \operatorname{Re}(x) = \operatorname{Re}(p), \ |x| = |p| \}.$$

From (1.13), we have

$$\overline{p} \in [p].$$

The collections of all *n*-column vectors with entries in \mathbb{R} , \mathbb{C} and \mathbb{H} are denoted by \mathbb{R}^n , \mathbb{C}^n and \mathbb{H}^n , respectively.

For $x \in \mathcal{K}^n$, where $\mathcal{K} \in \{\mathbb{R}, \mathbb{C}, \mathbb{H}\}$, the transpose of x is x^T . If $x = [x_1, \ldots, x_n]^T$, the conjugate of x is defined as $\overline{x} := [\overline{x_1}, \ldots, \overline{x_n}]^T$ and the conjugate transpose of x is defined as $x^H := [\overline{x_1}, \ldots, \overline{x_n}]$.

Some elementary properties of the algebra of quaternions are listed below.

Proposition 1.1. Let $p, q, r \in \mathbb{H}$. Then

pp^H = pp^H;
 |p| = |p^H|;
 |.| is a norm on 𝔄, i.e., for all p, q ∈ 𝔄 we have:

$$\begin{split} |p| &\geq 0 \ \ with \ equality \ if \ and \ only \ if \ p=0; \\ |p+q| &\leq |p|+|q|; \\ |pq| &= |qp| = |p||q|; \end{split}$$

- 4. $\mathbf{j}c\mathbf{j}^H = \mathbf{k}c\mathbf{k}^H = \overline{c} \text{ for every } c \in \mathbb{C};$
- 5. $(pq)^H = q^H p^H;$
- 6. $p = p^H$ if and only if $p \in \mathbb{R}$;
- 7. $\alpha p = p\alpha$ for every $p \in \mathbb{H}$ if and only if $\alpha \in \mathbb{R}$;
- 8. every $p \in \mathbb{H} \setminus \{0\}$ has an inverse $p^{-1} = \frac{p^H}{|p|^2} \in \mathbb{H}$; in more detail,

$$p \times \left(\frac{p^H}{|p|^2}\right) = \left(\frac{p^H}{|p|^2}\right) \times p = 1;$$

- 9. (pq)r = p(qr);
- 10. in general, $(p+q)^2 \neq p^2 + 2pq + q^2$;
- 11. $p^2 = -1$ has infinitely many solutions over \mathbb{H} ;
- 12. for every $p \in \mathbb{H}$, p can be uniquely written as $p = c_1 + c_2 \mathbf{j}$, where $c_1, c_2 \in \mathbb{C}$.

We now briefly consider vector norm to be used in the subsequent development.

Definition 1.2. A function $\|\cdot\| : \mathbb{H}^n \to \mathbb{R}$ is said to be a quaternion norm on \mathbb{H}^n (or a quaternion vector norm) if $\|\cdot\|$ satisfies the following conditions:

- $||y|| = 0 \Leftrightarrow y = 0.$
- $\|\alpha y\| = |\alpha| \|y\|$ for $\alpha \in \mathbb{H}$ and $y \in \mathbb{H}^n$.
- $||y+z|| \le ||y|| + ||z||$ for $y, z \in \mathbb{H}^n$.

For $y, z \in \mathbb{H}^n$, define $\langle y, z \rangle := z^H y$ as an inner product and $\|y\|_2 := \sqrt{\langle y, y \rangle}$, the norm on \mathbb{H}^n . For $y \in \mathbb{H}^n$, the vector *p*-norm on \mathbb{H}^n is defined as

$$||y||_p := \begin{cases} (\sum_{j=1}^n |y_j|^p)^{1/p}, & \text{for } 1 \le p < \infty, \\ \max_{1 \le j \le n} |y_j|, & \text{for } p = \infty. \end{cases}$$

1.2.1. Quaternionic matrices

The sets of $m \times n$ real, complex, and quaternionic matrices are denoted by $M_{m \times n}(\mathbb{R})$, $M_{m \times n}(\mathbb{C})$, and $M_{m \times n}(\mathbb{H})$, respectively. These sets are simply denoted by $M_n(\mathcal{K})$, $\mathcal{K} \in \{\mathbb{R}, \mathbb{C}, \mathbb{H}\}$, when m = n. I_n denotes the $n \times n$ identity matrix. For $A = (a_{ij}) \in M_{m \times n}(\mathcal{K})$, the conjugate, transpose and conjugate transpose of A are defined as $\overline{A} = (\overline{a_{ij}})$, $A^T = (a_{ji}) \in M_{n \times m}(\mathcal{K})$, and $A^H = (\overline{A})^T \in M_{n \times m}(\mathcal{K})$, respectively. Let $A_j \in M_n(\mathcal{K})$ $(1 \le j \le k)$. Then, we denote the block diagonal matrix by diag (A_1, A_2, \ldots, A_k) , or by diag (A_j) . For $\mu_j \in \mathbb{H}$ $(1 \le j \le n)$, define

$$\operatorname{diag}(\mu_j) := \operatorname{diag}(\mu_1, \mu_2, \dots, \mu_n)$$

The Jordan block of size m associated with $\lambda \in \mathbb{H}$ is defined as

(1.14)
$$J_m(\lambda) := \begin{bmatrix} \lambda & 1 & 0 & \dots & 0 \\ 0 & \lambda & 1 & \dots & 0 \\ \vdots & \vdots & \ddots & \ddots & 0 \\ \vdots & \vdots & & \lambda & 1 \\ 0 & 0 & \dots & 0 & \lambda \end{bmatrix} \in M_m(\mathbb{H}), \quad \lambda \in \mathbb{H}.$$

For $A := (a_{ij}) \in M_n(\mathcal{K})$, the deleted absolute row and column sums of A are defined

as

(1.15)
$$r_i(A) := \sum_{j=1, j \neq i}^n |a_{ij}| \text{ and } c_i(A) := \sum_{j=1, j \neq i}^n |a_{ji}| \quad (1 \le i \le n),$$

respectively. Similarly, the absolute row and column sums of A are defined by

(1.16)
$$r'_i(A) := r_i(A) + |a_{ii}| \text{ and } c'_i(A) := c_i(A) + |a_{ii}| \quad (1 \le i \le n),$$

respectively. Let $A, B \in M_{m \times n}(\mathbb{H})$. Then we have the following properties:

- $\alpha A = A\alpha$, for all $\alpha \in \mathbb{R}$.
- $(\alpha A + \beta B)^H = A^H \alpha^H + B^H \beta^H$, for all $\alpha, \beta \in \mathbb{H}$.
- $(A\alpha + B\beta)^H = \alpha^H A^H + \beta^H B^H$, for all $\alpha, \beta \in \mathbb{H}$.

• $(A^H)^H = A$.

Let $A := (a_{ij}) \in M_n(\mathcal{K})$ be partitioned into $k \times k$ real/complex/quaternionic blocks

(1.17)
$$A := (A_{rs}) := \begin{bmatrix} A_{11} & A_{12} & \dots & A_{1k} \\ A_{21} & A_{22} & \dots & A_{2k} \\ \vdots & \vdots & \ddots & \vdots \\ A_{k1} & A_{k2} & \dots & A_{kk} \end{bmatrix}_{n \times n}$$

where $A_{r,s} \in M_{n_r \times n_s}(\mathcal{K})$ $(1 \le r, s \le k)$ is the (r, s) block of A such that $n_1 + \cdots + n_k = n$.

Definition 1.3. A matrix $A \in M_n(\mathbb{H})$ is said to be invertible (nonsingular) if there exists $B \in M_n(\mathbb{H})$ such that $AB = BA = I_n$, where I_n is the $n \times n$ identity matrix. If A is not invertible, then A is said to be singular.

Definition 1.4. Let $A \in M_n(\mathbb{H})$. Then the matrix A is said to be nilpotent if $A^k = 0$ for a least exponent k, where k is a positive integer and 0_n is the $n \times n$ zero matrix.

Definition 1.5. A function $\|.\| : M_{m \times n}(\mathbb{H}) \to \mathbb{R}$ is a quaternionic matrix norm on $M_{m \times n}(\mathbb{H})$ if it satisfies the following conditions:

- definiteness, $E \neq 0 \Rightarrow ||E|| \ge 0$;
- homogeneity, $\|\alpha E\| = |\alpha| \|E\|$;
- the triangle inequality, $||E + F|| \le ||E|| + ||F||$,

where $E, F \in M_{m \times n}(\mathbb{H})$ are arbitrary matrices and $\alpha \in \mathbb{H}$.

For $A \in M_n(\mathcal{K})$, the 1-norm, ∞ -norm, 2-norm (operator norm) and the Frobenius norm of $A \in M_n(\mathcal{K})$ are defined as

$$\|A\|_{1} := \max_{1 \le j \le n} \sum_{i=1}^{n} |a_{ij}| = \|A^{H}\|_{\infty}, \quad \|A\|_{\infty} := \max_{1 \le i \le n} \sum_{j=1}^{n} |a_{ij}| = \|A^{H}\|_{1},$$
$$\|A\|_{2} := \sup_{x \ne 0} \left\{ \frac{\|Ax\|_{2}}{\|x\|_{2}} : x \in \mathcal{K}^{n} \right\} = \|A^{H}\|_{2}, \text{ and } \|A\|_{F} := \left[\operatorname{trace} \left(A^{H} A \right) \right]^{1/2},$$

respectively.

Definition 1.6. Let $A \in M_n(\mathbb{H})$ be an invertible matrix. Then the condition number of A with respect to the matrix 2-norm is defined as

$$K_2(A) := \|A\|_2 \|A^{-1}\|_2.$$

Definition 1.7. Let $A := (a_{ij}) \in M_n(\mathbb{H})$. Then A is said to be

- 1. Hermitian if and only if $A^H = A$;
- 2. normal if and only if $A^H A = A A^H$;
- 3. unitary if and only if $A^H A = A A^H = I_n$;
- 4. upper triangular if and only if $a_{ij} = 0$ for all j < i;
- 5. lower triangular if and only if $a_{ij} = 0$ for all j > i;
- 6. η -Hermitian if and only if $A^{\eta H} = A$, where $A^{\eta H} = -\eta A^H \eta$ and $\eta \in \{\mathbf{i}, \mathbf{j}, \mathbf{k}\};$
- 7. η -anti-Hermitian if and only if $A^{\eta H} = -A$, where $A^{\eta H} = -\eta A^H \eta$ and $\eta \in \{\mathbf{i}, \mathbf{j}, \mathbf{k}\}$;
- 8. positive semidefinite if A is Hermitian matrix and $y^H A y \ge 0$ for all nonzero vector $y \in \mathbb{H}^n$;
- 9. positive definite if A is Hermitian matrix and $y^H A y > 0$ for all nonzero vector $y \in \mathbb{H}^n$.

Definition 1.8. Let $A \in M_n(\mathbb{H})$. Then A is said to be a central closed matrix if there exists an invertible matrix T such that

$$T^{-1}AT = \operatorname{diag}(\lambda_1, \lambda_2, \dots, \lambda_n),$$

where $\lambda_i \in \mathbb{R} \ (1 \le i \le n)$.

Some properties of matrices over the complex field do not hold over the skew field of quaternions. In general, for $A \in M_{m \times n}(\mathbb{H})$ and $B \in M_{n \times p}(\mathbb{H})$, $\overline{AB} \neq \overline{AB}$. We have the following facts for quaternionic matrices.

Proposition 1.9. [61] Let $A \in M_{m \times n}(\mathbb{H})$ and let $B \in M_{n \times p}(\mathbb{H})$. Then

- 1. $\left(\overline{A}\right)^T = \overline{A^T};$
- 2. $(AB)^H = B^H A^H;$
- 3. $(AB)^{-1} = B^{-1}A^{-1}$ if A and B are invertible;
- 4. $(A^{H})^{-1} = (A^{-1})^{H}$ if A is invertible;
- 5. in general, $(AB)^T \neq B^T A^T$;
- 6. in general, $(\overline{A})^{-1} \neq \overline{(A^{-1})};$
- 7. in general, $(A^T)^{-1} \neq (A^{-1})^T$.

1.2.2. Complex adjoint matrix

Definition 1.10. Let $x \in \mathbb{H}^n$. Then x can be uniquely expressed as $x = x_1 + x_2 \mathbf{j}$, where $x_1, x_2 \in \mathbb{C}^n$. Define a function $\psi : \mathbb{H}^n \to \mathbb{C}^{2n}$ by

$$\psi_x := \begin{bmatrix} x_1 \\ -\overline{x_2} \end{bmatrix}.$$

The vector ψ_x is called the complex adjoint vector of x. This function ψ is an injective linear transformation from \mathbb{H}^n to \mathbb{C}^{2n} .

Definition 1.11. Let $A \in M_n(\mathbb{H})$. Then A can be uniquely expressed as $A = A_1 + A_2 \mathbf{j}$, where $A_1, A_2 \in M_n(\mathbb{C})$. Define a function $\Psi : M_n(\mathbb{H}) \to M_{2n}(\mathbb{C})$ by

(1.18)
$$\Psi_A := \begin{bmatrix} A_1 & A_2 \\ -\overline{A_2} & \overline{A_1} \end{bmatrix}.$$

The matrix Ψ_A is called the complex adjoint matrix of A. This function Ψ is an injective H-homomorphism.

Theorem 1.12. [61] Let $A, B \in M_n(\mathbb{H})$ and let Ψ be the function which is defined in (1.18). Then

1. $\Psi_{I_n} = I_{2n};$

2.
$$\Psi_{AB} = \Psi_A \Psi_B$$
;

- 3. $\Psi_{\alpha A} = \alpha \Psi_A$, where $\alpha \in \mathbb{R}$;
- 4. $\Psi_{A+B} = \Psi_A + \Psi_B;$
- 5. $\Psi_{A^H} = (\Psi_A)^H;$
- 6. $\Psi_{A^{-1}} = (\Psi_A)^{-1}$ if A^{-1} exists;
- 7. Ψ_A is unitary, invertible, diagonalizable, Hermitian or normal if and only if A is unitary, invertible, diagonalizable, Hermitian or normal, respectively.

Lemma 1.13. [34] Let $A \in M_n(\mathbb{H})$. Then we have

$$\|A\|_{2} := \max_{\|x\|_{2} \neq 0} \frac{\|Ax\|_{2}}{\|x\|_{2}} = \max_{\|y\|_{2} \neq 0} \frac{\|\Psi_{A}y\|_{2}}{\|y\|_{2}} =: \|\Psi_{A}\|_{2},$$

where $\|x\|_{2} = \left(\sum_{i=1}^{n} \overline{x_{i}} x_{i}\right)^{\frac{1}{2}}$ on \mathbb{H}^{n} and $\|y\|_{2} = \left(\sum_{i=1}^{2n} \overline{y_{i}} y_{i}\right)^{\frac{1}{2}}$ on \mathbb{C}^{2n} .

1.2.3. The eigenvalue problem

We start this subsection with the definition of an eigenvalue of a complex matrix.

Definition 1.14. Let $A \in M_n(\mathbb{C})$ and let $\lambda \in \mathbb{C}$. Then λ is said to be an eigenvalue of A if $Ax = \lambda x$ for some nonzero $x \in \mathbb{C}^n$. The set

$$\Lambda(A) := \{ \lambda \in \mathbb{C} : Ax = \lambda x \text{ for some nonzero } x \in \mathbb{C}^n \}$$

is called the complex spectrum of A.

For example, let $A = \begin{bmatrix} \mathbf{i} & 2 + \mathbf{i} \\ 0 & 3\mathbf{i} \end{bmatrix}$. Then we have $\Lambda(A) = \{\mathbf{i}, 3\mathbf{i}\}$.

Unlike the complex case, there are two types of eigenvalues for quaternionic matrices, namely left and right eigenvalues which are defined as follows.

Definition 1.15. Let $A \in M_n(\mathbb{H})$ and let $\lambda \in \mathbb{H}$. Then λ is said to be a left eigenvalue of A if $Ay = \lambda y$ for some nonzero $y \in \mathbb{H}^n$. The set

$$\Lambda_l(A) := \{ \lambda \in \mathbb{H} : Ay = \lambda y \text{ for some nonzero } y \in \mathbb{H}^n \}$$

is called the left spectrum of A.

Similarly, λ is said to be a right eigenvalue of A if $Ay = y\lambda$ for some nonzero $y \in \mathbb{H}^n$. The set

$$\Lambda_r(A) := \{\lambda \in \mathbb{H} : Ay = y\lambda \text{ for some nonzero } y \in \mathbb{H}^n\}$$

is called the right spectrum of A.

Due to the commutativity of quaternions with real numbers, real right eigenvalues of $A \in M_n(\mathbb{H})$ are also left eigenvalues of A and vice versa. Thus, the quaternionic matrix A may have at most n real right eigenvalues. Hence the quaternionic matrix A may have at most n real left eigenvalues.

It is known that every quaternionic matrix $A \in M_n(\mathbb{H})$ has exactly *n* complex right eigenvalues which are contained in \mathbb{C}^+ . These are called the standard right eigenvalues of *A*.

Definition 1.16. Let $A \in M_n(\mathbb{H})$ and let $\lambda \in \mathbb{C}$. Then λ ($\Im(\lambda) \ge 0$) is said to be a standard right eigenvalue of A if $Ay = y\lambda$ for some nonzero $y \in \mathbb{H}^n$. The set

$$\Lambda_s(A) := \{ \lambda \in \mathbb{C} : Ay = y\lambda \text{ for some nonzero } y \in \mathbb{H}^n, \, \Im(\lambda) \ge 0 \}$$

is called the standard right spectrum of A.

Definition 1.17. Let $A \in M_n(\mathbb{H})$. Then the left spectral radius and the right spectral radius of A are given as

$$\rho_l(A) := \max\{|\lambda| : \lambda \in \Lambda_l(A)\}, \text{ and } \rho_r(A) := \max\{|\lambda| : \lambda \in \Lambda_r(A)\},\$$

respectively.

By applying the determinant of a complex matrix and the complex adjoint matrix of a quaternionic matrix, we present the Cayley-Hamilton theorem for the quaternion case.

Theorem 1.18. [61] Let $A \in M_n(\mathbb{H})$ and $P_A(\lambda) = \det(\lambda I_{2n} - \Psi_A)$, called the characteristic polynomial of A, where λ is a complex indeterminant. Then $P_A(A) = 0$ and $P_A(\lambda_0) = 0$ if and only if λ_0 is a right eigenvalue.

Proposition 1.19. [62] Let $A \in M_n(\mathbb{H})$ and let $\lambda \in \mathbb{H}$. Then λ is a left eigenvalue of A if and only if

$$\det\left[\Psi_{(A-\lambda I_n)}\right] = 0.$$

We have the following relation between left and right eigenvalues of a square real matrix.

Theorem 1.20. [61] Let A be a square real matrix. Then the left and right eigenvalues of A are same, i.e., $\Lambda_l(A) = \Lambda_r(A)$.

We can also compute the right eigenvalues of a quaternionic matrix with the help of the standard right eigenvalues of that quaternionic matrix. Let λ_i $(1 \leq i \leq n)$ be the standard right eigenvalues of a matrix $A \in M_n(\mathbb{H})$. Then, the set of right eigenvalues of A is given as

$$\Lambda_r(A) := \bigcup_{i=1}^n [\lambda_i],$$

where $[\lambda_i]$ are equivalence classes of λ_i $(1 \le i \le n)$, respectively.

Proposition 1.21. [27] Let $A \in M_n(\mathbb{H})$. Then A has exactly 2n complex right eigenvalues.

Proposition 1.22. [61] Let $A \in M_n(\mathbb{H})$. Then A has exactly n complex right eigenvalues which are contained in the closed upper complex halfplane \mathbb{C}^+ . These right eigenvalues are called the standard right eigenvalues of A. **Definition 1.23.** Let $A \in M_n(\mathbb{H})$. Then A is said to be a diagonalizable matrix if there exists an invertible quaternionic matrix P such that

$$P^{-1}AP = \operatorname{diag}(\lambda_1, \lambda_2, \dots, \lambda_n),$$

where $\lambda_i \in \Lambda_s(A) \ (1 \le i \le n).$

Definition 1.24. Let $A \in M_n(\mathbb{H})$. Then the continuous-time quaternionic system

$$\frac{\mathrm{d}}{\mathrm{dt}}\,w(t) = A\,w(t)$$

is stable if and only if $\Lambda_r(A) \subset S_{\mathbb{H}^-}$ (defined in (1.8)).

Definition 1.25. Let $A \in M_n(\mathbb{H})$. Then the discrete-time quaternionic system

$$w(t+1) = Aw(t)$$

is asymptotically stable if and only if $\Lambda_r(A) \subset S_{\mathbb{H}}$ (defined in (1.9)).

Next, we present the Jordan canonical form, singular-value decomposition, and Schur decomposition of a quaternionic matrix.

Proposition 1.26. [59] Let $A \in M_n(\mathbb{H})$. Then there exists an invertible matrix $Y \in M_n(\mathbb{H})$ such that

(1.19)
$$Y^{-1}AY = \operatorname{diag}(J_{m_1}(\lambda_1), J_{m_2}(\lambda_2), \dots, J_{m_k}(\lambda_k)),$$

where $\lambda_i \in \mathbb{H}, \lambda_i \in \Lambda_s(A)$ $(1 \leq i \leq k)$ and $J_{m_i}(\lambda_i)$ are the $m_i \times m_i$ Jordan blocks with right eigenvalues λ_i , respectively. Moreover, the right hand side of (1.19) is uniquely determined by A up to permutation of diagonal blocks, and up to replacement of each λ_j with any similar quaternion μ_j .

Theorem 1.27. [61] Let $A \in M_{m \times n}(\mathbb{H})$ be of rank r. Then there are two unitary matrices $U \in M_m(\mathbb{H})$ and $V \in M_n(\mathbb{H})$ such that

$$U^H A V = \begin{bmatrix} \Sigma_r & 0 \\ 0 & 0 \end{bmatrix},$$

where $\Sigma_r = \text{diag}(\sigma_1, \ldots, \sigma_r)$ and σ_i are the singular values of A.

Theorem 1.28. [9] Let $A \in M_n(\mathbb{H})$. Then there exist matrices $T, V \in M_n(\mathbb{H})$ such that:

1. $V^H A V = T$, where V is an unitary matrix and T is an upper triangular matrix;

2. every diagonal entry of T is contained in the closed upper complex halfplane \mathbb{C}^+ .

Theorem 1.29. [9] Let $A \in M_n(\mathbb{H})$ be normal. Then there exist an unitary matrix $V \in M_n(\mathbb{H})$ and a diagonal matrix $D \in M_n(\mathbb{H})$ such that:

- 1. $V^H A V = D;$
- 2. every diagonal entry of D is contained in the closed upper complex halfplane \mathbb{C}^+ .

We also need the following result for the development of our theory.

Proposition 1.30. [46] Let $A \in M_m(\mathbb{H})$ and let $B \in M_n(\mathbb{H})$. Then the equation

 $AX = XB, X \in M_{m \times n}(\mathbb{H})$

has only the trivial solution X = 0 if and only if $\Lambda_r(A) \cap \Lambda_r(B) = \emptyset$.

Further, we present some basic known facts on A and Ψ_A for the development of our theory.

Theorem 1.31. [61] Let $A \in M_n(\mathbb{H})$. Then the following statements are equivalent:

- 1. A is invertible;
- 2. Ax = 0 has a unique solution;
- 3. det $(\Psi_A) \neq 0$, *i.e.*, Ψ_A is invertible;
- A has no zero eigenvalue. More precisely, if Ax = λx or Ax = xλ for some λ ∈ ℍ
 and some vector 0 ≠ x ∈ ℍⁿ, then λ ≠ 0;
- 5. A is the product of elementary quaternionic matrices.

We next derive the following theorem which gives a method for diagonalization of a quaternionic matrix.

Theorem 1.32. [21] Let $A \in M_n(\mathbb{H})$. Then A is diagonalizable if and only if Ψ_A is diagonalizable. If Ψ_A is diagonalizable, then there exists an invertible matrix $T \in M_{2n}(\mathbb{C})$ such that

$$T^{-1}\Psi_A T = \begin{bmatrix} D & 0\\ 0 & \overline{D} \end{bmatrix} = \Psi_D \Leftrightarrow \Psi_A T = T\Psi_D,$$

where $D = \operatorname{diag}(\lambda_1, \lambda_2, \dots, \lambda_n)$ with $\lambda_i \in \Lambda_s(A)$ $(1 \le i \le n)$. Let

$$Y = \frac{1}{4} \begin{bmatrix} I_n & -\mathbf{j}I_n \end{bmatrix} (T + S_n^{-1}\overline{T}S_n) \begin{bmatrix} I_n \\ \mathbf{j}I_n \end{bmatrix},$$
where

$$S_n = \begin{bmatrix} 0 & -I_n \\ I_n & 0 \end{bmatrix}$$

Then $Y^{-1}AY = \text{diag}(\lambda_1, \lambda_2, \dots, \lambda_n)$, where Y is an invertible quaternionic matrix.

1.2.4. Quaternionic polynomials and their companion matrices

Due to the noncommutativity of quaternions, there are three types of quaternionic polynomials since the coefficients of polynomial can be taken to be on the left, on the right or on both sides of the indeterminant. However, throughout this thesis, we follow the following quaternionic polynomials:

(1.20)
$$p_l(z) := q_m z^m + q_{m-1} z^{m-1} + \dots + q_1 z + q_0,$$

(1.21)
$$p_r(z) := z^m q_m + z^{m-1} q_{m-1} + \dots + zq_1 + q_0,$$

where q_j , $z \in \mathbb{H}$, $(0 \leq j \leq m)$. The polynomials (1.20) and (1.21) are called "simple" and "monic" when $q_m = 1$. These polynomials play an important role in quaternion linear algebra since they are connected with linear difference and differential equations with quaternion coefficients.

The set of zeros of a quaternionic polynomial p(z) is denoted by $Z_{\mathbb{H}}(p(z))$. The set of complex zeros of a quaternionic polynomial p(z) is denoted by $Z_{\mathbb{C}}(p(z))$. For example, let us consider the quaternionic polynomial

$$p_l(z) := z^6 + \mathbf{j}z^5 + \mathbf{i}z^4 - z^2 - \mathbf{j}z - \mathbf{i}.$$

Then $Z_{\mathbb{C}}(p_l(z)) = \{1, -1, \mathbf{i}, -\mathbf{i}\}$ and

$$Z_{\mathbb{H}}(p_l(z)) = \{1, -1, [\mathbf{i}], (0.5 - 0.5\mathbf{i} - 0.5\mathbf{j} - 0.5\mathbf{k}), (-0.5 + 0.5\mathbf{i} - 0.5\mathbf{j} - 0.5\mathbf{k})\}.$$

The corresponding companion matrices of the simple monic polynomials $p_l(z)$ and $p_r(z)$ are given by

$$C_{p_l} := \begin{bmatrix} 0 & 1 & 0 \\ \vdots & \ddots & \\ 0 & 0 & 1 \\ -q_0 & -q_1 & \dots & -q_{m-1} \end{bmatrix} \text{ and } C_{p_r} := C_{p_l}^T,$$

respectively.

We now define the block forms of companion matrices C_{p_l} and C_{p_r} as follows.

$$C_{p_{l}} := \begin{bmatrix} 0 & 1 & 0 \\ \vdots & \ddots & \\ 0 & 0 & 1 \\ \hline -q_{0} & -q_{1} & \dots & -q_{m-1} \end{bmatrix} := \begin{bmatrix} 1 & m-1 \\ 0 & I \\ \hline C_{p_{l}}(m,1) & C_{p_{l}}(m,2:m) \end{bmatrix} \text{ and } C_{p_{r}} := C_{p_{l}}^{T}$$

respectively, where $C_{p_l}(m, 1) := -q_0$ and $C_{p_l}(m, 2:m) := [-q_1 \dots - q_{m-1}].$

For $p, q \in \mathbb{H}$, define $p \times q := pq$. For $0 \neq p \in \mathbb{H}$ and $q \in \mathbb{H}$, define

$$\frac{1}{p} \times q := p^{-1} \times q := p^{-1}q$$
, and $q \times \frac{1}{p} := q \times p^{-1} := qp^{-1}$

Consider $q_0 \neq 0$, i.e., z = 0 is not a zero of the simple monic polynomials $p_l(z)$ and $p_r(z)$. Define simple monic reversal polynomials of the polynomials $p_l(z)$ and $p_r(z)$ as follows.

(1.22)
$$q_l(z) := \frac{1}{q_0} \times p_l\left(\frac{1}{z}\right) \times z^m = z^m + q_0^{-1}q_1z^{m-1} + \dots + q_0^{-1}q_{m-1}z + q_0^{-1},$$

(1.23) $q_r(z) := z^m \times p_r\left(\frac{1}{z}\right) \times \frac{1}{q_0} = z^m + z^{m-1}q_1q_0^{-1} + \dots + zq_{m-1}q_0^{-1} + q_0^{-1},$

respectively. Similarly, the corresponding companion matrices of the simple monic reversal polynomials $q_l(z)$ and $q_r(z)$ are denoted by C_{q_l} and C_{q_r} , respectively. We observe that the zeros of $q_l(z)$ and $q_r(z)$ are reciprocals of the zeros of $p_l(z)$ and $p_r(z)$, respectively.

Remark 1.33. If the simple quaternionic polynomial $p_l(z)$ is not monic, we can always factorize. For example, if $p_l(z) := q_m z^m + q_{m-1} z^{m-1} + \cdots + q_1 z + q_0$ with $q_m \neq 0$, then $p_l(z) = q_m p'_l(z)$ and $p'_l(z)$ is a simple monic polynomial.

Remark 1.34. To compute bounds for the zeros of the quaternionic polynomial $p_r(z) = \sum_{j=0}^{m} z^j q_j$, we define a quaternionic polynomial

$$P_l(z) := \sum_{j=0}^m \overline{q_j} z^j.$$

If z_0 is a zero of $P_l(z)$, then

$$\overline{P_l(z_0)} = \overline{\sum_{j=0}^m \overline{q_j} z^j} = \sum_{j=0}^m \overline{z}^j q_j = p_r(\overline{z_0}) = 0.$$

Thus, $\overline{z_0}$ is a zero of $p_r(z)$ and, therefore, it is sufficient to treat only one type of polynomial, preferably the polynomial $p_l(z)$.

Also, we need the following result for improvement of our theory.

Proposition 1.35. [50] Let $p_l(z)$ be a simple monic polynomial over \mathbb{H} of degree m. Then the set of zeros of $p_l(z)$ belongs to at most m equivalence classes of quaternions.

Proposition 1.36. [50] Let $\lambda \in \mathbb{H}$. Then λ is a zero of the simple monic polynomial $p_l(z)$ if and only if λ is a left eigenvalue of its corresponding companion matrix C_{p_l} .

In general, a right eigenvalue of C_{p_l} is not necessarily a zero of the simple monic polynomial $p_l(z)$. For example, consider a simple monic polynomial $p_l(z) = z^2 + \mathbf{j}z + 2$. Then its companion matrix is given by

$$C_{p_l} = \begin{bmatrix} 0 & 1 \\ -2 & -\mathbf{j} \end{bmatrix}.$$

Here **i** is a right eigenvalue of C_{p_l} . However, **i** is not a zero of $p_l(z)$.

Analogous to Proposition 1.36, the following result is presented for $p_r(z)$.

Proposition 1.37. Let $\lambda \in \mathbb{H}$. Then λ is a zero of the simple monic polynomial $p_r(z)$ if and only if λ is a left eigenvalue of its corresponding companion matrix C_{p_r} .

Proof. Let λ be a left eigenvalue of C_{p_r} . Then there exists some nonzero $x \in \mathbb{H}^n$ such that $C_{p_r}x = \lambda x$. Let $x := [x_1, \ldots, x_m]^T \in \mathbb{H}^n$. Then

(1.24)
$$\begin{bmatrix} 0 & \dots & 0 & -q_0 \\ 1 & 0 & -q_1 \\ & \ddots & & \\ 0 & 1 & -q_{m-1} \end{bmatrix} \begin{bmatrix} x_1 \\ x_2 \\ \vdots \\ x_m \end{bmatrix} = \lambda \begin{bmatrix} x_1 \\ x_2 \\ \vdots \\ x_m \end{bmatrix}.$$

(1.24) gives the following system of linear equations

$$-q_0 x_m = \lambda x_1,$$
$$x_1 - q_1 x_m = \lambda x_2,$$
$$x_2 - q_2 x_m = \lambda x_3,$$
$$\vdots$$
$$x_{m-1} - q_{m-1} x_m = \lambda x_m.$$

By solving the above system of linear equations, we have

$$\left(\lambda^m + \lambda^{m-1}q_{m-1} + \dots + \lambda q_1 + q_0\right)x_m = 0$$

Since x cannot be the zero quaternionic vector, $x_m \neq 0$. Hence we conclude that

$$\lambda^m + \lambda^{m-1}q_{m-1} + \dots + \lambda q_1 + q_0 = 0.$$

Thus, the left eigenvalue λ is a zero of $p_r(z)$.

Lemma 1.38. [50] Let $\lambda \in \mathbb{H}$ be a left eigenvalue of C_{p_l} . Then it is also a right eigenvalue of C_{p_l} .

Unlike the case of complex polynomials, quaternionic polynomials may have infinitely many zeros. For example, let us consider the following polynomial $p_l(z)$ over \mathbb{H} :

$$p_l(z) = z^6 + (\mathbf{i} + 3\mathbf{k})z^5 + (3 + \mathbf{j})z^4 + (5\mathbf{i} + 15\mathbf{k})z^3 + (-4 + 5\mathbf{j})z^2 + (6\mathbf{i} + 18\mathbf{k})z + (6\mathbf{j} - 12).$$

The set of zeros of $p_l(z)$ are given in [50] as follows.

$$Z_{\mathbb{H}}(p_l(z)) = \{-\mathbf{i} - 2\mathbf{k}, [\mathbf{i}\sqrt{3}], [\mathbf{i}\sqrt{2}], -0.6\mathbf{i} - 0.8\mathbf{k}\}.$$

We observe that the following relation between $Z_{\mathbb{H}}(p_l(z))$, $\Lambda_l(C_{p_l})$ and $\Lambda_r(C_{p_l})$.

• From Proposition 1.36, we have $Z_{\mathbb{H}}(p_l(z)) = \Lambda_l(C_{p_l})$. For instance, let

$$p_l(z) := z^4 + 2z^3 - z^2 + 2z + 1.$$

Then, the set of right eigenvalues of C_{p_l} is given by

$$\Lambda_r(C_{p_l}) := \{-2.6180, -0.3820, [0.5000 + 0.8660\mathbf{i}]\}.$$

From Theorem 1.20, we have $\Lambda_l(C_{p_l}) = \Lambda_r(C_{p_l})$. Thus, the set of zeros of $p_l(z)$ is given by

$$Z_{\mathbb{H}}(p_l(z)) = \{-2.6180, -0.3820, [0.5000 + 0.8660\mathbf{i}]\}.$$

Hence, $Z_{\mathbb{H}}(p_l(z)) = \Lambda_l(C_{p_l}).$

• Let $\lambda \in Z_{\mathbb{H}}(p_l(z))$. Then from Proposition 1.36, we have $\lambda \in \Lambda_l(C_{p_l})$. From Lemma 1.38, we obtain $\lambda \in \Lambda_r(C_{p_l})$. This implies that $Z_{\mathbb{H}}(p_l(z)) \subseteq \Lambda_r(C_{p_l})$. For example, let us consider the quaternionic polynomial

$$p_l(z) := z^6 + \mathbf{j}z^5 + \mathbf{i}z^4 - z^2 - \mathbf{j}z - \mathbf{i}.$$

Then, the set of right eigenvalues of C_{p_l} is given by

$$\Lambda_r(C_{p_l}) := \{1, -1, [\mathbf{i}], [0.5 + 0.5\mathbf{i}], [-0.5 - 0.5\mathbf{i}]\}$$

The set of zeros of $p_l(z)$ is given in [19] as follows.

$$Z_{\mathbb{H}}(p_l(z)) = \{1, -1, [\mathbf{i}], (0.5 - 0.5\mathbf{i} - 0.5\mathbf{j} - 0.5\mathbf{k}), (-0.5 + 0.5\mathbf{i} - 0.5\mathbf{j} - 0.5\mathbf{k})\}.$$

Since 0.5 - 0.5i - 0.5j - 0.5k is similar to 0.5 + 0.5i and -0.5 + 0.5i - 0.5j - 0.5k is similar to -0.5 - 0.5i, we have

$$Z_{\mathbb{H}}(p_l(z)) \subseteq \Lambda_r(C_{p_l})$$

• It is never true that $Z_{\mathbb{H}}(p_l(z)) \supset \Lambda_r(C_{p_l})$. From Proposition 1.35, it is known that all the zeros of $p_l(z)$ belong to at most m equivalence classes of standard right eigenvalues of C_{p_l} .

1.2.5. The generalized eigenvalue problem

Let $\mathbb{L}'_1(M_n(\mathbb{H}))$ be the space of matrix pencils over a quaternion division algebra. $\mathbb{L}'_1 \in \mathbb{L}'_1(M_n(\mathbb{H}))$ is defined as

(1.25)
$$\mathbf{L}_{1}^{\prime}(\lambda) := A + \lambda B,$$

where $\lambda \in \mathbb{H}$ and $A, B \in M_n(\mathbb{H})$. Throughout this thesis we consider the following three cases:

Case 1: when $\lambda \in \mathbb{R}$ and $A, B \in M_n(\mathbb{H})$,

Case 2: when $\lambda \in \mathbb{H}$ and $A, B \in M_n(\mathbb{H})$,

Case 3: when $\lambda \in \mathbb{C}$ and $A, B \in M_n(\mathbb{C})$.

Case 1. Let $\mathbb{L}_1(M_n(\mathbb{H}))$ be the space of matrix pencils over a quaternion division algebra. $\mathbf{L}_1 \in \mathbb{L}_1(M_n(\mathbb{H}))$ is defined as

(1.26)
$$\mathbf{L}_1(\lambda) := A + \lambda B,$$

where $A, B \in M_n(\mathbb{H})$ and λ commutes with the quaternionic matrices. This matrix pencil over a quaternion division algebra can be found in [33, 46, 47]. Now we turn to define generalized right eigenvalue of $\mathbf{L}_1 \in \mathbb{L}_1(M_n(\mathbb{H}))$ of the form (1.26) as follows.

Definition 1.39. Let $\mathbf{L}_1 \in \mathbb{L}_1(M_n(\mathbb{H}))$ be as in (1.26) and let $\mu \in \mathbb{H}$. Then μ is called a generalized right eigenvalue of \mathbf{L}_1 if

$$Ax = Bx\mu$$
19

for some nonzero $x \in \mathbb{H}^n$. Here x is called the right eigenvector corresponding to the generalized right eigenvalue μ . The set of generalized right eigenvalues of \mathbf{L}_1 is called right spectrum of \mathbf{L}_1 , denoted by $\Lambda_r(\mathbf{L}_1)$.

Definition 1.40. Let $\mathbf{L}_1 \in \mathbb{L}_1(M_n(\mathbb{H}))$ be as in (1.26). Then the matrix pencil \mathbf{L}_1 is called regular if there exists $\alpha \in \mathbb{R}$ such that $A + \alpha B$ is an invertible matrix.

Let $\mathbb{P}_1(M_{2n}(\mathbb{C}))$ be the space of complex matrix pencils. $P_1 \in \mathbb{P}_1(M_{2n}(\mathbb{C}))$ is defines as

$$(1.27) P_1(\mu) := \Psi_A + \mu \Psi_B,$$

where $A, B \in M_n(\mathbb{H})$ and $\mu \in \mathbb{C}$.

Then, we have the following relation between quaternionic matrix pencils and complex matrix pencils.

Lemma 1.41. [33] Let $\mathbf{L}_1 \in \mathbb{L}_1(M_n(\mathbb{H}))$ be of the form $\mathbf{L}_1(\lambda) := A + \lambda B$, where λ commutes with the quaternionic matrices. Let $P_1 \in \mathbb{P}_1(M_{2n}(\mathbb{C}))$ be of the form $P_1(\mu) := \Psi_A + \mu \Psi_B$, where $\mu \in \mathbb{C}$. Then

- the quaternionic matrix pencil L₁ is regular if and only if the complex matrix pencil P₁ is regular;
- the quaternionic matrix pencil L₁ is regular if and only if a quaternionic matrix pencil L₂ is regular, where L₂(λ) := B + λA.

We now give a relation between the set of generalized right eigenvalues of a quaternionic matrix pencil and the set of eigenvalues of the complex adjoint matrix of that quaternionic matrix pencil as follows.

Proposition 1.42. [33] Let $\mathbf{L} \in \mathbb{L}_1(M_n(\mathbb{H}))$ be of the form $\mathbf{L}(\lambda) := A + \lambda B$, where λ commutes with A and B. Let $P \in \mathbb{P}_1(M_{2n}(\mathbb{C}))$ be of the form $P(\mu) := \Psi_A + \mu \Psi_B$, where $\mu \in \mathbb{C}$. Then

(1.28)
$$\begin{cases} \Lambda_r(\mathbf{L}) \cap \mathbb{C} = \Lambda(P), \\ \Lambda_r(\mathbf{L}) = \{\rho^{-1}\lambda\rho : \lambda \in \Lambda(P), \ 0 \neq \rho \in \mathbb{H} \} \end{cases}$$

Assume that B = I in Proposition 1.42, we have the following corollary which also can be found from [33].

Corollary 1.43. Let $A \in M_n(\mathbb{H})$. Then

$$\Lambda_r(A) \cap \mathbb{C} = \Lambda(\Psi_A),$$

$$\Lambda_r(A) = \{\rho^{-1}\lambda\rho : \lambda \in \Lambda(\Psi_A), \ 0 \neq \rho \in \mathbb{H}\}$$

Case 2. We define generalized left eigenvalue of $\mathbf{L}'_1 \in \mathbb{L}'_1(M_n(\mathbb{H}))$ of the form (1.25) as follows.

Definition 1.44. Let $\mathbf{L}'_1 \in \mathbb{L}'_1(M_n(\mathbb{H}))$ be of the form (1.25) and let $\mu \in \mathbb{H}$. Then μ is called a generalized left eigenvalue of \mathbf{L}'_1 if

$$Ax = \mu Bx$$

for some nonzero $x \in \mathbb{H}^n$. Here x is called the left eigenvector corresponding to the generalized left eigenvalue μ . The set of generalized left eigenvalues of \mathbf{L}'_1 is called generalized left spectrum of \mathbf{L}'_1 , denoted by $\Lambda_l(\mathbf{L}'_1)$.

1.2.6. The polynomial eigenvalue problem

Let $\mathcal{P}_m(M_n(\mathbb{C}))$ be the space of complex matrix polynomials. $\mathcal{P} \in \mathcal{P}_m(M_n(\mathbb{C}))$ is defined by

$$\mathcal{P}(\lambda) := \sum_{i=0}^{m} \lambda^{i} A_{i},$$

where $A_i \in M_n(\mathbb{C})$ $(0 \le i \le m)$ and $\lambda \in \mathbb{C}$. Then the eigenvalue problem $\mathcal{P}(\lambda)x = 0$ is referred as a complex polynomial eigenvalue problem. The polynomial $\mathcal{P} \in \mathcal{P}_m(M_n(\mathbb{C}))$ is said to be regular if $\det(\mathcal{P}(\lambda)) \ne 0$ for some $\lambda \in \mathbb{C}$. The spectrum of a regular polynomial \mathcal{P} is denoted by $\Lambda(\mathcal{P})$ and is defined by

$$\Lambda(\mathcal{P}) := \{\lambda \in \mathbb{C} : \det(\mathcal{P}(\lambda)) = 0\}.$$

Let $\mathbb{P}_m(M_{2n}(\mathbb{C}))$ be the space of complex matrix polynomials. $P \in \mathbb{P}_m(M_{2n}(\mathbb{C}))$ is defined as

(1.29)
$$P(\mu) := \sum_{i=0}^{m} \Psi_{A_i} \mu^i,$$

where $A_i \in M_n(\mathbb{H})$ $(0 \le i \le m)$ and $\mu \in \mathbb{C}$.

Let $\mathbb{L}'_m(M_n(\mathbb{H}))$ be the space of matrix polynomials over a quaternion division algebra. $\mathbf{L}' \in \mathbb{L}'_m(M_n(\mathbb{H}))$ is defined as

(1.30)
$$\mathbf{L}'(\xi) := \sum_{i=0}^{m} \xi^i A_i,$$

where $\xi \in \mathbb{H}$ and $A_i \in M_n(\mathbb{H})$ $(0 \le i \le m)$. Throughout this thesis we consider the following three cases:

Case 1: when $\xi \in \mathbb{R}$ and $A_i \in M_n(\mathbb{H}) \ (0 \le i \le m)$,

Case 2: when $\xi \in \mathbb{H}$ and $A_i \in M_n(\mathbb{H}) \ (0 \le i \le m)$,

Case 3: when $\xi \in \mathbb{C}$ and $A_i \in M_n(\mathbb{C}) \ (0 \le i \le m)$.

Case 1. Let $\mathbb{L}_m(M_n(\mathbb{H}))$ be the space of matrix polynomials over a quaternion division algebra. $\mathbf{L} \in \mathbb{L}_m(M_n(\mathbb{H}))$ is defined as

(1.31)
$$\mathbf{L}(\lambda) := \sum_{i=0}^{m} A_i \lambda^i,$$

where $A_i \in M_n(\mathbb{H})$ $(0 \leq i \leq m)$ and λ commutes with the quaternionic coefficients of the matrix polynomial. This polynomial over a quaternion division algebra can be found in [46–48].

We now turn to define right eigenvalue of $\mathbf{L} \in \mathbb{L}_m(M_n(\mathbb{H}))$ of the form (1.31) as follows.

Definition 1.45. Let $\mathbf{L} \in \mathbb{L}_m(M_n(\mathbb{H}))$ be as in (1.31) and let $\mu \in \mathbb{H}$. Then μ is called a right eigenvalue of \mathbf{L} if

$$A_0 x + A_1 x \mu + A_2 x \mu^2 + \dots + A_m x \mu^m = 0$$

for some nonzero $x \in \mathbb{H}^n$. Here x is called the right eigenvector corresponding to the right eigenvalue μ . The set of right eigenvalues of **L** is called right spectrum of **L**, denoted by Λ_r (**L**).

Case 2. Left eigenvalue of $\mathbf{L}' \in \mathbb{L}'_m(M_n(\mathbb{H}))$ of the form (1.30) is defined as follows.

Definition 1.46. Let $\mathbf{L}' \in \mathbb{L}'_m(M_n(\mathbb{H}))$ be of the form (1.30) and let $\mu \in \mathbb{H}$. Then μ is called a left eigenvalue of \mathbf{L}' if

$$A_0 x + \mu A_1 x + \mu^2 A_2 x + \dots + \mu^m A_m x = 0$$

for some nonzero $x \in \mathbb{H}^n$. Here x is called the left eigenvector corresponding to the left eigenvalue μ . The set of left eigenvalues of \mathbf{L}' is called left spectrum of \mathbf{L}' , denoted by $\Lambda_l(\mathbf{L}')$.

We write the linearization of the matrix polynomial $\mathbf{L} \in \mathbb{L}_m(M_n(\mathbb{H}))$ as follows.

• For the right eigenvalues: The polynomial $\mathbf{L} \in \mathbb{L}_m(M_n(\mathbb{H}))$ of the form (1.31) can be written in the form:

$$C_{\mathbf{L}} + \lambda X,$$

where $C_{\mathbf{L}}, X \in M_{mn}(\mathbb{H})$ are of the forms

$$C_{\mathbf{L}} := \begin{bmatrix} 0 & I_n & 0 & \dots & 0 \\ 0 & 0 & I_n & & 0 \\ \vdots & \vdots & & \ddots & \vdots \\ 0 & 0 & 0 & \dots & I_n \\ -A_0 & -A_1 & -A_2 & \dots & -A_{m-1} \end{bmatrix}, \quad X := \begin{bmatrix} I_n & 0 & 0 & \dots & 0 \\ 0 & I_n & 0 & \dots & 0 \\ 0 & 0 & I_n & & 0 \\ \vdots & \vdots & & \ddots & \vdots \\ 0 & 0 & 0 & \dots & A_m \end{bmatrix}$$

and λ commutes with quaternionic coefficients of matrix polynomial. When $A_m = I_n$, the identity matrix, the matrix polynomial (1.31) is said to be monic matrix polynomial and its linearization is given by

$$C_{\mathbf{L}} + \lambda E$$
, where $E := I_{nm}$.

We next find the corresponding linearization of the matrix polynomial $\mathbf{L}' \in \mathbb{L}'_m(M_n(\mathbb{H}))$ as follows.

• For the left eigenvalues: The polynomial $\mathbf{L}' \in \mathbb{L}'_m(M_n(\mathbb{H}))$ of the form (1.30) can be written in the linearization form:

$$C_{\mathbf{L}'} + \xi X,$$

where $\xi \in \mathbb{H}, C_{\mathbf{L}'}, X \in M_{mn}(\mathbb{H})$ are of the forms

$$C_{\mathbf{L}'} := \begin{bmatrix} 0 & 0 & 0 & \dots & -A_0 \\ I_n & 0 & 0 & \dots & -A_1 \\ & \ddots & \vdots & \dots & \vdots \\ 0 & 0 & I_n & -A_{m-2} \\ 0 & 0 & \dots & I_n & -A_{m-1} \end{bmatrix}, \quad X := \begin{bmatrix} I_n & 0 & 0 & \dots & 0 \\ 0 & I_n & 0 & \dots & 0 \\ 0 & 0 & I_n & 0 \\ \vdots & \vdots & \ddots & \vdots \\ 0 & 0 & 0 & \dots & A_m \end{bmatrix}$$
23

When $A_m = I_n$, the identity matrix, then the polynomial $\mathbf{L}' \in \mathbb{L}'_m(M_n(\mathbb{H}))$ of the form (1.30) is said to be monic matrix polynomial and its linearization is given by

 $C_{\mathbf{L}'} + \xi E$, where $E := I_{nm}$.

CHAPTER 2

LOCALIZATION THEOREMS OF MATRICES OVER A QUATERNION DIVISION ALGEBRA

In this chapter, Gerschgorin, Ostrowski, and Brauer type theorems are derived for the left and right eigenvalues of a quaternionic matrix. Generalizations of Gerschgorin type theorems are discussed for the left and right eigenvalues of a quaternionic matrix. Thereafter a sufficient condition for the stability of a continuous-time quaternionic system is given that generalizes the stability condition for a continuous-time complex system. Finally, a characterization of bounds for the zeros of quaternionic polynomials is presented.

2.1. Introduction

This chapter attempts to study localization theorems for matrices over a quaternion division algebra, which include the Ostrowski, Brauer, and Gerschgorin type theorems. Bounds for the zeros of quaternionic polynomials $p_l(z)$ and $p_r(z)$ (defined in (1.20) and (1.21)) are also considered.

In Section 2.2, we provide a general framework for localization theorems for quaternionic matrices. Let $M_n(\mathbb{H})$ be the space of all $n \times n$ quaternionic matrices. Then, for any $A := (a_{ij}) \in M_n(\mathbb{H})$, we prove a Ostrowski type theorem which states that all the left eigenvalues of A are located in the union of n balls

$$T_i(A) := \{ z \in \mathbb{H} : |z - a_{ii}| \le r_i(A)^{\gamma} c_i(A)^{1 - \gamma} \},\$$

where $r_i(A)$ and $c_i(A)$ are defined in (1.15) and $\forall \gamma \in [0, 1]$. We deduce a sufficient condition for invertibility of a quaternionic matrix. We proved that Ostrowski type theorem is also valid for right eigenvalues when all the diagonal entries of the quaternionic matrix A are real. We find that the Brauer type theorem, proved in [22] for the left eigenvalues in the case of deleted absolute column sums of a quaternionic matrix, is incorrect. We prove a corrected version of the Brauer type theorem. In addition, we derive some better results than [22, Theorems 6, 7] and [63, Theorem 4.3]. In the case of the generalized Hölder inequality over the skew field of quaternions, we show that all the left eigenvalues of $A = (a_{ij}) \in M_n(\mathbb{H})$ are contained in the union of n generalized balls

$$B_i(A) := \{ z \in \mathbb{H} : |z - a_{ii}| \le (n-1)^{\frac{1-\gamma}{q}} r_i(A)^{\gamma} (n_i^{(p)}(A))^{1-\gamma} \},\$$

where $\gamma \in [0, 1]$, $n_i^{(p)}(A) := \left(\sum_{j=1, j \neq i}^n |a_{ij}|^p\right)^{\frac{1}{p}}$, for any $p, q \in (1, \infty)$ with $\frac{1}{p} + \frac{1}{q} = 1$. Further, we prove that all the right eigenvalues of the quaternionic matrix A with all real diagonal entries are contained in the union of n generalized balls $B_i(A)$.

In Section 2.3, we provide bounds for the zeros of quaternionic polynomials $p_l(z)$ and $p_r(z)$ by using the aforementioned localization theorems. Some recent developments on the location and computation of zeros of quaternionic polynomials $p_l(z)$ and $p_r(z)$ can be found in [12, 19, 20, 30, 41, 42, 45, 50]. As a consequence of the localization theorems for quaternionic matrices, we provide sharper bounds compared to the bound introduced by G. Opfer in [42] for the zeros of quaternionic polynomials $p_l(z)$ and $p_r(z)$. Finally, we provide sharper bounds for the zeros of quaternionic polynomials $p_l(z)$ and $p_r(z)$ in terms of powers of the companion matrices associated with the quaternionic polynomials $p_l(z)$ and $p_r(z)$ in terms of powers of the companion matrices are better than that in [42].

2.2. Distribution for the left and right eigenvalues of quaternionic

matrices

It is known from [47, Corollary 3.2] that a quaternionic matrix A and its conjugate transpose A^H have the same right eigenvalues. However, A and A^H may not have the same left eigenvalues. For example, the matrices

$$A = \begin{bmatrix} \mathbf{i} & 0 \\ 0 & \mathbf{j} \end{bmatrix} \text{ and } A^H = \begin{bmatrix} -\mathbf{i} & 0 \\ 0 & -\mathbf{j} \end{bmatrix}$$

do not have the same left eigenvalues. We now present the following lemma for left eigenvalues of A and A^{H} .

Lemma 2.1. Let $A \in M_n(\mathbb{H})$ and let $\lambda \in \mathbb{H}$. Then λ is a left eigenvalue of A if and only if $\overline{\lambda}$ is a left eigenvalue of A^H .

Proof. Let λ be a left eigenvalue of A. Then there exists $x \neq 0 \in \mathbb{H}^n$ such that $(A - \lambda I_n)x = 0$. This can be written as $\Psi_{(A-\lambda I_n)}\psi_x = 0$. Hence it follows that λ is a left eigenvalue of A if and only if det $[\Psi_{(A-\lambda I_n)}] = 0 \Leftrightarrow \det [\Psi_{(A-\lambda I_n)}] = 0$. Thus, $\overline{\lambda}$ is a left eigenvalue of A^H .

The Gerschgorin type theorem is proved in [62] for the left eigenvalues in the case of deleted absolute row sums of a matrix $A \in M_n(\mathbb{H})$. However, Gerschgorin type theorem for the left eigenvalues has not established for the deleted absolute column sums of A. We state the following Gerschgorin type theorem for the deleted absolute column sums of A.

Theorem 2.2. Let $A := (a_{ij}) \in M_n(\mathbb{H})$. Then all the left eigenvalues of A are located in the union of n Gerschgorin balls $\Omega_i(A) := \{z \in \mathbb{H} : |z - a_{ii}| \le c_i(A)\}$ $(1 \le i \le n), i.e.,$

$$\Lambda_l(A) \subseteq \Omega(A) := \bigcup_{i=1}^n \Omega_i(A).$$

Proof. Let λ be a left eigenvalue of A. Then from Lemma 2.1, $\overline{\lambda}$ is a left eigenvalue of A^H . Then there exists some nonzero $x \in \mathbb{H}^n$ such that $A^H x = \overline{\lambda} x$. Let $x := [x_1, \ldots, x_n]^T \in \mathbb{H}^n$ and let x_t be an element of x such that $|x_t| \ge |x_i| \forall i \ (1 \le i \le n)$. Then, $|x_t| > 0$. From the *t*-th equation of $A^H x = \overline{\lambda} x$, we have

$$\sum_{j=1}^{n} \overline{a_{jt}} x_j = \overline{\lambda} x_t.$$

This shows

$$|\lambda - a_{tt}| \leq \sum_{j=1, j \neq t}^{n} |a_{jt}| := c_t(A). \blacksquare$$

We now have the following localization theorem for the deleted absolute row and column sums of a matrix $A \in M_n(\mathbb{H})$ which is known as *Ostrowski type theorem*.

Theorem 2.3. (Ostrowski type theorem for the left eigenvalues) Let $A := (a_{ij}) \in M_n(\mathbb{H})$ and let $\gamma \in [0,1]$. Then all the left eigenvalues of A are located in the union of n balls $T_i(A) := \{z \in \mathbb{H} : |z - a_{ii}| \le r_i(A)^{\gamma} c_i(A)^{1-\gamma}\}$ $(1 \le i \le n), i.e.,$

$$\Lambda_l(A) \subseteq T(A) := \bigcup_{i=1}^n T_i(A).$$

Proof. Let λ be a left eigenvalue of A. Then by [62, Theorem 6], for $\gamma \in [0, 1]$, we have

(2.1)
$$|\lambda - a_{ii}|^{\gamma} \le r_i(A)^{\gamma} \qquad (1 \le i \le n).$$

Similarly, from Theorem 2.2, we obtain

(2.2)
$$|\lambda - a_{ii}|^{1-\gamma} \le c_i(A)^{1-\gamma} \quad (1 \le i \le n).$$

Combining (2.1) and (2.2), we get

$$|\lambda - a_{ii}| \le r_i(A)^{\gamma} c_i(A)^{1-\gamma} \qquad (1 \le i \le n).$$

Thus, all the left eigenvalues of A are located in the union of n balls $T_i(A)$.

Next, we derive Ostrowski type theorem for right eigenvalues of $A \in M_n(\mathbb{H})$ with all real diagonal entries.

Theorem 2.4. Let $A := (a_{ij}) \in M_n(\mathbb{H})$ with $a_{ii} \in \mathbb{R}$ and let $\gamma \in [0, 1]$. Then all the right eigenvalues of A are located in the union of n balls

$$G_i(A) := \left\{ z \in \mathbb{H} : |z - a_{ii}| \le r_i(A)^{\gamma} c_i(A)^{1-\gamma} \right\} \quad (1 \le i \le n), \ i.e.,$$
$$\Lambda_r(A) \subseteq G(A) := \cup_{i=1}^n G_i(A).$$

Proof. Let λ be a right eigenvalue of A. Then there exists some nonzero $x \in \mathbb{H}^n$ such that $Ax = x\lambda$. Let $x := [x_1, \ldots, x_n]^T \in \mathbb{H}^n$ and let x_t be an element of x such that $|x_t| \ge |x_i| \forall i \ (1 \le i \le n)$. From the t-th equation of $Ax = x\lambda$, we have

(2.3)
$$a_{tt}x_t + \sum_{j=1, j \neq t}^n a_{tj}x_j = x_t\lambda.$$

Since $a_{tt} \in \mathbb{R}$, $a_{tt}x_t = x_t a_{tt}$. Proceeding as in the proof of Theorem 2.2, we obtain

(2.4)
$$|\lambda - a_{tt}| \le \sum_{j=1, \, j \ne t}^{n} |a_{jt}| =: r_t(A)$$

From [47, Corollary 2.7], λ is also a right eigenvalue of A^{H} . Then

(2.5)
$$|\lambda - a_{tt}| \le \sum_{j=1, j \ne t}^{n} |a_{tj}| =: c_t(A).$$

Let $\gamma \in [0, 1]$. Then from (2.4) and (2.5), we obtain

(2.6)
$$|\lambda - a_{tt}|^{\gamma} \le r_t^{\gamma}(A),$$

(2.7)
$$|\lambda - a_{tt}|^{1-\gamma} \le c_t^{1-\gamma}(A).$$

Combining (2.6) and (2.7), we get

$$|\lambda - a_{tt}| \le r(A)_t^{\gamma} c(A)_t^{1-\gamma}. \blacksquare$$

Corollary 2.5. For any $A := (a_{ij}) \in M_n(\mathbb{H}), n \geq 2$ and for any $\gamma \in [0, 1]$. Assume that

(2.8)
$$|a_{ii}| > r_i(A)^{\gamma} r_i(A)^{1-\gamma} \quad \forall \ i \ (1 \le i \le n).$$

Then A is invertible.

Proof. On the contrary, suppose A is not invertible. Then by Theorem 1.31, there is a left eigenvalue $\lambda = 0$ of A. Now from Theorem 2.3, we obtain $|a_{ii}| \leq r_i(A)^{\gamma} c_i(A)^{1-\gamma}$. This contradicts our assumption (2.8). Hence A is invertible.

It is known that a matrix $A \in M_n(\mathbb{H})$ may have at most 2n complex right eigenvalues. From Theorem 2.4, all the complex right eigenvalues of a matrix $A = (a_{ij}) \in M_n(\mathbb{H})$ with all real diagonal entries lie in the union of *n*-discs

$$\mathcal{E}_i(A) := \{ z \in \mathbb{C} : |z - a_{ii}| \le r_i(A)^{\gamma} c_i(A)^{1-\gamma} \} \ (1 \le i \le n), \text{ i.e.,}$$

(2.9)
$$\Lambda_c(A) \subseteq \mathcal{E}(A) := \bigcup_{i=1}^n \mathcal{E}_i(A),$$

where $\Lambda_c(A) := \{\lambda \in \mathbb{C} : Ax = x\lambda, \ 0 \neq x \in \mathbb{H}^n\}.$

The Brauer type theorem is proved in [22] for the left eigenvalues in the case of deleted absolute column sums of a matrix $A \in M_n(\mathbb{H})$, i.e., if $\lambda \in \Lambda_l(A)$, then its conjugate $\overline{\lambda}$ lies in the union of $\frac{n(n-1)}{2}$ ovals of Cassini. However, this is incorrect as the following example suggest:

Example 2.6. Let $A = \begin{bmatrix} \mathbf{i} & \mathbf{k} \\ 0 & \mathbf{j} \end{bmatrix}$. Then by [22, Theorem 5], oval of Cassini is given by $\{z \in \mathbb{H} : |z - \mathbf{i}| | |z - \mathbf{j}| \le 0\}$. Here, \mathbf{i} is a left eigenvalue of A and its conjugate $-\mathbf{i}$ is not contained in the above oval of Cassini.

According to [22, Theorem 5], if $\lambda \in \Lambda_l(A)$, then $\overline{\lambda} \in \bigcup_{\substack{i,j=1,\\i\neq j}}^n F_{ij}(A)$, where

$$F_{ij}(A) := \{ z \in \mathbb{H} : |z - a_{ii}| \ |z - a_{jj}| \le c_i(A)c_j(A) \} \quad (1 \le i, j \le n; i \ne j)$$

However, this result is not necessarily true as

$$|\overline{\lambda} - a_{ii}| |\overline{\lambda} - a_{jj}| > c_i(A)c_j(A) \quad \forall i, j (1 \le i, j \le n; i \ne j)$$

which follows from Example 2.6. Now, we derive a corrected version of [22, Theorem 5] as follows.

Theorem 2.7. Let $A := (a_{ij}) \in M_n(\mathbb{H})$. Then all the left eigenvalues of A are located in the union of $\frac{n(n-1)}{2}$ ovals of Cassini

$$F_{ij}(A) := \{ z \in \mathbb{H} : |z - a_{ii}| \ |z - a_{jj}| \le c_i(A)c_j(A) \} \quad (1 \le i, j \le n; i \ne j), \ i.e.,$$
$$\Lambda_l(A) \subseteq F(A) := \bigcup_{\substack{i,j=1, \\ i \ne j}}^n F_{ij}(A).$$

Proof. Let λ be a left eigenvalue of A. Then by Lemma 2.1, $\overline{\lambda}$ is a left eigenvalue of A^H , so that there exists some nonzero $x \in \mathbb{H}^n$ such that $A^H x = \overline{\lambda} x$. Let $x := [x_1, \ldots, x_n]^T \in \mathbb{H}^n$ and let x_s be an element of x such that $|x_s| \ge |x_i| \forall i \ (1 \le i \le n)$. Then, $|x_s| > 0$. Clearly, if all the other elements of x are zero, then the required result holds.

Let x_s and x_t be two nonzero elements of x such that $|x_s| \ge |x_t| \ge |x_i| \forall i \ (1 \le i \le n, i \ne s)$. From the s-th equation of $A^H x = \overline{\lambda} x$, we have

$$\sum_{j=1}^{n} \overline{a_{js}} x_j = \overline{\lambda} x_s,$$

which implies

$$(\overline{\lambda} - \overline{a_{ss}})x_s = \sum_{j=1, \, j \neq s}^n \overline{a_{js}}x_j.$$

Thus

(2.10)
$$|\lambda - a_{ss}| \le \left(\frac{|x_t|}{|x_s|}\right) c_s(A).$$

Similarly, from $A^H x = \overline{\lambda} x$, we obtain

(2.11)
$$|\lambda - a_{tt}| \le \left(\frac{|x_s|}{|x_t|}\right) c_t(A).$$

Combining (2.10) and (2.11), we have

$$|\lambda - a_{ss}| |\lambda - a_{tt}| \le c_s(A)c_t(A).$$

Hence, all the left eigenvalues of A are located in the union of $\frac{n(n-1)}{2}$ ovals of Cassini $F_{ij}(A)$ $(1 \le i, j \le n, i \ne j)$.

Theorem 7 of [22] was stated for a central closed quaternionic matrix. Now we generalize this result for all quaternionic matrices as follows.

Theorem 2.8. Let $A := (a_{ij}) \in M_n(\mathbb{H})$ and let $\gamma \in [0, 1]$. Then all the left eigenvalues of A are located in the union of $\frac{n(n-1)}{2}$ ovals of Cassini

$$K_{ij}(A) := \left\{ z \in \mathbb{H} : |z - a_{ii}| \ |z - a_{jj}| \le r_i(A)^{\gamma} \ r_j(A)^{\gamma} \ c_i(A)^{1-\gamma} \ c_j(A)^{1-\gamma} \right\}$$

 $(1 \leq i, j \leq n; i \neq j), i.e.,$

$$\Lambda_l(A) \subseteq K(A) := \bigcup_{\substack{i,j=1\\i\neq j}}^n K_{ij}(A).$$

Proof. Let λ be a left eigenvalue of A. Then by [22, Theorem 4] and Theorem 2.7, for $\gamma \in [0, 1]$, we have

(2.12)
$$|\lambda - a_{ii}|^{\gamma} |\lambda - a_{jj}|^{\gamma} \le r_i(A)^{\gamma} r_j(A)^{\gamma} \quad (1 \le i, j \le n; i \ne j),$$

and

(2.13)
$$|\lambda - a_{ii}|^{1-\gamma} |\lambda - a_{jj}|^{1-\gamma} \le c_i(A)^{1-\gamma} c_j(A)^{1-\gamma} \quad (1 \le i, j \le n; i \ne j).$$

Combining (2.12) and (2.13), we have

$$|\lambda - a_{ii}| |\lambda - a_{jj}| \le r_i(A)^{\gamma} r_j(A)^{\gamma} c_i(A)^{1 - \gamma} c_j(A)^{1 - \gamma} \ (1 \le i, j \le n; i \ne j). \blacksquare$$

Corollary 2.9. For any $A := (a_{ij}) \in M_n(\mathbb{H}), n \geq 2$ and for any $\gamma \in [0, 1]$. Assume that

$$|a_{ii}||a_{jj}| > r_i(A)^{\gamma} r_j(A)^{\gamma} c_i(A)^{1-\gamma} c_j(A)^{1-\gamma} \quad \forall i, j \ (1 \le i, j \le n, \ i \ne j).$$

Then A is invertible.

Corollary 2.10. Let $A := (a_{ij}) \in M_n(\mathbb{H})$. Then all the left eigenvalues of A are located in the union of $\frac{n(n-1)}{2}$ ovals of Cassini

$$\Lambda_l(A) \subseteq \Phi(A) := \bigcup_{\substack{i,j=1\\i \neq j}}^n \left\{ z \in \mathbb{H} : |z - a_{ii}| \ |z - a_{jj}| \le \min\{r_i(A)r_j(A), \ c_i(A)c_j(A)\} \right\}.$$

Proof. Substituting $\gamma = 0, 1$ in Theorem 2.8, we obtain the following:

(a)
$$\Lambda_l(A) \subseteq E(A) := \bigcup_{\substack{i,j=1\\ i\neq j}}^n \{ z \in \mathbb{H} : |z - a_{ii}| |z - a_{jj}| \le c_i(A)c_j(A) \} .$$

(b)
$$\Lambda_l(A) \subseteq F(A) := \bigcup_{\substack{i,j=1\\i\neq j}}^{i\neq j} \{ z \in \mathbb{H} : |z - a_{ii}| |z - a_{jj}| \le r_i(A)r_j(A) \} .$$

Combining (a) and (b), we get the required result.

The following result provides better estimation compare to Theorem 2.4.

Theorem 2.11. Let $A := (a_{ij}) \in M_n(\mathbb{H})$ with $a_{ii} \in \mathbb{R}$ and let $\gamma \in [0, 1]$. Then all the right eigenvalues of A are located in the union of $\frac{n(n-1)}{2}$ ovals of Cassini $\mathcal{G}_{ij}(A) := \{z \in \mathbb{H} : |z - a_{ii}| |z - a_{jj}| \leq r_i(A)^{\gamma} r_j(A)^{\gamma} c_i(A)^{1-\gamma} c_j(A)^{1-\gamma} \}$ $(1 \leq i, j \leq n, i \neq j), i.e.,$

$$\Lambda_r(A) \subseteq \mathcal{G}(A) := \bigcup_{\substack{i,j=1\\i\neq j}}^n \mathcal{G}_{ij}(A).$$

Proof. Proof follows from the proof method of Theorem 2.8 and by applying [63, Theorem 4.1, Corollary 4.1]. ■

From Theorem 2.11, all the complex right eigenvalues of a matrix $A := (a_{ij}) \in M_n(\mathbb{H})$ with $a_{ii} \in \mathbb{R} \ \forall i \ (1 \le i \le n)$ are contained in the union of $\frac{n(n-1)}{n}$ ovals of Cassini $\mathcal{F}_{ij}(A) := \{z \in \mathbb{C} : |z - a_{ii}| \ |z - a_{jj}| \le r_i(A)^{\gamma} r_j(A)^{\gamma} \ c_i(A)^{1-\gamma} c_j(A)^{1-\gamma} \} \quad (1 \le i, j \le n, \ i \ne j), \text{ i.e.},$

(2.14)
$$\Lambda_c(A) \subseteq \mathcal{F}(A) := \bigcup_{\substack{i,j=1\\i\neq j}}^n \mathcal{F}_{ij}(A),$$

The following theorem shows that Theorem 2.8 is sharper than Theorem 2.3.

Theorem 2.12. Let $A := (a_{ij}) \in M_n(\mathbb{H})$ with $n \ge 2$ and let $\gamma \in [0, 1]$. Then

$$K(A) \subseteq T(A),$$

where G(A) and $\mathcal{G}(A)$ are defined in Theorem 2.3 and Theorem 2.8, respectively.

Proof. Let $z \in K_{ij}(A)$ and fix any i and j, $(1 \le i, j \le n, i \ne j)$. Then from Theorem 2.8, we have

(2.15)
$$|z - a_{ii}| |z - a_{jj}| \le r_i(A)^{\gamma} r_j(A)^{\gamma} c_i(A)^{1 - \gamma} c_j(A)^{1 - \gamma}.$$

Now the following two cases are possible.

Case 1: If $r_i(A)^{\gamma} r_j(A)^{\gamma} c_i(A)^{1-\gamma} c_j(A)^{1-\gamma} = 0$, then $z = a_{ii}$ or $z = a_{jj}$. However, from Theorem 2.3, we have $a_{ii} \in T_i(A)$ and $a_{jj} \in T_j(A)$. Thus $z \in T_i(A) \cup T_j(A)$.

Case 2: If $r_i(A)^{\gamma} r_j(A)^{\gamma} c_i(A)^1 - \gamma c_j(A)^1 - \gamma > 0$, then by (2.15)

(2.16)
$$\left(\frac{|z-a_{ii}|}{r_i(A)^{\gamma}c_i(A)^{1-\gamma}}\right)\left(\frac{|z-a_{jj}|}{r_j(A)^{\gamma}c_j(A)^{1-\gamma}}\right) \le 1.$$

As the left side of (2.16) cannot exceed unity, then one of the factors of the left side can be at most unity, i.e., $z \in T_i(A)$ or $z \in T_j(A)$. Hence $z \in T_i(A) \cup T_j(A)$. Thus

(2.17)
$$K_{ij} \subseteq T_i(A) \cup T_j(A).$$

From Theorem 2.3 and Theorem 2.8, we obtain

$$K(A) := \bigcup_{\substack{i,j=1\\i\neq j}}^{n} K_{ij}(A) \subseteq \bigcup_{\substack{i,j=1\\i\neq j}}^{n} \{T_i(A) \cup T_j(A)\} = \bigcup_{k=1}^{n} T_k(A) =: T(A). \blacksquare$$

Similarly, we have the following relation between Theorem 2.11 and Theorem 2.4.

Theorem 2.13. Let $A := (a_{ij}) \in M_n(\mathbb{H}), n \geq 2$ with $a_{ii} \in \mathbb{R}$ and let $\gamma \in [0, 1]$. Then

 $\mathcal{G}(A) \subseteq G(A),$

where G(A) and $\mathcal{G}(A)$ are defined in Theorem 2.4 and Theorem 2.11, respectively.

Proof. The proof is immediate from the proof method of Theorem 2.12 and by applying Theorems 2.4 and 2.11. ■

The following example verifies Theorem 2.13 for complex right eigenvalues of a matrix $A := (a_{ij}) \in M_n(\mathbb{H})$ with $a_{ii} \in \mathbb{R} \ \forall i \ (1 \le i \le n)$.

Example 2.14. Let $A = \begin{bmatrix} 3 & 1+\mathbf{i}+\mathbf{j}-\mathbf{k} & 2+3\mathbf{j}-\sqrt{3}\mathbf{k} \\ 5+\sqrt{2}\mathbf{j}+3\mathbf{k} & -2 & 3\mathbf{j}+4\mathbf{k} \\ 4+3\mathbf{j} & 2-\mathbf{i}-2\mathbf{k} & -5 \end{bmatrix}$. Substituting $\gamma = 1/4$ in (2.9), we get the following three discs:

 $\begin{aligned} \mathcal{E}_1(A) &:= \{ z \in \mathbb{C} : |z-3| \le 9.4533 \}, \\ \mathcal{E}_2(A) &:= \{ z \in \mathbb{C} : |z+2| \le 6.0894 \}, \\ \mathcal{E}_3(A) &:= \{ z \in \mathbb{C} : |z+5| \le 8.7389 \}. \end{aligned}$

Similarly, let $\gamma = 1/4$ in (2.14), we get the following three ovals of Cassini:

$$\mathcal{F}_{12}(A) := \{ z \in \mathbb{C} : |z-3| |z+2| \le 57.5649 \},$$

$$\mathcal{F}_{23}(A) := \{ z \in \mathbb{C} : |z+2| |z+5| \le 53.2145 \},$$

$$\mathcal{F}_{31}(A) := \{ z \in \mathbb{C} : |z+5| |z-3| \le 82.6108 \}.$$

In this example, there are six complex right eigenvalues λ_j $(1 \le j \le 6)$ which are shown in FIGURE 2.1. The set $\mathcal{F}(A) := \mathcal{F}_{12}(A) \cup \mathcal{F}_{23}(A) \cup \mathcal{F}_{31}(A)$ is represented by shaded region in FIGURE 2.1. From FIGURE 2.1, it is clear that $\mathcal{F}(A) \subset \mathcal{E}(A)$, where $\mathcal{E}(A) :=$ $\mathcal{E}_1(A) \cup \mathcal{E}_2(A) \cup \mathcal{E}_3(A)$.

For $A := (a_{ij}) \in M_n(\mathbb{H})$, define

$$n_i^{(p)}(A) := \left(\sum_{j=1, \ j \neq i}^n |a_{ij}|^p\right)^{\frac{1}{p}} \quad (1 \le i \le n); \ p \in (1, \infty).$$

We are now ready to derive the following localization theorem for left eigenvalues of a quaternionic matrix.



FIGURE 2.1. Location of complex right eigenvalues of A (Example 2.14) as +.

Theorem 2.15. Let $A := (a_{ij}) \in M_n(\mathbb{H})$ and let $\gamma \in [0, 1]$. Then all the left eigenvalues of A are contained in the union of n generalized balls

$$B_{i}(A) := \left\{ z \in \mathbb{H} : |z - a_{ii}| \le (n-1)^{\frac{1-\gamma}{q}} r_{i}(A)^{\gamma} (n_{i}^{(p)}(A))^{1-\gamma} \right\} \quad (1 \le i \le n), \ i.e.,$$
$$\Lambda_{l}(A) \subseteq B(A) := \cup_{i=1}^{n} B_{i}(A),$$
for any $p, q \in (1, \infty)$ with $\frac{1}{p} + \frac{1}{q} = 1.$

Proof. Let μ be a left eigenvalue of A. Then there exists some nonzero $x \in \mathbb{H}^n$ such that $Ax = \mu x$. Let $x := [x_1, \ldots, x_n]^T \in \mathbb{H}^n$ and let x_t be an element of x such that $|x_t| \ge |x_i| \forall i \ (1 \le i \le n)$. Then from $Ax = \mu x$, we have

$$a_{tt}x_t + \sum_{j=1,\,j\neq t}^n a_{tj}x_j = \mu x_t$$

this implies

(2.18)
$$|\mu - a_{tt}||x_t| = \left|\sum_{j=1, j \neq t}^n a_{tj}x_j\right| \le \sum_{j=1, j \neq t}^n |a_{tj}| \ |x_j|.$$

Applying the generalized Hölder inequality to (2.18), we have

$$|\mu - a_{tt}||x_t| \leq \left(\sum_{j=1, j \neq t}^n |a_{tj}|^p\right)^{\frac{1}{p}} \left(\sum_{j=1, j \neq t}^n |x_j|^q\right)^{\frac{1}{q}}.$$
34

Since $|x_t| \ge |x_i| \forall i \ (1 \le i \le n)$, we have

$$|\mu - a_{tt}||x_t| \leq n_t^{(p)}(A) \left((n-1)|x_t|^q\right)^{\frac{1}{q}}$$
, i.e.,

(2.19)
$$|\mu - a_{tt}| \le n_t^{(p)}(A) (n-1)^{\frac{1}{q}}.$$

Similarly, using $|x_t| \ge |x_i| \forall i \ (1 \le i \le n)$ in (2.18), we get

(2.20)
$$|\mu - a_{tt}| \le \sum_{j=1, \, j \ne t}^{n} |a_{tj}| = r_t(A).$$

Combining (2.19) and (2.20) for $\gamma \in [0, 1]$, we have

(2.21)
$$|\mu - a_{tt}|^{1-\gamma} \le (n_t^{(p)}(A))^{1-\gamma}(n-1)^{\frac{1-\gamma}{q}} \text{ and } |\mu - a_{tt}|^{\gamma} \le r_t(A)^{\gamma}, \text{ i.e.,}$$

 $|\mu - a_{tt}| \le (n-1)^{\frac{1-\gamma}{q}}(n_t^{(p)}(A))^{1-\gamma}r_t(A)^{\gamma}.$

Now, we present the following results from the literature:

• Assuming p = q = 2 and $\gamma = 1$ in Theorem 2.15. We obtain that all the left eigenvalues of $A := (a_{ij}) \in M_n(\mathbb{H})$ are contained in the union of n Greschgorin balls $B_i(A) := \{z \in \mathbb{H} : |z - a_{ii}| \le r_i(A)\}$ $(1 \le i \le n)$, i.e.,

$$\Lambda_l(A) \subseteq B(A) := \bigcup_{i=1}^n B_i(A).$$

This can be found in [62, Theorem 6].

• Assuming p = q = 2 and $\gamma = 0$ in Theorem 2.15. We obtain that all the left eigenvalues of $A := (a_{ij}) \in M_n(\mathbb{H})$ are contained in the union of n balls $B_i(A) := \left\{z \in \mathbb{H} : |z - a_{ii}| \le (n-1)^{\frac{1}{2}} n_i^{(2)}(A)\right\} (1 \le i \le n)$, i.e.,

$$\Lambda_l(A) \subseteq B(A) := \bigcup_{i=1}^n B_i(A).$$

This can be seen in [60, Theorem 1].

We present a generalization of [62, Theorem 7] and [63, Theorem 3.1] by applying the generalized Hölder inequality over the skew field of quaternions. For a general matrix $A := (a_{ij}) \in M_n(\mathbb{H})$, all the right eigenvalues may not lie in the union of n generalized balls $B_i(A)$ $(1 \le i \le n)$. On the other hand, we show that every connected region of the generalized balls $B_i(A)$ $(1 \le i \le n)$ contains some right eigenvalues of A. **Theorem 2.16.** Let $A := (a_{ij}) \in M_n(\mathbb{H})$ and let $\gamma \in [0,1]$. For every right eigenvalue μ of A there exists a nonzero quaternion β such that $\beta^{-1}\mu\beta$ (which is also a right eigenvalue) is contained in the union of n generalized balls

$$B_{i}(A) := \left\{ z \in \mathbb{H} : |z - a_{ii}| \le (n-1)^{\frac{1-\gamma}{q}} r_{i}(A)^{\gamma} (n_{i}^{(p)}(A))^{1-\gamma} \right\} \quad (1 \le i \le n), \ i.e.,$$
$$\left\{ z^{-1} \mu z : 0 \ne z \in \mathbb{H} \right\} \cap \cup_{i=1}^{n} B_{i}(A) \ne \emptyset,$$

where $p, q \in (1, \infty)$ with $\frac{1}{p} + \frac{1}{q} = 1$.

Proof. Let μ be a right eigenvalue of A. Then there exists some nonzero $x \in \mathbb{H}^n$ such that $Ax = x\mu$. Let $x := [x_1, \ldots, x_n]^T \in \mathbb{H}^n$ and choose x_t from x as given in Theorem 2.15. Consider $\rho \in \mathbb{H}$ such that $x_t\mu = \rho x_t$. Then we have

(2.22)
$$|\rho - a_{tt}| |x_t| = \left| \sum_{j=1, j \neq t}^n a_{tj} x_j \right| \le \sum_{j=1, j \neq t}^n |a_{tj}| |x_j|.$$

Now from the proof method of Theorem 2.15, we have

$$|\rho - a_{tt}| \le (n-1)^{\frac{1-\gamma}{q}} (n_t^{(p)}(A))^{1-\gamma} r_t(A)^{\gamma}.$$

We can see the following results from the literature:

• Substituting p = q = 2 and $\gamma = 1$ in Theorem 2.16, we obtain

$$\{z^{-1}\mu z: 0 \neq z \in \mathbb{H}\} \cap \bigcup_{i=1}^n \{z \in \mathbb{H}: |z - a_{ii}| \le r_i(A)\} \neq \emptyset.$$

This can be seen in [62, Theorem 7].

• Substituting p = q = 2 and $\gamma = 0$ in Theorem 2.16, we get

$$\{z^{-1}\mu z : 0 \neq z \in \mathbb{H}\} \cap \bigcup_{i=1}^{n} \{z \in \mathbb{H} : |z - a_{ii}| \le \sqrt{n-1} \ n_{i}^{(2)}(A)\} \neq \emptyset.$$

This can be found in [63, Theorem 3.1].

We next present a sufficient condition for the stability of a matrix $A \in M_n(\mathbb{H})$ for the case of a continuous-time quaternionic system.

Proposition 2.17. Let $A := (a_{ij}) \in M_n(\mathbb{H})$ and let $\gamma \in [0, 1]$. Assume that

(2.23)
$$Re(a_{ii}) + (n-1)^{\frac{1-\gamma}{q}} r_i(A)^{\gamma} (n_i^{(p)}(A))^{1-\gamma} < 0 \quad \forall \ i \ (1 \le i \le n),$$

where $\frac{1}{p} + \frac{1}{q} = 1$ with $p, q \in (1, \infty)$. Then the matrix A is stable.

Proof. Let $\lambda \in \Lambda_r(A)$. Then from Theorem 2.16 there exists $0 \neq \rho \in \mathbb{H}$ such that $\rho^{-1}\lambda\rho \in \bigcup_{i=1}^n B_i(A)$. Without loss of generality, we assume $\rho^{-1}\lambda\rho \in B_l(A)$, i.e.,

$$|\rho^{-1}\lambda\rho - a_{ll}| \le (n-1)^{\frac{1-\gamma}{q}} r_l(A)^{\gamma} (n_l^{(p)}(A))^{1-\gamma}.$$

Consider $\lambda := \lambda_1 + \lambda_2 \mathbf{i} + \lambda_3 \mathbf{j} + \lambda_4 \mathbf{k}$ and $a_{ll} := a_l + b_l \mathbf{i} + c_l \mathbf{j} + d_l \mathbf{k}$. Then from (2.23), we obtain

(2.24)
$$|(\lambda_1 - a_l) + (\rho^{-1}\lambda_2\mathbf{i}\rho - b_l\mathbf{i}) + (\rho^{-1}\lambda_3\mathbf{j}\rho - c_l\mathbf{j}) + \xi_1| < -\operatorname{Re}(a_{ll}) = -a_l$$

where $\xi_1 = \rho^{-1}\lambda_4 \mathbf{k}\rho - d_l \mathbf{k}$. (2.24) is possible when $\lambda_1 < 0$, i.e., $\operatorname{Re}(\lambda) < 0$, hence $\lambda \in \mathbb{H}^-$. This shows that the matrix A is stable.

When all the diagonal entries of a matrix $A \in M_n(\mathbb{H})$ are real, then we have the following theorem.

Theorem 2.18. Let $A := (a_{ij}) \in M_n(\mathbb{H})$ with $a_{ii} \in \mathbb{R}$ and let $\gamma \in [0, 1]$. Then all the right eigenvalues of A are contained in the union of n generalized balls

$$B_i(A) := \left\{ z \in \mathbb{H} : |z - a_{ii}| \le (n-1)^{\frac{1-\gamma}{q}} r_i(A)^{\gamma} (n_i^{(p)}(A))^{1-\gamma} \right\} \quad (1 \le i \le n), \ i.e.,$$
$$\Lambda_r(A) \subseteq B(A) := \cup_{i=1}^n B_i(A),$$

where $p, q \in (1, \infty)$ with $\frac{1}{p} + \frac{1}{q} = 1$.

Proof. Let λ be a right eigenvalue of A. Then there exists some nonzero $x \in \mathbb{H}^n$ such that $Ax = x\lambda$. Let $x := [x_1, \ldots, x_n]^T \in \mathbb{H}^n$ and let x_t be an element of x such that $|x_t| \ge |x_i| \forall i \ (1 \le i \le n)$. Then $|x_t| > 0$. Thus from $Ax = x\lambda$, we have

$$a_{tt}x_t + \sum_{j=1, \, j \neq t}^n a_{tj}x_j = x_t\lambda.$$

Since $a_{tt} \in \mathbb{R}$, so $a_{tt}x_t = x_t a_{tt}$. Then from the proof method of Theorem 2.15, we have

$$|\lambda - a_{tt}| \le (n-1)^{\frac{1-\gamma}{q}} (n_t^{(p)}(A))^{1-\gamma} r_t(A)^{\gamma}.$$

The above result has great significance as Hermitian and η -Hermitian matrices have all real diagonal entries. In general, η -Hermitian matrices arise widely in applications [17, 55, 56]. To that end, we state the following proposition when all diagonal entries of $A \in M_n(\mathbb{H})$ are real. In particular, this result gives a sufficient condition for the stability of a matrix $A \in M_n(\mathbb{H})$. **Proposition 2.19.** Let $A := (a_{ij}) \in M_n(\mathbb{H})$ with $a_{ii} \in \mathbb{R}$ and let $\gamma \in [0, 1]$. Assume that

$$a_{ii} + (n-1)^{\frac{1-\gamma}{q}} r_i(A)^{\gamma} (n_i^{(p)}(A))^{1-\gamma} < 0 \ \forall i \ (1 \le i \le n),$$

where $p, q \in (1, \infty)$ with $\frac{1}{p} + \frac{1}{q} = 1$. Then the matrix A is stable.

From Theorem 2.18, all the complex right eigenvalues of a matrix $A = (a_{ij}) \in M_n(\mathbb{H})$ with all real diagonal entries lie in the union of n-discs

$$D_i(A) := \{ z \in \mathbb{C} : |z - a_{ii}| \le (n-1)^{\frac{1-\gamma}{q}} r_i(A)^{\gamma} (n_i^{(p)}(A))^{1-\gamma} \} \ (1 \le i \le n), \text{ i.e.},$$

(2.25)
$$\Lambda_c(A) \subseteq D(A) := \bigcup_{i=1}^n D_i(A).$$

However, if diagonal entries are from $\mathbb{C} \setminus \mathbb{R}$, then it is not necessary that all the complex right eigenvalues of A are contained in the union of n-discs $D_i(A)$ $(1 \le i \le n)$ as the following examples suggest.

Example 2.20. Let
$$A := \begin{bmatrix} 1-2\mathbf{i} & \mathbf{j} & \mathbf{k} \\ 0 & -2\mathbf{i} & -\mathbf{i} \\ 0 & \mathbf{k} & 3+\mathbf{i} \end{bmatrix}$$
. Then the set of complex right eigenvalues of A is given by

ues of A is given by

$$\Lambda_c(A) := \{\lambda_1, \lambda_2, \lambda_3, \lambda_4, \lambda_5, \lambda_6\}$$

where $\lambda_1 = -0.0164 + 2.0083\mathbf{i}, \lambda_2 = -0.0164 - 2.0083\mathbf{i}, \lambda_3 = 1 + 2\mathbf{i}, \lambda_4 = 1 - 2\mathbf{i}, \lambda_5 = -0.0164 - 2.0083\mathbf{i}, \lambda_5 = -0.0164 + 2.0083\mathbf{i}, \lambda_5 = -0.0083\mathbf{i}, \lambda_5 = -0.0083\mathbf{$ 3.0164 + 1.0324i, and $\lambda_6 = 3.0164 + 1.0324i$. For $\gamma = 1$ in (2.25), the discs $D_1(A), D_2(A)$, and $D_3(A)$ are given as follows:

$$D_1(A) := \{ z \in \mathbb{C} : |z - 1 + 2\mathbf{i}| \le 2 \},$$

$$D_2(A) := \{ z \in \mathbb{C} : |z + 2\mathbf{i}| \le 1 \}, \text{ and}$$

$$D_3(A) := \{ z \in \mathbb{C} : |z - 3 - \mathbf{i}| \le 1 \}.$$

From FIGURE 2.2, it is clear that λ_1, λ_3 , and λ_6 lie outside of the discs $D_1(A), D_2(A)$, and $D_3(A)$.

Example 2.21. Let
$$A = \begin{bmatrix} -4 & 1+\mathbf{j}+\sqrt{2}\mathbf{k} & \mathbf{j} \\ \mathbf{i}+\mathbf{j} & -10 & 2\mathbf{j}-\mathbf{k} \\ \mathbf{i}-2\mathbf{j}+2\mathbf{k} & \sqrt{3}+2\mathbf{j}-3\mathbf{k} & -8 \end{bmatrix}$$
. In this example, there

are six complex right eigenvalues λ_j (1 $\leq j \leq 6$) which are shown in FIGURE 2.3.



FIGURE 2.2. Location of complex right eigenvalues of A (Example 2.20) as +.

Substituting $\gamma = 1$ in (2.25), then all the complex right eigenvalues of the matrix A are contained in the union of three discs $D_1(A), D_2(A)$, and $D_3(A)$, where

$$D_1(A) := \{ z \in \mathbb{C} : |z+4| \le 3 \},$$

$$D_2(A) := \{ z \in \mathbb{C} : |z+10| \le \sqrt{2} + \sqrt{5} \}, \text{ and}$$

$$D_3(A) := \{ z \in \mathbb{C} : |z+8| \le 7 \}.$$

From FIGURE 2.3, the standard right eigenvalues of A are λ_1 , λ_3 , and λ_5 . Thus

$$\Lambda_r(A) = [\lambda_1] \cup [\lambda_3] \cup [\lambda_5].$$

Also, from FIGURE 2.3, we observe that $\operatorname{Re}(\lambda_i) \in \mathbb{H}^-$ (i = 1, 3, 5). Hence

$$\operatorname{Re}(\lambda_1) = \operatorname{Re}(\rho^{-1}\lambda_1\rho), \ \operatorname{Re}(\lambda_2) = \operatorname{Re}(\tau^{-1}\lambda_2\tau), \ \operatorname{Re}(\lambda_3) = \operatorname{Re}(\nu^{-1}\lambda_3\nu) \quad \forall \rho, \tau, \nu \in \mathbb{H}$$

Thus the matrix A is stable.

In general, similar quaternionic matrices may not have the same left eigenvalues, see, [62, Example 3.3]. However, the following result is true.

Proposition 2.22. Let $A \in M_n(\mathbb{H})$ and let W be any invertible real matrix. Then A and WAW^{-1} have the same left eigenvalues.



FIGURE 2.3. Location of complex right eigenvalues of A (Example 2.21) as +.

Proof. Let λ be a left eigenvalue of A. Then there exists some nonzero $x \in \mathbb{H}^n$ such that $Ax = \lambda x$. Let W be an invertible real matrix. Then

$$WAx = W\lambda x = \lambda Wx.$$

Now, $WAW^{-1}Wx = \lambda Wx$. Putting Wx = y. Then $WAW^{-1}y = \lambda y$.

Let $A := (a_{ij}) \in M_n(\mathbb{H})$. Suppose $W = \text{diag}(w_1, w_2, \dots, w_n)$ with $w_i \in \mathbb{R}^+$ $(1 \le i \le n)$. Then

$$W^{-1}AW = \left(\frac{a_{ij}w_j}{w_i}\right)$$
 and $\Lambda_l(A) = \Lambda_l(W^{-1}AW).$

Define

$$r_i^W(A) := \sum_{j=1, j \neq i}^n \frac{|a_{ij}|w_j}{w_i} \text{ and } c_i^W(A) := \sum_{j=1, j \neq i}^n \frac{|a_{ji}|w_i}{w_j} \quad (1 \le i \le n).$$

Applying Theorem 2.3 to $W^{-1}AW$, we get the following theorem which may be sharper than Theorem 2.3 depending upon the choice of W.

Theorem 2.23. Let $A := (a_{ij}) \in M_n(\mathbb{H})$. Then all the left eigenvalues of A are contained in the union of n balls

$$T_i^W(A) := \{ z \in \mathbb{H} : |z - a_{ii}| \le (r_i^W(A))^{\gamma} (c_i^W(A))^{1 - \gamma} \} \quad (1 \le i \le n), \ i.e.,$$
$$\Lambda_l(A) = \Lambda_l(W^{-1}AW) \subseteq T^W(A) := \cup_{i=1}^n T_i^W(A).$$

Since the above theorem holds for every $W = \text{diag}(w_1, w_2, \ldots, w_n)$, where $w_i \in \mathbb{R}^+$, we have

$$\Lambda_l(A) = \Lambda_l(W^{-1}AW) \subseteq \bigcap_{W \in M_n(S)} T^W(A) =: T^S(A),$$

where $M_n(S)$ is a set of invertible real diagonal matrices and $T^S(A)$ is called the minimal Ostrowski type set for the matrix A.

Substituting $\gamma = 1$ in Theorem 2.23, we obtain the following.

(a)
$$\Lambda_l(A) = \Lambda_l(W^{-1}AW) \subseteq \eta^W(A) := \bigcup_{i=1}^n \eta^W_i(A),$$

where $\eta_i^W(A) := \left\{ z \in \mathbb{H} : |z - a_{ii}| \le r_i^W(A) \right\}$. Now from here, we get

$$\Lambda_l(A) = \Lambda_l(W^{-1}AW) \subseteq \bigcap_{W \in M_n(S)} \eta^W(A) =: \eta^S(A),$$

where $\eta^{S}(A)$ is called the first minimal Gerschgorin type set for the matrix A.

For $\gamma = 0$ in Theorem 2.23, we have the following.

(**b**)
$$\Lambda_l(A) = \Lambda_l(W^{-1}AW) \subseteq \Omega^W(A) := \bigcup_{i=1}^n \Omega_i^W(A),$$

where $\Omega_i^W(A) := \left\{ z \in \mathbb{H} : |z - a_{ii}| \le c_i^W(A) \right\}$. Then

$$\Lambda_l(A) = \Lambda_l(W^{-1}AW) \subseteq \bigcap_{W \in M_n(S)} \Omega^W(A) =: \Omega^S(A),$$

where $\Omega^{S}(A)$ is called the second minimal Gerschgorin type set for the matrix A.

Equivalently, applying Theorem 2.8 to $W^{-1}AW$, we get the following theorem.

Theorem 2.24. Let $A := (a_{ij}) \in M_n(\mathbb{H})$ and let $\gamma \in [0, 1]$. Then all the left eigenvalues of A are contained in the union of $\frac{n(n-1)}{2}$ ovals of Cassini $K_{ij}^W(A) := \{z \in \mathbb{H} : |z - a_{ii}| |z - a_{jj}| \le (r_i^W(A))^{\gamma} (r_j^W(A))^{\gamma} (c_i^W(A))^{1-\gamma} (c_j^W(A))^{1-\gamma}\}$ $(1 \le i, j \le n; i \ne j), i.e.,$

$$\Lambda_l(A) = \Lambda_l(W^{-1}AW) \subseteq K^W(A) := \bigcup_{\substack{i,j=1\\i \neq j}}^n K^W_{ij}(A).$$

Since Theorem 2.24 holds for every $W = \text{diag}(w_1, w_2, \ldots, w_n)$ with $w_i \in \mathbb{R}^+$. Then

$$\Lambda_l(A) = \Lambda_l(W^{-1}AW) \subseteq \bigcap_{W \in M_n(S)} K^W(A) =: K^S(A),$$

 $K^{S}(A)$ is called the minimal Brauer type set for the matrix A.

Example 2.25. Let $A = \begin{bmatrix} \mathbf{j} & \mathbf{k} & 2\mathbf{j} + \sqrt{5}\mathbf{k} \\ 0 & \mathbf{i} + \mathbf{k} & \sqrt{2}\mathbf{i} + \mathbf{j} - \mathbf{k} \\ 0 & 0 & 2 - \mathbf{i} \end{bmatrix}$. Substituting $\gamma = 1$ in Theorem 2.3, we have the following three Gerschgorin type balls:

e have the following three Gerschgorin type bans.

 $G_1(A) := \{ z \in \mathbb{H} : |z - \mathbf{j}| \le 4 \},$ $G_2(A) := \{ z \in \mathbb{H} : |z - \mathbf{i} - \mathbf{k}| \le 2 \}, \text{ and}$ $G_3(A) := \{ z \in \mathbb{H} : |z - \mathbf{2} + \mathbf{i}| \le 0 \}.$

If $W = \text{diag}(w_1, w_2, w_3)$ with $w_1 = 8$, $w_2 = 4$, $w_3 = 1$. Then by (a), we have the following balls:

$$\begin{split} \eta_1^W(A) &:= \{ z \in \mathbb{H} : |z - \mathbf{j}| \le 7/8 \}, \\ \eta_2^W(A) &:= \{ z \in \mathbb{H} : |z - \mathbf{i} - \mathbf{k}| \le 1/2 \}, \text{ and} \\ \eta_3^W(A) &:= \{ z \in \mathbb{H} : |z - \mathbf{2} + \mathbf{i}| \le 0 \}. \end{split}$$

Hence it is clear that $\eta_1^W(A) \subset G_1(A)$ and $\eta_2^W(A) \subset G_2(A)$.

For $\gamma = 1$, Theorem 2.8 gives the following ovals of Cassini:

$$K_{12}(A) := \{ z \in \mathbb{H} : |z - \mathbf{j}| | z - \mathbf{i} - \mathbf{k}| \le 8 \},$$

$$K_{23}A) := \{ z \in \mathbb{H} : |z - \mathbf{i} - \mathbf{k}| | z - \mathbf{2} + \mathbf{i}| \le 0 \}, \text{ and}$$

$$K_{31}(A) := \{ z \in \mathbb{H} : |z - \mathbf{2} + \mathbf{i}| | z - \mathbf{j}| \le 0 \}.$$

Consider $W = \text{diag}(w_1, w_2, w_3)$ with $w_1 = w_2 = 6$, and $w_3 = 1$. Then by Theorem 2.24 with $\gamma = 1$, we obtain the following ovals of Cassini:

$$\begin{split} K_{12}^{W}(A) &:= \{ z \in \mathbb{H} : |z - \mathbf{j}| \, |z - \mathbf{i} - \mathbf{k}| \leq 1/2 \}, \\ K_{23}^{W}(A) &:= \{ z \in \mathbb{H} : |z - \mathbf{i} - \mathbf{k}| \, |z - \mathbf{2} + \mathbf{i}| \leq 0 \}, \text{ and} \\ K_{31}^{W}(A) &:= \{ z \in \mathbb{H} : |z - \mathbf{2} + \mathbf{i}| \, |z - \mathbf{j}| \leq 0 \}. \end{split}$$

Hence $K_{12}^W(A) \subset K_{12}(A)$.

2.3. Bounds for the zeros of quaternionic polynomials

First, in this section, we present bounds for the zeros of quaternionic polynomial $p_l(z)$ as follows, which is an extension of the result given in [15] for the case of the zeros of complex polynomials.

Theorem 2.26. Let $p_l(z)$ be a simple monic polynomial over \mathbb{H} of degree m. Then every zero \tilde{z} of $p_l(z)$ satisfies the following inequality:

$$\left(\max_{1 \le i \le m} \left(r_i'(C_{q_l})^{\gamma} c_i'(C_{q_l})^{1-\gamma} \right) \right)^{-1} \le |\tilde{z}| \le \max_{1 \le i \le m} \left(r_i'(C_{p_l})^{\gamma} c_i'(C_{p_l})^{1-\gamma} \right),$$

for every $\gamma \in [0, 1]$.

Proof. From Proposition 1.36, zeros of $p_l(z)$ and left eigenvalues of C_{p_l} are same. Thus, if \tilde{z} is a zero of $p_l(z)$, then \tilde{z} is a left eigenvalue of C_{p_l} . By applying Theorem 2.3 (Ostrowski type theorem) to C_{p_l} , we obtain

$$|\tilde{z}| \leq \max_{1 \leq i \leq m} \left(r'_i(C_{p_l})^{\gamma} c'_i(C_{p_l})^{1-\gamma} \right).$$

Proof for lower bounds: We use the respective upper bounds for the zeros of the simple monic reversal polynomial $q_l(z)$ for the desired lower bounds for the zeros of $p_l(z)$.

Corollary 2.27. Let $p_l(z)$ be a simple monic polynomial over \mathbb{H} of degree m. Then every zero \tilde{z} of $p_l(z)$ satisfies the following inequalities:

$$1. \quad \frac{|q_0|}{\max_{1 \le i \le (m-1)} \{1, |q_0| + |q_i|\}} \le |\tilde{z}| \le \max_{1 \le i \le (m-1)} \{|q_0|, 1 + |q_i|\}.$$

$$2. \quad \frac{|q_0|}{\max\left\{|q_0|, 1 + \sum_{i=1}^{m-1} |q_i|\right\}} \le |\tilde{z}| \le \max\left\{1, \sum_{i=0}^{m-1} |q_i|\right\}.$$

Proof. Substituting $\gamma = 0, 1$ in Theorem 2.26, we obtain the desired results.

Next, we derive the following lemma which gives a better bound than G. Opfer's bound [42, Theorem 4.2] for $|q_0| \ge 1$.

Lemma 2.28. Assume that $|q_0| \ge 1$. Then $\alpha \le \mathcal{T}$, where $\alpha := \max_{1 \le i \le m-1} \{ |q_0|, 1 + |q_i| \}$ and $\mathcal{T} := \max \{ 1, \sum_{i=0}^{m-1} |q_i| \}.$

Proof. Case 1: If $|q_0| = 1$, then

$$\alpha = \max_{1 \le i \le m-1} \left\{ |q_0|, 1 + |q_i| \right\} = \max_{1 \le i \le m-1} \left\{ 1 + |q_i| \right\}.$$
 Also

 $\mathcal{T} := \max\left\{1, \sum_{i=0}^{m-1} |q_i|\right\} = \max\left\{1, |q_0| + \sum_{i=1}^{m-1} |q_i|\right\} = 1 + \sum_{i=1}^{m-1} |q_i|.$ **Case 2:** If $|q_0| > 1$, then

$$\alpha = \max_{1 \le i \le (m-1)} \{ |q_0|, 1 + |q_i| \} = |q_0| \text{ or } \max_{1 \le i \le (m-1)} \{ 1 + |q_i| \} \text{ and}$$

 $\mathcal{T} := \max\{1, \sum_{i=0}^{m-1} |q_i|\} = \max\{1, |q_0| + \sum_{i=1}^{m-1} |q_i|\} = |q_0| + \sum_{i=1}^{m-1} |q_i|. \text{ Thus } \alpha \leq \mathcal{T}.$ This completes the proof. \blacksquare

On the other hand, if $|q_0| < 1$, then $\alpha \leq \mathcal{T}$ or $\alpha > \mathcal{T}$ follows from a simple monic polynomial $p'_l(z) := z^3 + (\mathbf{i} + 2\mathbf{j} + 2\mathbf{k})z^2 - 2\mathbf{k}z + 0.5\mathbf{k}$. Then $\alpha = 4$ and $\mathcal{T} = 5.5$. Hence $\alpha < \mathcal{T}$ and further, if we consider $p_l''(z) = z^3 + +0.5\mathbf{j}z^2 + (0.2\mathbf{i}+0.3\mathbf{j})z + 0.5\mathbf{i}$. Then $\alpha = 1.5$ and $\mathcal{T} = 1.36$. Hence $\alpha > \mathcal{T}$.

Theorem 2.29. Let $w_i \in \mathbb{R}^+$ $(1 \leq i \leq m)$ and let $\gamma \in [0,1]$. Then every zero \tilde{z} of the simple monic polynomial $p_l(z)$ satisfies the following inequality:

$$\left[\max_{1 \le i \le m} \left\{ r'_i (WC_{q_l} W^{-1})^{\gamma} c'_i (WC_{q_l} W^{-1})^{1-\gamma} \right\} \right]^{-1} \le |\tilde{z}| \le \xi_1,$$

where $\xi_1 = \max_{1 \le i \le m} \{ r'_i (WC_{p_l} W^{-1})^{\gamma} c'_i (WC_{p_l} W^{-1})^{1-\gamma} \}$ and $W := \operatorname{diag}(w_1, w_2, \dots, w_m).$

Proof. The companion matrix of $p_l(z)$ is given by

$$C_{p_l} = \prod_{1}^{m-1} \begin{bmatrix} 0 & I \\ \hline -q_0 & [-q_1 \dots - q_{m-1}] \end{bmatrix}.$$

Then

$$WC_{p_l}W^{-1} = \prod_{1}^{m-1} \begin{bmatrix} 0 & \text{diag}\left(\frac{w_1}{w_2}, \dots, \frac{w_{m-1}}{w_m}\right) \\ -\frac{w_m}{w_1}q_0 & -\frac{w_m}{w_2}q_1\dots - q_{m-1} \end{bmatrix}.$$

By Proposition 2.22, C_{p_l} and $WC_{p_l}W^{-1}$ have the same left eigenvalues. Rest of the proof follows from the proof method of Theorem 2.26. \blacksquare

Corollary 2.30. Let $p_l(z)$ be a simple monic polynomial over \mathbb{H} of degree m. Then every zero \tilde{z} of $p_l(z)$ satisfies the following inequalities:

$$1. \left[\max_{\substack{0 \le j \le m-1 \\ 0 \le j \le m-1 \\ 0.}} \left\{ \frac{(|q_0|w_j + w_m|q_{m-j}|)}{|q_0|d_{j+1}} \right\} \right]^{-1} \le |\tilde{z}| \le \max_{\substack{0 \le j \le m-1 \\ 0 \le j \le m-1 \\ 1 \le j \le m-1 \\ 0.}} \left\{ \frac{w_j}{w_{j+1}}, \sum_{i=0}^{m-1} \frac{w_m|q_i|}{|q_0|w_{i+1}} \right\} \right]^{-1} \le |\tilde{z}| \le \max_{\substack{1 \le j \le m-1 \\ 1 \le j \le m-1 \\ 0.}} \left\{ \frac{w_j}{w_{j+1}}, \sum_{i=0}^{m-1} \frac{w_m|q_i|}{w_{i+1}} \right\}.$$

Proof. Substituting $\gamma = 0, 1$ in Theorem 2.29, we get the desired results.

Let $w_j = w_m |q_j| (1 \le j \le m - 1)$ in the part (1) of Corollary 2.30. Then we obtain

$$|\tilde{z}| \leq \max_{1 \leq j \leq m-1} \left\{ \left| \frac{q_0}{q_1} \right|, 2 \left| \frac{q_j}{q_{j+1}} \right| \right\}.$$

This is called the Kojima type bound for the zeros of the simple monic polynomial $p_l(z)$.

Bounds for the zeros of quaternionic polynomial $p_r(z)$: For computation of bounds of the zeros of $p_r(z)$, we define the following polynomial:

$$\tilde{p}_l(z) := \overline{p_r(\overline{z})} := \sum_{j=0}^m \overline{q_j} z^j.$$

Now, we discuss the following theorem which shows relation between the zeros of $p_r(z)$ and $\tilde{p}_l(z)$.

Theorem 2.31. Let $\lambda \in \mathbb{H}$. Then λ is a zero of the simple monic polynomial $p_r(z)$ if and only if $\overline{\lambda}$ is a zero of the simple monic polynomial $\tilde{p}_l(z)$.

Proof. The corresponding companion matrices of $p_r(z)$ and $\tilde{p}_l(z)$ are given by

$$C_{p_r} := C_{p_l}^T$$
 and $C_{\tilde{p}_l} := C_{p_r}^H$,

respectively. By Lemma 2.1, if λ is a left eigenvalue of C_{p_r} , then $\overline{\lambda}$ is a left eigenvalue of $C_{p_r}^H = C_{\tilde{p}_l}$. By Propositions 1.36 and 1.37, the left eigenvalues of C_{p_r} and $C_{\tilde{p}_l}$ imply the zeros of $p_r(z)$ and $\tilde{p}_l(z)$, respectively. Hence if λ is a zero of $p_r(z)$, then $\overline{\lambda}$ is also a zero of $\tilde{p}_l(z)$.

Remark 2.32. Similar results can be obtained for the quaternionic polynomial $p_r(z)$ as well.

Bounds for the zeros of quaternionic polynomials by using the powers of companion matrices: First, we present some preliminary results for the powers of companion matrices C_{p_l} and C_{p_r} . In general, if λ is a left eigenvalue of a quaternionic matrix A, then λ^2 is not necessarily a left eigenvalue of A^2 follows from the following quaternionic matrix

$$A = \begin{bmatrix} 0 & \mathbf{i} \\ -\mathbf{i} & 0 \end{bmatrix}.$$

Then

$$\Lambda_l(A) := \left\{ \mu : \mu = \alpha + \beta \mathbf{j} + \gamma \mathbf{k}, \alpha^2 + \beta^2 + \gamma^2 = 1 \right\}.$$
⁴⁵

Now, we have

$$A^{2} = \begin{bmatrix} 1 & 0 \\ 0 & 1 \end{bmatrix}$$
. So $\Lambda_{l}(A^{2}) := \{1\}.$

Here, **j** is a left eigenvalue of A. However, \mathbf{j}^2 is not a left eigenvalue of A^2 .

Now, we prove the following result for left eigenvalues of C_{p_l} and $C_{p_l}^t$ (t is a nonzero integer).

Proposition 2.33. If λ is a left eigenvalue of C_{p_l} with respect to the eigenvector $x \in \mathbb{H}^n$, then λ^t (t is a nonzero integer) is a left eigenvalue of $C_{p_l}^t$ corresponding to the same eigenvector $x \in \mathbb{H}^n$.

Proof. Case (a): When t is a nonzero positive integer. Let λ be a left eigenvalue of C_{p_l} . Then, there exists $0 \neq x := [1, \lambda, \lambda^2, \dots, \lambda^{m-1}]^T \in \mathbb{H}^n$ such that $C_{p_l}x = \lambda x$. Therefore,

$$C_{p_l}^2 x = C_{p_l}(C_{p_l}x) = C_{p_l}x\lambda = x\lambda^2$$

$$\vdots$$

$$C_{p_l}^t x = C_{p_l}^{t-1}(C_{p_l}x) = C_{p_l}^{t-1}x\lambda = \dots = x\lambda^t = \lambda^t x.$$

Thus, λ^t is a left eigenvalue of matrix $C_{p_l}^t$ corresponding to the same eigenvector $x \in \mathbb{H}^n$.

Case (b): When t is a negative integer. From **Case (a)**, we have $C_{p_l}x = x\lambda$. This implies $C_{p_l}^{-1}x = x\lambda^{-1}$. Therefore,

$$C_{p_{l}}^{-2}x = C_{p_{l}}^{-1}(C_{p_{l}}^{-1}x) = C_{p_{l}}^{-1}x\lambda^{-1} = x\lambda^{-2}$$

$$\vdots$$

$$C_{p_{l}}^{t}x = C_{p_{l}}^{(t+1)}(C_{p_{l}}^{-1}x) = C_{p_{l}}^{(t+1)}x\lambda^{-1} = \dots = x\lambda^{t} = \lambda^{t}x.$$

Thus, λ^t is a left eigenvalue of $C_{p_l}^t$ with respect to the same eigenvector $x \in \mathbb{H}^n$.

Next, we state the following result for left eigenvalues of C_{p_r} and $C_{p_r}^t$ (t is a nonzero integer).

Proposition 2.34. If λ is a left eigenvalue of C_{p_r} with respect to the eigenvector $x \in \mathbb{H}^n$, then λ^t (t is a nonzero integer) is a left eigenvalue of $C_{p_r}^t$ corresponding to the same eigenvector $x \in \mathbb{H}^n$.

Proof. Case (a): When t is a nonzero positive integer. Let λ be a left eigenvalue of C_{p_r} . Now from Lemma 2.1, $\overline{\lambda}$ is a left eigenvalue of $C_{p_r}^H$. Then there exists $0 \neq x := [1, \overline{\lambda}, (\overline{\lambda})^2, \dots, (\overline{\lambda})^{m-1}] \in \mathbb{H}^n$ such that $C_{p_r}^H x = \overline{\lambda} x = x\overline{\lambda}$. This gives

$$(C_{p_r}^H)^2 x = C_{p_r}^H (C_{p_r}^H x) = C_{p_r}^H x \overline{\lambda} = x(\overline{\lambda})^2$$

$$\vdots$$

$$(C_{p_r}^H)^t x = (C_{p_r}^H)^{t-1} (C_{p_r}^H x) = (C_{p_r}^H)^{t-1} x \overline{\lambda} = \dots = x(\overline{\lambda})^t = (\overline{\lambda})^t x.$$

Thus, $(\overline{\lambda})^t$ is a left eigenvalue of $(C_{p_r}^H)^t$. Then by Lemma 2.1, λ^t is a left eigenvalue of $C_{p_r}^t$.

Case (b): When t is a negative integer. From **Case (a)**, we have $C_{p_r}^H x = \overline{\lambda} x = x\overline{\lambda}$. This implies $(C_{p_r}^H)^{-1}x = x(\overline{\lambda})^{-1}$. Thus

$$(C_{p_r}^H)^{-2}x = (C_{p_r}^H)^{-1}\{(C_{p_r}^H)^{-1}x\} = (C_{p_r}^H)^{-1}x(\overline{\lambda})^{-1} = x(\overline{\lambda})^{-2}$$

$$\vdots$$

$$(C_{p_r}^H)^t x = (C_{p_r}^H)^{(t+1)}\{(C_{p_r}^H)^{-1}x\} = (C_{p_r}^H)^{(t+1)}x(\overline{\lambda})^{-1} = \dots = x(\overline{\lambda})^t = (\overline{\lambda})^t x.$$

Thus, $(\overline{\lambda})^t$ is a left eigenvalue of $(C_{p_r}^H)^t$. Then by Lemma 2.1, λ^t is a left eigenvalue of $C_{p_r}^t$.

Further, we present a framework to find the powers of the companion matrix C_{p_l} which can be derived in a simple procedure as follows, keeping in view that quaternions do not commute.

Theorem 2.35. Consider
$$C_{p_l} = \begin{bmatrix} 1 & m-1 \\ 0 & I \end{bmatrix}$$
.
(a) If $t < m$ is a positive integer, then

(2.26)
$$C_{p_l}^t = \begin{array}{c|c} t & m-t \\ m-t \\ t \end{array} \begin{bmatrix} 0 & I \\ \hline C & D \end{bmatrix},$$

$$47$$

(b) if $t \ge m$, then

(2.27)
$$C_{p_{l}}^{t} = \begin{bmatrix} C_{p_{l}}^{t-(m-1)}(m, 1:m) \\ C_{p_{l}}^{t-(m-2)}(m, 1:m) \\ \vdots \\ C_{p_{l}}^{t-1}(m, 1:m) \\ C_{p_{l}}^{t}(m, 1:m) \end{bmatrix}_{m \times m}^{t}$$

where

$$\begin{split} C_{p_{l}}^{t}(m,1) &:= C_{p_{l}}^{t-1}(m,m)C_{p_{l}}(m,1), \\ C_{p_{l}}^{t}(m,2:m) &:= C_{p_{l}}^{t-1}(m,1:m-1) + C_{p_{l}}^{t-1}(m,m)C_{p_{l}}(m,2:m), \\ C &:= \begin{bmatrix} C_{p_{l}}(m,1:t) \\ C_{p_{l}}^{2}(m,1:t) \\ \vdots \\ C_{p_{l}}^{t}(m,1:t) \end{bmatrix}_{t \times t}, \text{ and } D &:= \begin{bmatrix} C_{p_{l}}(m,t+1:m) \\ C_{p_{l}}^{2}(m,t+1:m) \\ \vdots \\ C_{p_{l}}^{t}(m,t+1:m) \end{bmatrix}_{t \times (m-t)}. \end{split}$$

Note that $C_{p_l}(k, 1:m)$ denotes the k-th row of the matrix C_{p_l} .

Proof. Assuming t = 1, then (2.26) becomes

$$C_{p_l} = {\begin{array}{c|c}m-1\\ 1\end{array}} \begin{bmatrix} 1 & m-1\\ \\ 0 & I\\ \hline C_{p_l}(m,1) & C_{p_l}(m,2:m) \end{bmatrix},$$

where $C_{p_l}(m,1) := -q_0, C_{p_l}(m,2:m) := [-q_1 \dots - q_{m-1}]$. Thus the theorem is true for t = 1. Now, let us consider C_{p_l} as

$$C_{p_l} = \frac{k}{m-k} \begin{bmatrix} \frac{m-k}{k} & k \\ \frac{A'}{C'} & B' \\ \hline C' & D' \end{bmatrix}, \text{ where}$$

 $A' := C_{p_l}(1:k,1:m-k), B' := C_{p_l}(k+1:m,m-k+1:m), C' := C_{p_l}(k+1:m,1:m), C' := C_{p_l$ $(m-k), D' := C_{p_l}(k+1:m,m-k+1:m).$ For t = k = 3, we get

$$C_{p_{l}}^{3} = {}^{m-2} \begin{bmatrix} 2 & m-2 & m-2 & 2 \\ 0 & I \\ \hline C & D \end{bmatrix} {}^{2} \begin{bmatrix} A' & B' \\ \hline C' & D' \end{bmatrix} = {}^{m-2} \begin{bmatrix} C' & D' \\ \hline CA' + DC' & CB' + DD' \end{bmatrix}$$

$$48$$

Note that in each step, size of the identity matrix I reduces by order 1 and the size of matrix C increases by order 1. Similarly, the matrix D increases by 1 row and decreases by 1 column. Finally, after rearranging and separating 0 and I matrices we get

$$\begin{array}{c|cccc} & & & & & & & & & \\ m-2-1 & & & & & & \\ m-2+1 & & & & & & \\ \hline \hline C & & & D & \\ \hline \end{array} \right],$$

where C and D are of size 3×3 and $3 \times (m-3)$, respectively. Assuming that the theorem is true for t = k. Then we have

where the corresponding C and D matrices are given in the statement of the theorem. Similarly, we can prove for $t \ge m$.

In the case of quaternionic matrix, $C_{p_l} = C_{p_r}^T$ but $C_{p_r}^t \neq (C_{p_l}^t)^T$ for $t \ge 2$. It follows from the following example.

Example 2.36. Consider the following simple monic polynomials over \mathbb{H} :

$$p_l(z) = z^3 - \mathbf{k}z^2 + (\mathbf{k} - \mathbf{j})z + (\mathbf{i} + \mathbf{j}) \text{ and } p_r(z) = z^3 - z^2\mathbf{k} + z(\mathbf{k} - \mathbf{j}) + (\mathbf{i} + \mathbf{j}).$$

The corresponding companion matrices of $p_l(z)$ and $p_r(z)$ are given by

$$C_{p_l} = {\begin{array}{*{20}c} 1 & 2 \\ 0 & I \\ 1 & C_{p_l}(3,1) & C_{p_l}(3,2:3) \end{array}} \text{ and } C_{p_r} = C_{p_l}^T,$$

respectively, where $C_{p_l}(3,1) = -\mathbf{i} - \mathbf{j}$ and $C_{p_l}(3,2:3) := [\mathbf{j} - \mathbf{k}, \mathbf{k}]$. Then

$$C_{p_{l}}^{2} = \begin{bmatrix} 0 & 0 & 1 \\ -\mathbf{i} - \mathbf{j} & \mathbf{j} - \mathbf{k} & \mathbf{k} \\ \mathbf{i} - \mathbf{j} & \mathbf{1} - 2\mathbf{i} - \mathbf{j} & \mathbf{j} - \mathbf{k} - \mathbf{1} \end{bmatrix} \text{ and } C_{p_{r}}^{2} = \begin{bmatrix} 0 & -\mathbf{i} - \mathbf{j} & \mathbf{j} - \mathbf{i} \\ 0 & \mathbf{j} - \mathbf{k} & \mathbf{1} - \mathbf{j} \\ 1 & \mathbf{k} & \mathbf{j} - \mathbf{k} - \mathbf{1} \end{bmatrix}.$$

This shows that $C_{p_r}^2 \neq (C_{p_l}^2)^T$.

Hence, we can write the similar algorithm to Theorem 2.35 for the case of $C_{p_r}^t$, $t \ge 2$.

Theorem 2.37. Consider $C_{p_r} = \frac{1}{m-1} \begin{bmatrix} 0 & C_{p_r}(1,m) \\ \hline I & C_{p_r}(2:m,m) \end{bmatrix}$. (a) If t < m is a positive integer, then

(2.28)
$$C_{p_r}^t = \begin{array}{c} m-t & t \\ \hline 0 & C \\ \hline I & D \end{array} \right],$$

(b) if $t \ge m$, then

 $C_{p_r}^t = \left[\begin{array}{ccc} C_{p_r}^{t-(m-1)}(1:m,m) & C_{p_r}^{t-(m-2)}(1:m,m) & \dots & C_{p_r}^{t-1}(1:m,m) & C_{p_r}^t(1:m,m) \end{array} \right]_{m \times m},$

where

Proof. The proof follows from the proof method of Theorem 2.35.

Now from Example 2.36, we write

$$\tilde{p}_l(z) := \overline{p_r(\overline{z})} = z^3 + \mathbf{k}z^2 + (\mathbf{j} - \mathbf{k})z + (-\mathbf{i} - \mathbf{j}), \text{ and}$$

 $\tilde{p}_r(z) := \overline{p_l(\overline{z})} = z^3 + z^2\mathbf{k} + z(\mathbf{j} - \mathbf{k}) - (\mathbf{i} + \mathbf{j}).$

Thus the companion matrices corresponding to $\tilde{p}_l(z)$ and $\tilde{p}_r(z)$ are given by

$$C_{\tilde{p}_l} = \overline{C_{p_l}} \text{ and } C_{\tilde{p}_r} = \overline{C_{p_r}},$$

respectively. Next,

$$C_{\tilde{p}_l}^2 = \begin{bmatrix} 0 & 0 & 1 \\ \mathbf{i} + \mathbf{j} & -\mathbf{j} + \mathbf{k} & -\mathbf{k} \\ \mathbf{i} - \mathbf{j} & \mathbf{1} + \mathbf{j} & \mathbf{k} - \mathbf{j} - \mathbf{1} \end{bmatrix} \text{ and } C_{\tilde{p}_r}^2 = \begin{bmatrix} 0 & \mathbf{i} + \mathbf{j} & \mathbf{j} - \mathbf{i} \\ 0 & -\mathbf{j} + \mathbf{k} & \mathbf{1} + 2\mathbf{i} + \mathbf{j} \\ 1 & -\mathbf{k} & -\mathbf{1} - \mathbf{j} + \mathbf{k} \end{bmatrix}.$$

Then

(a)
$$\max_{1 \le i \le 3} \left[(r'_i(C_{p_l}^2))^{1/2} \right] = 2.3655 \text{ and } \max_{1 \le i \le 3} \left[(r'_i(C_{\tilde{p_r}}^2))^{1/2} \right] = 1.9656.$$
(b) $\max_{1 \le i \le 3} \left[\left(r'_i \left(C_{p_r}^2 \right) \right)^{1/2} \right] = 1.9319 \text{ and } \max_{1 \le i \le 3} \left[\left(r'_i (C_{\tilde{p}_i}^2) \right)^{1/2} \right] = 2.1355.$

Now, we have

$$\max_{1 \le i \le 3} \left[(r'_i(C_{p_l}^2))^{1/2} \right] \neq \max_{1 \le i \le 3} \left[(r'_i(C_{\tilde{p_r}}^2))^{1/2} \right] \text{ and}$$
$$\max_{1 \le i \le 3} \left[(r'_i(C_{p_r}^2))^{1/2} \right] \neq \max_{1 \le i \le 3} \left[(r'_i(C_{\tilde{p_l}}^2))^{1/2} \right].$$

Hence, we have the following bounds for the zeros of $p_l(z)$ and $p_r(z)$ for $\gamma \in [0, 1]$.

Theorem 2.38. Let $p_l(z)$ and $p_r(z)$ be the simple monic polynomials over \mathbb{H} of degree m and let $C_{p_l}^t$ and $C_{p_r}^t$ $(t \ge 2)$ be the t-th power of the companion matrices C_{p_l} and C_{p_r} , corresponding to $p_l(z)$ and $p_r(z)$, respectively. Then, for $\gamma \in [0, 1]$

1. bounds for every zero \tilde{z} of $p_l(z)$ satisfies the following inequalities:

(a) $(\xi'_1)^{-1} \le |\tilde{z}| \le \xi_1,$ (b) $(\xi'_2)^{-1} \le |\tilde{z}| \le \xi_2,$

where

$$\begin{aligned} \xi_{1} &= \max_{1 \leq i \leq m} \left[\left(r'_{i} \left(C_{p_{l}}^{t} \right) \right)^{\gamma/t} \left(c'_{i} \left(C_{p_{l}}^{t} \right) \right)^{(1-\gamma)/t} \right], \\ \xi'_{1} &= \max_{1 \leq i \leq m} \left[\left(r'_{i} \left(C_{q_{l}}^{t} \right) \right)^{\gamma/t} \left(c'_{i} \left(C_{q_{l}}^{t} \right) \right)^{(1-\gamma)/t} \right], \\ \xi_{2} &= \max_{1 \leq i \leq m} \left[\left(r'_{i} \left(C_{\tilde{p}_{r}}^{t} \right) \right)^{\gamma/t} \left(c'_{i} \left(C_{\tilde{p}_{r}}^{t} \right) \right)^{(1-\gamma)/t} \right], \\ \xi'_{2} &= \max_{1 \leq i \leq m} \left[\left(r'_{i} \left(C_{\tilde{q}_{r}}^{t} \right) \right)^{\gamma/t} \left(c'_{i} \left(C_{\tilde{q}_{r}}^{t} \right) \right)^{(1-\gamma)/t} \right]; \end{aligned}$$

2. bounds for every zero \tilde{z} of $p_r(z)$ satisfies the following inequalities:

(a)
$$\left(\max_{1 \le i \le m} \left[\left(r_i' \left(C_{q_r}^t \right) \right)^{\gamma/t} \left(c_i' \left(C_{q_r}^t \right) \right)^{(1-\gamma)/t} \right] \right)^{-1} \le |\tilde{z}| \le \xi_3,$$

(b) $\left(\max_{1 \le i \le m} \left[\left(r_i' \left(C_{\tilde{q}_l}^t \right) \right)^{\gamma/t} \left(c_i' \left(C_{\tilde{q}_l}^t \right) \right)^{(1-\gamma)/t} \right] \right)^{-1} \le |\tilde{z}| \le \xi_4,$

where

$$\begin{aligned} \xi_{3} &= \max_{1 \leq i \leq m} \left[\left(r'_{i} \left(C_{p_{r}}^{t} \right) \right)^{\gamma/t} \left(c'_{i} \left(C_{p_{r}}^{t} \right) \right)^{(1-\gamma)/t} \right], \\ \xi'_{3} &= \max_{1 \leq i \leq m} \left[\left(r'_{i} \left(C_{q_{r}}^{t} \right) \right)^{\gamma/t} \left(c'_{i} \left(C_{q_{r}}^{t} \right) \right)^{(1-\gamma)/t} \right], \\ \xi_{4} &= \max_{1 \leq i \leq m} \left[\left(r'_{i} \left(C_{\tilde{p}_{l}}^{t} \right) \right)^{\gamma/t} \left(c'_{i} \left(C_{\tilde{p}_{l}}^{t} \right) \right)^{(1-\gamma)/t} \right], \\ \xi'_{4} &= \max_{1 \leq i \leq m} \left[\left(r'_{i} \left(C_{\tilde{q}_{l}}^{t} \right) \right)^{\gamma/t} \left(c'_{i} \left(C_{\tilde{q}_{l}}^{t} \right) \right)^{(1-\gamma)/t} \right]. \end{aligned}$$

Proof. 1(*a*). Let λ be a left eigenvalue of C_{p_l} . Then by Proposition 2.33, λ^t ($t \geq 2$ is positive integer) is a left eigenvalue of $C_{p_l}^t$. Hence by applying Theorem 2.3, we get the desired result.

1(b). By Lemma 2.1, $\overline{\lambda}$ is a left eigenvalue of $C_{\tilde{p}_r}$ and by Proposition 2.34, $(\overline{\lambda})^t$ is a left eigenvalue of $(C_{\tilde{p}_r})^t$. Then from Theorem 2.3, we get the desired result.

For 2(a) and 2(b), the proofs are similar to the proof methods of 1(a) and 1(b), respectively. Substituting t = 2 and $\gamma = 1$ in Theorem 2.38, we have the following corollary.

Corollary 2.39. Let $p_l(z)$ and $p_r(z)$ be the simple monic polynomials over \mathbb{H} of degree m. Then

1. bounds for every zero \tilde{z} of $p_l(z)$ satisfies the following inequalities:

(a)
$$\frac{1}{\beta_1} \le |\tilde{z}| \le \alpha_1,$$

(b) $\frac{1}{\beta_2} \le |\tilde{z}| \le \alpha_2,$

where

$$\alpha_1 = \max\left\{1, \left(\sum_{j=0}^{m-1} |q_j|\right)^{1/2}, \left(\sum_{j=0}^{m-1} |q_{m-1}q_j - q_{j-1}|\right)^{1/2}\right\},\$$

$$\begin{aligned} \alpha_{2} &= \max_{2 \leq j \leq m-1} \left\{ \left(|q_{0}| + |\overline{q_{0}} \ \overline{q_{m-1}}| \right)^{1/2}, \left(|q_{1}| + |\overline{q_{1}} \ \overline{q_{m-1}} - \overline{q_{0}}| \right)^{1/2}, \\ \left(1 + |q_{j}| + |\overline{q_{j}} \ \overline{q_{m-1}} - \overline{q_{j-1}}| \right)^{1/2} \right\}, \\ \beta_{1} &= \max \left\{ 1, \left(\sum_{j=1}^{m-1} |q_{0}^{-1}q_{j}| \right)^{1/2}, \left(\sum_{j=0}^{m-1} |q_{0}^{-1}q_{1}q_{0}^{-1}q_{m-j} - q_{0}^{-1}q_{m-j+1}| \right)^{1/2} \right\}, \\ \beta_{2} &= \max_{2 \leq j \leq m-1} \left\{ \left(|q_{0}^{-1}| + |\overline{q_{0}^{-1}} \ \overline{q_{1}q_{0}^{-1}}| \right)^{1/2}, \left(|q_{m-1}q_{0}^{-1}| + |\overline{q_{m-1}q_{0}^{-1}} \ \overline{q_{1}q_{0}^{-1}} - \overline{q_{0}^{-1}}| \right)^{1/2} \right\}, \\ \left(1 + |q_{m-j}q_{0}^{-1}| + |\overline{q_{m-j}q_{0}^{-1}} \ \overline{q_{1}q_{0}^{-1}} - \overline{q_{m-j+1}q_{0}^{-1}}| \right)^{1/2} \right\}; \end{aligned}$$

2. bounds for every zero \tilde{z} of $p_r(z)$ satisfies the following inequalities:

(a)
$$\frac{1}{\beta_3} \le |\tilde{z}| \le \alpha_3,$$

(b) $\frac{1}{\beta_4} \le |\tilde{z}| \le \alpha_4,$

where

$$\begin{aligned} \alpha_{3} &= \max_{2 \leq j \leq m-1} \left\{ \left(|q_{0}| + |q_{0}|q_{m-1}| \right)^{1/2}, \left(|q_{1}| + |q_{1}|q_{m-1} - q_{0}| \right)^{1/2}, \\ \left(1 + |q_{j}| + |q_{j}|q_{m-1} - q_{j-1}| \right)^{1/2} \right\}, \\ \alpha_{4} &= \max \left\{ 1, \left(\sum_{j=0}^{m-1} |q_{j}| \right)^{1/2}, \left(\sum_{j=0}^{m-1} |\overline{q_{m-1}} \ \overline{q_{j}} - \overline{q_{j-1}}| \right)^{1/2} \right\}, \\ \beta_{3} &= \max_{2 \leq j \leq m-1} \left\{ \left(|q_{0}^{-1}| + |q_{0}^{-1}| \ q_{1}q_{0}^{-1}| \right)^{1/2}, \left(|q_{m-1}q_{0}^{-1}| + |q_{m-1}q_{0}^{-1}| \ q_{1}q_{0}^{-1} - q_{m-j+1}q_{0}^{-1}| \right)^{1/2}, \\ \left(1 + |q_{m-j}q_{0}^{-1}| + |q_{m-j}q_{0}^{-1}| \ q_{1}q_{0}^{-1} - q_{m-j+1}q_{0}^{-1}| \right)^{1/2} \right\}, \\ \beta_{4} &= \max \left\{ 1, \left(\sum_{j=1}^{m-1} |q_{0}^{-1}q_{j}| \right)^{1/2}, \left(\sum_{j=0}^{m-1} |\overline{q_{0}^{-1}q_{1}} \ \overline{q_{0}^{-1}q_{m-j}} - \overline{q_{0}^{-1}q_{m-j+1}}| \right)^{1/2} \right\}, \\ q_{-1} = 0 = q_{m+1}, q_{m} = 1. \end{aligned}$$

Proof. The proof follows from Theorem 2.38 and APPENDIX-A. ■

Example 2.40. Consider the following polynomials $p_l(z)$ and $p_r(z)$ over \mathbb{H} :

$$p_l(z) = z^6 + (\mathbf{i} + 3\mathbf{k})z^5 + (3 + \mathbf{j})z^4 + (5\mathbf{i} + 15\mathbf{k})z^3 + (-4 + 5\mathbf{j})z^2 + (6\mathbf{i} + 18\mathbf{k})z + (6\mathbf{j} - 12),$$

$$p_r(z) = z^6 + z^5(\mathbf{i} + 3\mathbf{k}) + z^4(3 + \mathbf{j}) + z^3(5\mathbf{i} + 15\mathbf{k}) + z^2(-4 + 5\mathbf{j}) + z(6\mathbf{i} + 18\mathbf{k}) + (6\mathbf{j} - 12).$$

The zeros of $p_l(z)$ are given in [50]. Moreover, we find the zeros of $p_r(z)$ by Niven's algorithm [41].

$\mathbf{Z}_{\mathbb{H}}(p_l(z)) =: z_1$	$\operatorname{Abs}(z_1)$	$\mathbf{Z}_{\mathbb{H}}(p_r(z)) =: z_2$	$\operatorname{Abs}(z_2)$
-i-2k	2.2361	$-0.4\mathbf{i} - 2.2\mathbf{k}$	2.2361
$[\mathbf{i}\sqrt{3}]$	1.7321	$[\mathbf{i}\sqrt{3}]$	1.7321
$[\mathbf{i}\sqrt{2}]$	1.4142	$[\mathbf{i}\sqrt{2}]$	1.4142
$-0.6\mathbf{i} - 0.8\mathbf{k}$	1	$-\mathbf{k}$	1

TABLE 2.1. The zeros of $p_l(z)$ and $p_r(z)$ and their absolute values.

 $z_1 :=$ the set of zeros of $p_l(z)$, $z_2 :=$ the set of zeros of $p_r(z)$,

Denote LB:= Lower Bound, UB:= Upper Bound.

Example 2.40	\mathbf{LB}	UB
Corollary 2.27 (1)	0.4142	19.9737
Corollary 2.27 (2)	0.2766	60.9291
Theorem 2.26, $\gamma = 1/4$	0.3744	8.1415

TABLE 2.2. Lower and upper bounds for the zeros of $p_l(z)$ and $p_r(z)$.

Example 2.36	\mathbf{LB}	UB
Corollary 2.39 $1(a)$	0.6156	2.3655
Corollary 2.39 $1(b)$	0.6078	1.9656
Corollary 2.39 $2(a)$	0.6078	1.9319
Corollary 2.39 $2(b)$	0.6436	2.1355

TABLE 2.3. Lower and upper bounds for the zeros of $p_l(z)$ and $p_r(z)$.

CHAPTER 3

LOCATION OF ZEROS OF POLYNOMIALS OVER A QUATERNION DIVISION ALGEBRA

In this chapter, inclusion regions for the left and right eigenvalues of quaternionic matrices are derived. The location of zeros of quaternionic polynomials is discussed via left eigenvalues of the companion matrices. The present investigation shows that the obtained ovals of Cassini are smaller than the existing ovals of Cassini for polynomials over the complex field under certain conditions.

3.1. Introduction

The work carried out in this chapter is based on Theorems 2.3, 2.4, 2.8, and 2.11. In this work, we discuss a theory for polynomials with quaternion coefficients. Thereafter we move to study similar results for polynomials with complex coefficients. Then, inclusion regions for the left and right eigenvalues of a quaternionic matrix are derived. Further, the location of zeros of the quaternionic polynomials $p_l(z)$ and $p_r(z)$ (defined in (1.20) and (1.21)) is discussed. The present work also focuses on inclusion regions for the zeros of the quaternionic polynomials $p_l(z)$ and $p_r(z)$ by applying Theorem 2.3 and Theorem 2.8. It is found that Theorem 2.3 and Theorem 2.8 give two sets; the first one describes the union of two balls and the second one is the union of a ball and an oval of Cassini. These sets can be reduced into smaller sets which depend on a parameter $\gamma \in [0, 1]$. By considering $\omega(z) = z^2 + q_{m-1}z$, where $q_{m-1}, z \in \mathbb{H}$, we present two ovals of Cassini and each contains all the zeros of the quaternionic polynomials $p_l(z)$ and $p_r(z)$. We show that the inclusion regions obtained in the present work are comparatively smaller than the inclusion regions developed in [37] under certain conditions.

3.2. Inclusion regions for left eigenvalues of quaternionic com-

panion matrices

The corresponding companion matrices to the simple monic polynomials $p_l(z)$ and $p_r(z)$ are given by

$$C_{p_l} := \begin{bmatrix} 0 & 1 & 0 \\ \vdots & \ddots & \\ 0 & 0 & 1 \\ -q_0 & -q_1 & \dots & -q_{m-1} \end{bmatrix} \text{ and } C_{p_r} := C_{p_l}^T,$$

respectively. Left eigenvalues of C_{p_l} are the zeros of the simple monic polynomial $p_l(z)$. Also, left eigenvalues of C_{p_r} are the zeros of the simple monic polynomial $p_r(z)$. However, right eigenvalues of C_{p_l} need not be zeros of the simple monic polynomial $p_l(z)$. Now define $\omega(z)$ as:

(3.1)
$$\omega(z) := \sum_{j \in S} \alpha_j z^j,$$

where $\alpha_j, z \in \mathcal{K}, \mathcal{K} \in \{\mathbb{R}, \mathbb{C}, \mathbb{H}\}$, and S is defined as the set of integers when $A \in M_n(\mathcal{K})$ is invertible; otherwise S is the set of nonnegative integers. Throughout this chapter we consider the following three cases:

Case 1: when $\alpha_i, z \in \mathbb{H}$, and $A \in M_n(\mathbb{H})$,

Case 2: when $\alpha_j \in \mathbb{R}, z \in \mathbb{H}$, and $A \in M_n(\mathbb{H})$,

Case 3: when $\alpha_j, z \in \mathbb{C}$, and $A \in M_n(\mathbb{C})$.

In general, if λ is a left eigenvalue of a matrix $A \in M_n(\mathbb{H})$, then λ^2 is not necessarily a left eigenvalue of A^2 . If λ is a left eigenvalue of the quaternionic matrix A with corresponding eigenvector $x \in \mathbb{H}^n$, then λ^{-1} is a left eigenvalue of A^{-1} . However its corresponding eigenvector may be different to x. On the other hand, if λ is a left eigenvalue of C_{p_l} with corresponding eigenvector $y \in \mathbb{H}^n$, then λ^k is a left eigenvalue of $C_{p_l}^k$ with the same eigenvector y, where k is a nonzero integer. The above argument follows from Proposition 2.33. Now, we take **Case 1** for the development of our theory.

Case 1. We consider $\omega(z)$ as:

(3.2)
$$\omega(z) := \sum_{j \in S} \alpha_j z^j,$$

where $\alpha_j, z \in \mathbb{H}$ and S is defined as the set of integers when $A \in M_n(\mathbb{H})$ is invertible; otherwise S is the set of nonnegative integers. Then we have the following result.

Lemma 3.1. Let C_{p_l} be a companion matrix of the simple monic polynomial $p_l(z)$ and let $\omega(z)$ be defined in (3.2). If λ is a left eigenvalue of C_{p_l} , then $\omega(\lambda)$ is a left eigenvalue of $\omega(C_{p_l})$.

Proof. Let λ be a left eigenvalue of C_{p_l} . Then $C_{p_l}x = \lambda x$ for some nonzero $x \in \mathbb{H}^n$. So

$$\omega(C_{p_l})x = \left(\sum_{j \in S} \alpha_j C_{p_l}^j\right) x.$$

From Proposition 2.33, we obtain

$$\omega(C_{p_l})x = \left(\sum_{j \in S} \alpha_j \lambda^j\right) x = \omega(\lambda)x. \blacksquare$$

We next present a generalization of Theorems 2.3 and 2.8 for a quaternionic companion matrix as follows.

Theorem 3.2. Let $C_{p_l} := (c_{ij}) \in M_n(\mathbb{H})$ be a companion matrix of the simple monic polynomial $p_l(z)$ and let $\gamma \in [0,1]$. Then all the left eigenvalues of C_{p_l} are located in the union of n sets $B_i(C_{p_l}, \omega) := \{z \in \mathbb{H} : |\omega(z) - (\omega(C_{p_l}))_{ii}| \le r_i^{\gamma}(\omega(C_{p_l})) c_i^{1-\gamma}(\omega(C_{p_l}))\}$ $(1 \le i \le n), i.e.,$

$$\Lambda_l(C_{p_l}) \subseteq B(C_{p_l}, \omega) := \bigcup_{i=1}^n B_i(C_{p_l}, \omega),$$

where $\omega(z)$ is defined in (3.2).

Proof. Let λ be a left eigenvalue of C_{p_l} . Then from Lemma 3.1, $\omega(\lambda)$ is a left eigenvalue of $\omega(C_{p_l})$. Then there exists some nonzero $x \in \mathbb{H}^n$ such that $\omega(C_{p_l})x = \omega(\lambda)x$. Let $x := [x_1, \ldots, x_n]^T \in \mathbb{H}^n$ and let x_t be an element of x such that $|x_t| \ge |x_i| \forall i \ (1 \le i \le n)$. Then $|x_t| > 0$. From the t-th equation of $\omega(C_{p_l})x = \omega(\lambda)x$, we have

$$\sum_{j=1}^{n} \omega(C_{p_l})_{tj} x_j = \omega(\lambda) x_t.$$

Since $|x_t| \ge |x_i| \forall i \ (1 \le i \le n)$, then

(3.3)
$$|\omega(\lambda) - \omega(C_{p_l})_{tt}| \leq \sum_{\substack{j=1, j \neq t \\ 57}}^{n} |\omega(C_{p_l})_{tj}| := r_t(\omega(C_{p_l})).$$

From Lemma 2.1, $\overline{\lambda}$ is a left eigenvalue of A^H and hence $\overline{\omega(\lambda)}$ is a left eigenvalue of $(\omega(C_{p_l}))^H$. Similarly, we have

$$|\overline{\omega(\lambda)} - \overline{\omega(C_{p_l})_{tt}}| \le \sum_{j=1, j \neq t}^n |\overline{\omega(C_{p_l})_{jt}}|.$$

This implies

(3.4)
$$|\omega(\lambda) - \omega(C_{p_l})_{tt}| \leq \sum_{j=1, j \neq t}^n |\omega(C_{p_l})_{jt}| := c_t(\omega(C_{p_l})).$$

Then for any $\gamma \in [0, 1]$, (3.3) and (3.4) give

(3.5)
$$|\omega(\lambda) - \omega(C_{p_l})_{tt}|^{\gamma} \leq r_t^{\gamma}(\omega(C_{p_l})),$$

as well as

(3.6)
$$|\omega(\lambda) - \omega(C_{p_l})_{tt}|^{1-\gamma} \leq c_t^{1-\gamma}(\omega(C_{p_l})),$$

Combining (3.5) and (3.6), we obtain

$$|\omega(\lambda) - \omega(C_{p_l})_{tt}| \le r_t^{\gamma}(\omega(C_{p_l})) c_t^{1-\gamma}(\omega(C_{p_l})). \blacksquare$$

From Theorems 2.3 and 3.2, it is clear that all the left eigenvalues of C_{p_l} are contained in $T(C_{p_l}) \cap B(C_{p_l}, \omega)$.

Theorem 3.3. Let $C_{p_l} := (c_{ij}) \in M_n(\mathbb{H})$ be a companion matrix of the simple monic polynomial $p_l(z)$ and let $\gamma \in [0,1]$. Then all the left eigenvalues of C_{p_l} are located in the union of $\frac{n(n-1)}{2}$ sets $P_{ij}(C_{p_l}, \omega) := \{z \in \mathbb{H} : |\omega(z) - (\omega(C_{p_l}))_{ii}| |\omega(z) - (\omega(C_{p_l}))_{jj}| \le$ $r_i^{\gamma}(\omega(C_{p_l})) r_j^{\gamma}(\omega(C_{p_l})) c_i^{1-\gamma}(\omega(C_{p_l})) c_j^{1-\gamma}(\omega(C_{p_l}))\}$ $(1 \le i, j \le n; i \ne j)$, i.e.,

$$\Lambda_l(C_{p_l}) \subseteq P(C_{p_l}, \omega) := \bigcup_{\substack{i,j=1\\i\neq j}}^n P_{ij}(C_{p_l}, \omega),$$

where $\omega(z)$ is defined in (3.2).

Proof. Let λ be a left eigenvalue of C_{p_l} . Then by Lemma 3.1, $\omega(\lambda)$ is a left eigenvalue of $\omega(C_{p_l})$. Then there exists some nonzero $x \in \mathbb{H}^n$ such that $\omega(C_{p_l})x = \omega(\lambda)x$. Let $x := [x_1, \ldots, x_n]^T \in \mathbb{H}^n$ and let x_s be an element of x such that $|x_s| \ge |x_i| \forall i \ (1 \le i \le n)$. Then $|x_s| > 0$. Clearly, if all the other elements of x are zero, then the result holds. Now, suppose that there are at least two nonzero elements of x, and let x_t be an element with second largest absolute value, i.e., $|x_s| \ge |x_t| \ge |x_i| \forall i \ (1 \le i \le n, i \ne s)$, and $x_s \ne 0 \ne x_t$. From the s-th equation of $\omega(C_{p_l})x = \omega(\lambda)x$, we have

$$\sum_{j=1}^{n} \omega(C_{p_l})_{sj} x_j = \omega(\lambda) x_s,$$

which implies

$$(\omega(\lambda) - \omega(C_{p_l})_{ss})x_s = \sum_{j=1, j \neq s}^n \omega(C_{p_l})_{sj}x_j.$$

Thus

(3.7)
$$|\omega(\lambda) - \omega(C_{p_l})_{ss}| \le \left(\frac{|x_t|}{|x_s|}\right) r_s(\omega(C_{p_l})).$$

Similarly, from the *t*-th equation of $\omega(C_{p_l})x = \omega(\lambda)x$, we obtain

(3.8)
$$|\omega(\lambda) - \omega(C_{p_l})_{tt}| \le \left(\frac{|x_s|}{|x_t|}\right) r_t(\omega(C_{p_l}))$$

Combining (3.7) and (3.8), we have

(3.9)
$$|\omega(\lambda) - \omega(C_{p_l})_{ss}| |\omega(\lambda) - \omega(C_{p_l})_{tt}| \le r_s(\omega(C_{p_l})) r_t(\omega(C_{p_l}))$$

 $(1 \le s, t \le n, s \ne t).$

Now, from Lemma 2.1, $\overline{\lambda}$ is a left eigenvalue of $C_{p_l}^H$ and hence $\overline{\omega(\lambda)}$ is a left eigenvalue of $(\omega(C_{p_l}))^H$. Similarly, we can obtain

(3.10)
$$|\omega(\lambda) - \omega(C_{p_l})_{ss}| |\omega(\lambda) - \omega(C_{p_l})_{tt}| \le c_s(\omega(C_{p_l})) c_t(\omega(C_{p_l}))$$

 $(1 \le s, t \le n, s \ne t).$

For any $\gamma \in [0, 1]$, (3.9) and (3.10) yield

(3.11)
$$|\omega(\lambda) - \omega(C_{p_l})_{ss}|^{\gamma} |\omega(\lambda) - \omega(C_{p_l})_{tt}|^{\gamma} \le r_s^{\gamma}(\omega(C_{p_l})) r_t^{\gamma}(\omega(C_{p_l})),$$

and

(3.12)
$$|\omega(\lambda) - \omega(C_{p_l})_{ss}|^{1-\gamma} |\omega(\lambda) - \omega(C_{p_l})_{tt}|^{1-\gamma} \le c_s^{1-\gamma}(\omega(C_{p_l})) c_t^{1-\gamma}(\omega(C_{p_l})).$$

Combining (3.11) and (3.12), we obtain

$$|\omega(\lambda) - \omega(C_{p_l})_{ss}| \ |\omega(\lambda) - \omega(C_{p_l})_{tt}| \le r_s^{\gamma}(\omega(C_{p_l})) \ r_t^{\gamma}(\omega(C_{p_l})) \ c_s^{1-\gamma}(\omega(C_{p_l})) \ c_t^{1-\gamma}(\omega(C_{p_l})).$$

Hence, all the left eigenvalues of A are located in the union of $\frac{n(n-1)}{2}$ sets $P_{ij}(C_{p_l},\omega)$ $(1 \le i, j \le n, i \ne j)$.

From Theorems 2.8 and 3.3, we conclude that all the left eigenvalues of C_{p_l} are contained in $K(C_{p_l}) \cap P(C_{p_l}, \omega)$.

In general, similar quaternionic matrices may not have the same left eigenvalues, see, [62, Example 3.3]. However, if $A \in M_n(\mathbb{H})$ and W is any invertible real matrix, then A and WAW^{-1} have the same left eigenvalues (proved in Chapter 2).

Also, in general, $\omega(WAW^{-1}) \neq W\omega(A)W^{-1}$, where W is any quaternionic invertible matrix. For example, let

$$A = \begin{bmatrix} \mathbf{i} + \mathbf{j} & \mathbf{k} \\ \mathbf{j} & 1 - \mathbf{j} \end{bmatrix},$$

where $W = \begin{bmatrix} \mathbf{j} & 0 \\ 0 & \mathbf{k} \end{bmatrix}$ and $W^{-1} = \begin{bmatrix} -\mathbf{j} & 0 \\ 0 & -\mathbf{k} \end{bmatrix}$. Suppose $\omega(z) = z^2 + (\mathbf{j} + \mathbf{k})z$. Then
 $\omega(WAW^{-1}) \neq W\omega(A)W^{-1}$.

However, if W is any invertible real matrix, then we have the following proposition.

Proposition 3.4. Let $A \in M_n(\mathbb{H})$ and let W be a real invertible matrix. Then $\omega(WAW^{-1}) = W\omega(A)W^{-1}$, where $\omega(z)$ be defined in (3.2).

Proof. We have

$$\omega(WAW^{-1}) = \sum_{j \in S} \alpha_j (WAW^{-1})^j.$$

Since $\alpha_j W = W \alpha_j \ \forall j \ (0 \le j \le m), \ \alpha_0 W W^{-1} = W \alpha_0 W^{-1}, \ \text{and} \ (W \alpha_0 W^{-1})^t = W \alpha_0^t W^{-1}$ (*t* is a nonzero integer), we obtain

$$\omega(WAW^{-1}) = W\left(\sum_{j \in S} \alpha_j A^j\right) W^{-1} = W\omega(A)W^{-1}. \blacksquare$$

Remark 3.5. From Proposition 2.22, we observe that left eigenvalues of quaternionic matrices A and WAW^{-1} are same, where W is any invertible real matrix. Applying Theorems 2.3 and 2.8 to WAW^{-1} instead of A, we obtain different and potentially smaller inclusion regions.

Remark 3.6. By applying Theorems 3.2 and 3.3 to $\omega(WC_{p_l}W^{-1}) = W\omega(C_{p_l})W^{-1}$ instead of $\omega(C_{p_l})$, we obtain different and potentially sharper inclusion regions.

Remark 3.7. Similar results can be obtained for the quaternionic companion matrix C_{p_r} as well.

3.3. Inclusion regions for right eigenvalues of quaternionic ma-

trices

Case 2. We consider $\omega(z)$ as:

(3.13)
$$\omega(z) := \sum_{j \in S} \alpha_j z^j,$$

where $\alpha_j \in \mathbb{R}, z \in \mathbb{H}$ and S is defined as the set of integers when $A \in M_n(\mathbb{H})$ is invertible; otherwise S is the set of nonnegative integers. Thus, an extension of the Gerschgorin type theorem [62, Theorem 7] for right eigenvalues of a quaternionic matrix to the quaternionic matrix $\omega(A)$ is as follows.

Theorem 3.8. Let $A = (a_{ij}) \in M_n(\mathbb{H})$. For every right eigenvalue λ of A there exists a nonzero quaternion β such that $\beta^{-1}\omega(\lambda)\beta$ (which is a right eigenvalue of $\omega(A)$) is contained in the union of n sets

$$\mathcal{B}_i(A,\omega) := \{ z \in \mathbb{H} : |\omega(z) - (\omega(A))_{ii}| \le r_i(\omega(A)) \} \quad (1 \le i \le n), \ i.e.$$
$$\{ z^{-1}\omega(\lambda)z : 0 \ne z \in \mathbb{H} \} \cap \cup_{i=1}^n \mathcal{B}_i(A,\omega) \ne \emptyset,$$

where $\omega(z)$ is defined in (3.13).

Proof. Let λ be a right eigenvalue of A. Then, $\omega(\lambda)$ is a right eigenvalue of the matrix $\omega(A)$, so that there exists some nonzero $x \in \mathbb{H}^n$ such that $\omega(A)x = x\omega(\lambda)$. Let $x := [x_1, \ldots, x_n]^T \in \mathbb{H}^n$ and let x_t be an element of x such that $|x_t| \ge |x_i| \forall i \ (1 \le i \le n)$. Then $|x_t| > 0$. Thus from $\omega(A)x = xw(\lambda)$, we have

$$(\omega(A))_{tt}x_t + \sum_{j=1, j \neq t}^n (\omega(A))_{tj}x_j = x_t\omega(\lambda).$$

Since $x_t \neq 0$, consider $\rho \in \mathbb{H}$ such that $x_t \lambda = \rho x_t$. Then $x_t \lambda^j = \rho^j x_t$, where $j \in S$; the set of integers when $A \in M_n(\mathbb{H})$ is invertible; otherwise S is the set of nonnegative integers. Thus

$$(\omega(A))_{tt}x_t + \sum_{j=1, j \neq t}^n (\omega(A))_{tj}x_j = \omega(\rho)x_t,$$

which implies

$$(\omega(\rho) - \omega(A)_{tt})x_t = \sum_{j=1, j \neq t}^n (\omega(A))_{tj}x_j.$$

Since $|x_t| \ge |x_i| \ \forall i \ (1 \le i \le n)$, we obtain

$$|\omega(\rho) - (\omega(A))_{tt}| \le \sum_{j=1, j \ne t}^{n} |(\omega(A))_{tj}| := r_t(\omega(A)).$$

Remark 3.9. It is known that if λ is a right eigenvalue of $A \in M_n(\mathbb{H})$, then λ^k (k is an integer when A is nonsingular, whereas k is a nonzero positive integer when A is singular) is a right eigenvalue of A^k . Thus similar results to Theorems 3.2 and 3.3 can be obtained for right eigenvalues of $A = (a_{ij}) \in M_n(\mathbb{H})$, where $(\omega(A))_{ii} \in \mathbb{R}$ for all $i \ (1 \leq i \leq n)$ with $\omega(z)$ as defined in (3.13). Since similar quaternionic matrices have the same right eigenvalues. Thus we can obtain potentially sharper inclusion sets by applying similar results to the matrix WAW^{-1} or $\omega(WAW^{-1}) = W\omega(A)W^{-1}$, where W is any invertible real matrix.

3.4. Inclusion regions for zeros of quaternionic polynomials

Inclusion regions for zeros of the simple monic polynomials $p_l(z)$ and $p_r(z)$ are discussed in this section. We first define the following notation:

 $\xi_1^{(\gamma)} :=$ largest element of the set

$$\{|q_0|^{1-\gamma}, (1+|q_1|)^{1-\gamma}, \dots, (1+|q_{m-2}|)^{1-\gamma}\},\$$

 $\xi_2^{(\gamma)} :=$ second largest element of the set

$$\{|q_0|^{1-\gamma}, (1+|q_1|)^{1-\gamma}, \dots, (1+|q_{m-2}|)^{1-\gamma}\},\$$

$$\xi_3^{(\gamma)} := \left(\sum_{j=0}^{m-2} |q_j|\right)^{\gamma}, \text{ and } \xi_4^{(\gamma)} := \left(\sum_{j=0}^{m-3} |q_j|\right)^{\gamma},\$$

where $\gamma \in [0, 1]$.

Then by applying Theorems 2.3 and 2.8 to C_{p_l} , we have the following result.

Theorem 3.10. Let $p_l(z)$ be a simple monic polynomial over \mathbb{H} of degree m and let $\gamma \in [0, 1]$. Then all the zeros of $p_l(z)$ are contained in each of the following sets:

$$T^{(\gamma)} := \{ z \in \mathbb{H} : |z| \le \xi_1^{(\gamma)} \} \cup \{ z \in \mathbb{H} : |z + q_{m-1}| \le \xi_3^{(\gamma)} \}, and \\ K^{(\gamma)} := \{ z \in \mathbb{H} : |z| \le \sqrt{\xi_1^{(\gamma)} \xi_2^{(\gamma)}} \} \cup \{ z \in \mathbb{H} : |z| |z + q_{m-1}| \le \xi_1^{(\gamma)} \xi_3^{(\gamma)} \}.$$

Each of sets $T^{(\gamma)}$ is the union of two balls, whereas the set $K^{(\gamma)}$ for each $\gamma \in [0, 1]$ is the union of a ball and an oval of Cassini. Also, $K^{(\gamma)} \subseteq T^{(\gamma)} \forall \gamma \in [0, 1]$ which was proved in Theorem 2.12.

Case 1. We consider $\omega(z)$ as:

(3.14)
$$\omega(z) := \sum_{j \in S} \alpha_j z^j,$$

where $\alpha_j, z \in \mathbb{H}$ and S is defined as the set of integers when $A \in M_n(\mathbb{H})$ is invertible; otherwise S is the set of nonnegative integers. Now, we take $\omega(z)$ as follows:

$$\omega(z) := z^2 + q_{m-1}z.$$

To find $\omega(C_{p_l})$, we first construct $C_{p_l}^2$;

$$C_{p_l}^2 = \begin{bmatrix} 0 & 0 & 1 & & \\ & & \ddots & \\ & & & 1 \\ -q_0 & -q_1 & -q_2 & \dots & -q_{m-1} \\ q_{m-1}q_0 & q_{m-1}q_1 - q_0 & q_{m-1}q_2 - q_1 & \dots & q_{m-1}^2 - q_{m-2} \end{bmatrix}.$$

Therefore, we obtain

$$C_{p_l}^2 + q_{m-1}C_{p_l} = \begin{bmatrix} 0 & q_{m-1} & 1 & & & \\ 0 & 0 & q_{m-1} & \ddots & & \\ 0 & 0 & 0 & \ddots & 1 & & \\ & & & q_{m-1} & 1 & \\ -q_0 & -q_1 & -q_2 & \dots & -q_{m-2} & 0 \\ 0 & -q_0 & -q_1 & \dots & -q_{m-3} & -q_{m-2} \end{bmatrix}$$

Throughout this chapter for $\gamma \in [0, 1]$, we adopt the following notation.

$$\beta_{1}^{(\gamma)} := |q_{0}|^{(1-\gamma)} (1+|q_{m-1}|)^{\gamma},$$

$$\beta_{2}^{(\gamma)} := (|q_{0}|+|q_{1}|+|q_{m-1}|)^{(1-\gamma)} (1+|q_{m-1}|)^{\gamma},$$

$$\beta_{j}^{(\gamma)} := (|q_{j-2}|+|q_{j-1}|+|q_{m-1}|+1)^{(1-\gamma)} (1+|q_{m-1}|)^{\gamma}; (3 \le j \le m-2),$$

$$\beta_{m-1}^{(\gamma)} := \xi_{4}^{(\gamma)} (|q_{m-1}|+|q_{m-3}|+1)^{(1-\gamma)}.$$

(3.15) $\begin{cases} \eta_1^{(\gamma)} \text{ and } \eta_2^{(\gamma)} \text{ be the largest and second largest elements of a set } S, \\ \text{where } S := \{\beta_1^{(\gamma)}, \beta_2^{(\gamma)}, \dots, \beta_{m-2}^{(\gamma)}\}, \\ \eta_3^{(\gamma)} = \frac{1}{2} \left(|q_{m-2}| + \sqrt{|q_{m-2}|^2 + 4\eta_1^{(\gamma)}\beta_{m-1}^{(\gamma)}} \right). \end{cases}$

By applying Theorem 3.2 to $\omega(C_{p_l})$, we have the following theorem which derives inclusion regions for zeros of $p_l(z)$.

Theorem 3.11. Let $p_l(z)$ be a simple monic polynomial over \mathbb{H} of degree m and let $\gamma \in [0, 1]$. Then all the zeros of $p_l(z)$ are contained in $\mathcal{G}_1^{(\gamma)} \cup \mathcal{G}_2^{(\gamma)}$, where

$$\mathcal{G}_{1}^{(\gamma)} := \{ z \in \mathbb{H} : |\omega(z)| \le \eta_{1}^{(\gamma)} \}, \\ \mathcal{G}_{2}^{(\gamma)} := \{ z \in \mathbb{H} : |\omega(z) + q_{m-2}| \le \beta_{m-1}^{(\gamma)} \}$$

and $\omega(z) := z^2 + q_{m-1}z$ with $z, q_{m-1} \in \mathbb{H}$.

Proof. Applying Theorem 3.2 to $\omega(C_{p_l})$ and using $\eta_1^{(\gamma)}$ and $\beta_{m-1}^{(\gamma)}$, we have the desired result.

The union of $\mathcal{G}_1^{(\gamma)}$ and $\mathcal{G}_2^{(\gamma)}$ can easily be enclosed in one of the two ovals of Cassini which is as follows.

Theorem 3.12. Let $p_l(z)$ be a simple monic polynomial over \mathbb{H} of degree m and let $\gamma \in [0,1]$. Then all the zeros of $p_l(z)$ are contained in each of the following ovals of Cassini:

$$\mathcal{K}_{1}^{(\gamma)} := \{ z \in \mathbb{H} : |z| \ |z + q_{m-1}| \le \max\{\eta_{1}^{(\gamma)}, |q_{m-2}| + \beta_{m-1}^{(\gamma)}\} \},\$$
$$\mathcal{K}_{2}^{(\gamma)} := \{ z \in \mathbb{H} : |z^{2} + q_{m-1}z + q_{m-2}| \le \max\{|q_{m-2}| + \eta_{1}^{(\gamma)}, \beta_{m-1}^{(\gamma)}\} \}.$$

Proof. To prove the theorem, we prove the following conditions:

$$\mathcal{G}_1^{(\gamma)} \cup \mathcal{G}_2^{(\gamma)} \subseteq \mathcal{K}_1^{(\gamma)} \text{ and } \mathcal{G}_1^{(\gamma)} \cup \mathcal{G}_2^{(\gamma)} \subseteq \mathcal{K}_2^{(\gamma)}$$

Clearly $\mathcal{G}_{1}^{(\gamma)} \subseteq \mathcal{K}_{1}^{(\gamma)}$. Consider $z \in \mathcal{G}_{2}^{(\gamma)}$ implies $|z^{2} + q_{m-1}z + q_{m-2}| \leq \beta_{m-1}^{(\gamma)}$. Then $|z^{2} + q_{m-1}z| \leq |q_{m-2}| + \beta_{m-1}^{(\gamma)}$. This implies $z \in \mathcal{K}_{1}^{(\gamma)}$. Hence $\mathcal{G}_{2}^{(\gamma)} \subseteq \mathcal{K}_{1}^{(\gamma)}$. Thus from the above, we have $\mathcal{G}_{1}^{(\gamma)} \cup \mathcal{G}_{2}^{(\gamma)} \subseteq \mathcal{K}_{1}^{(\gamma)}$.

For the 2nd part: It is clear that $\mathcal{G}_2^{(\gamma)} \subseteq \mathcal{K}_2^{(\gamma)}$. Let $z \in \mathcal{G}_1^{(\gamma)}$. Then $|z^2 + q_{m-1}z| \leq \eta_1^{(\gamma)}$. This shows that $|z^2 + q_{m-1}z + q_{m-2}| - |q_{m-2}| \leq \eta_1^{(\gamma)}$. This implies $|z^2 + q_{m-1}z + q_{m-2}| \leq |q_{m-2}| + \eta_1^{(\gamma)}$. Hence $z \in \mathcal{K}_2^{(\gamma)}$ implies $\mathcal{G}_1^{(\gamma)} \subseteq \mathcal{K}_2^{(\gamma)}$. Therefore, $\mathcal{G}_1^{(\gamma)} \cup \mathcal{G}_2^{(\gamma)} \subseteq \mathcal{K}_2^{(\gamma)}$. We give the following example to illustrate our theory.

Example 3.13. Consider the following quaternionic polynomial $p_l(z)$:

$$p_l(z) = z^6 + \mathbf{j}z^5 + \mathbf{i}z^4 - z^2 - \mathbf{j}z - \mathbf{i}.$$

The zeros of $p_l(z)$ are given in [19] as follows:

$$Z_{\mathbb{H}}(p_l) = \{1, -1, [\mathbf{i}], \frac{1}{2}(1 - \mathbf{i} - \mathbf{j} - \mathbf{k}), \frac{1}{2}(-1 + \mathbf{i} - \mathbf{j} - \mathbf{k})\}.$$

The corresponding companion matrix to $p_l(z)$ is given as

$$C_{p_l} = \begin{bmatrix} 0 & 1 & 0 & 0 & 0 & 0 \\ 0 & 0 & 1 & 0 & 0 & 0 \\ 0 & 0 & 0 & 1 & 0 & 0 \\ 0 & 0 & 0 & 0 & 1 & 0 \\ 0 & 0 & 0 & 0 & 0 & 1 \\ \mathbf{i} & \mathbf{j} & 1 & 0 & -\mathbf{i} & \mathbf{j} \end{bmatrix}$$

•

We have the following expressions:

(3.16)
$$\beta_1^{(\gamma)} = 2^{\gamma},$$

(3.17)
$$\beta_2^{(\gamma)} = \beta_4^{(\gamma)} = \beta_5^{(\gamma)} = \eta_2^{(\gamma)} = 3^{1-\gamma} 2^{\gamma},$$

(3.18)
$$\beta_3^{(\gamma)} = \eta_1^{(\gamma)} = 4^{1-\gamma} 2^{\gamma},$$

(3.19)
$$\eta_3^{(\gamma)} = \frac{1}{2} \left[1 + \sqrt{1 + 2^{5 - 2\gamma} 3^{\gamma}} \right].$$

Substituting $\gamma = 1/2$ in Theorem 3.12, then we have the following ovals of Cassini:

$$\mathcal{K}_{1}^{(1/2)} := \{ z \in \mathbb{H} : |z| |z + \mathbf{j}| \le 3.4495 \},\$$

$$\mathcal{K}_{2}^{(1/2)} := \{ z \in \mathbb{H} : |z^{2} + \mathbf{j}z + \mathbf{i}| \le 3.8284 \}.\$$

From the above, it is clear that all the zeros of $p_l(z)$ are contained in the ovals of Cassini $\mathcal{K}_1^{(1/2)}$ and $\mathcal{K}_2^{(1/2)}$.

We now present the following inclusion regions for zeros of $p_l(z)$ by applying Theorem 3.3.

Theorem 3.14. Let $p_l(z)$ be a simple monic polynomial over \mathbb{H} of degree m and let $\gamma \in [0,1]$. Then all the zeros of $p_l(z)$ are contained in $\Gamma_1^{(\gamma)} \cup \Gamma_2^{(\gamma)} \cup \Gamma_3^{(\gamma)}$, where

$$\Gamma_{1}^{(\gamma)} := \{ z \in \mathbb{H} : |\omega(z)| \le \sqrt{\eta_{1}^{(\gamma)} \eta_{2}^{(\gamma)}} \},
\Gamma_{2}^{(\gamma)} := \{ z \in \mathbb{H} : |\omega(z) + q_{m-2}| \le \xi_{4}^{(\gamma)} \sqrt{\beta_{m-1}^{(\gamma)}} \},
\Gamma_{3}^{(\gamma)} := \{ z \in \mathbb{H} : |\omega(z)| \ |\omega(z) + q_{m-2}| \le \eta_{1}^{(\gamma)} \beta_{m-1}^{(\gamma)} \} \\$$

and $\omega(z) = z^2 + q_{m-1}z$ with $z, q_{m-1} \in \mathbb{H}$.

Proof. Applying Theorem 3.3 to $\omega(z)$, where $\omega(z) = z^2 + q_{m-1}z$ and from the definitions of $\eta_1^{(\gamma)}, \eta_2^{(\gamma)}, \xi_4^{(\gamma)}$, and $\beta_{m-1}^{(\gamma)}$, we have the desired result.

Before deriving next inclusion regions for the zeros of the simple monic polynomial $p_l(z)$, we require the following lemma.

Lemma 3.15. Let Υ_1, Υ_2 , and Υ_3 be the three sets such that

$$\begin{split} \Upsilon_1 &:= \{ z \in \mathbb{H} : |z| \, |z+a| \le \delta \}, \ \Upsilon_2 := \{ z \in \mathbb{H} : |z|^2 - |a| \, |z| \le \delta \}, \ and \\ \Upsilon_3 &:= \{ z \in \mathbb{H} : |z| \le \frac{1}{2} (|a| + \sqrt{|a|^2 + 4\delta}) \}, \end{split}$$

where $a \in \mathbb{H}$ and $\delta > 0$. Then $\Upsilon_1 \subseteq \Upsilon_2 := \Upsilon_3$.

Proof. Consider $z \in \Upsilon_1$. Then, $|z||z+a| \leq \delta$ implies $|z|^2 - |a||z| \leq \delta$. Hence $z \in \Upsilon_2$. This shows $\Upsilon_1 \subseteq \Upsilon_2$. Next, we assume that $z \in \Upsilon_2 \Leftrightarrow |z|^2 - |a||z| \leq \delta$, then

(3.20)
$$\left[|z| - \frac{1}{2} \left(|a| + \sqrt{|a|^2 + 4\delta} \right) \right] \left[|z| - \frac{1}{2} \left(|a| - \sqrt{|a|^2 + 4\delta} \right) \right] \le 0.$$
Since $|z| \ge \frac{1}{2} \left(|a| - \sqrt{|a|^2 + 4\delta} \right)$, then from (3.20), we have $|z| \le \frac{1}{2} \left(|a| + \sqrt{|a|^2 + 4\delta} \right).$

Hence $z \in \Upsilon_3$. Thus $\Upsilon_2 = \Upsilon_3$.

We give the following inclusion sets for the zeros of the quaternionic polynomial $p_l(z)$.

Theorem 3.16. Let $p_l(z)$ be a simple monic polynomial over \mathbb{H} of degree m and let $\gamma \in [0,1]$. Then all the zeros of $p_l(z)$ are contained in each of the following ovals of Cassini:

$$\Omega_1^{(\gamma)} := \{ z \in \mathbb{H} : |z| \ |z + q_{m-1}| \le \max\{\sqrt{\eta_1^{(\gamma)}\eta_2^{(\gamma)}}, \eta_3^{(\gamma)}, |q_{m-2}| + \sqrt{\beta_{m-1}^{(\gamma)}} \} \},\$$

$$\Omega_2^{(\gamma)} := \{ z \in \mathbb{H} : |z^2 + q_{m-1}z + q_{m-2}| \le \max\{|q_{m-2}| + \sqrt{\eta_1^{(\gamma)}\eta_2^{(\gamma)}}, \eta_3^{(\gamma)}, \sqrt{\beta_{m-1}^{(\gamma)}} \} \},$$

where $\eta_1^{(\gamma)}, \eta_2^{(\gamma)}$, and $\eta_3^{(\gamma)}$ are defined in (3.15).

Proof. To prove the theorem, we prove the following conditions:

$$\Gamma_1^{(\gamma)} \cup \Gamma_2^{(\gamma)} \cup \Gamma_3^{(\gamma)} \subseteq \Omega_1^{(\gamma)} \text{ and } \Gamma_1^{(\gamma)} \cup \Gamma_2^{(\gamma)} \cup \Gamma_3^{(\gamma)} \subseteq \Omega_2^{(\gamma)}.$$

Clearly $\Gamma_1^{(\gamma)} \subseteq \Omega_1^{(\gamma)}$. Consider $z \in \Gamma_2^{(\gamma)}$, this implies $|z^2 + q_{m-1}z + q_{m-2}| \leq \beta_{m-1}^{(\gamma)}$. This shows that $|z^2 + q_{m-1}z| \leq |q_{m-2}| + \beta_{m-1}^{(\gamma)}$. Then $z \in \Omega_1^{(\gamma)}$. Hence $\Gamma_2^{(\gamma)} \subseteq \Omega_1^{(\gamma)}$.

Next we assume $z \in \Gamma_3^{(\gamma)}$. This implies $|\omega(z)| |\omega(z) + q_{m-2}| \leq \eta_1^{(\gamma)} \beta_{m-1}^{(\gamma)}$. Hence $(|\omega(z)|^2 - |\omega(z)||q_{m-2}|) \leq \eta_1^{(\gamma)} \beta_{m-1}^{(\gamma)}$. Now, from Lemma 3.15, we obtain $|\omega(z)| \leq \eta_3^{(\gamma)}$. Hence $z \in \Omega_1^{(\gamma)}$. Thus from the above, we have $\Gamma_1^{(\gamma)} \cup \Gamma_2^{(\gamma)} \cup \Gamma_3^{(\gamma)} \subseteq \Omega_1^{(\gamma)}$.

On the other hand, it is clear that $\Gamma_2^{(\gamma)} \subseteq \Omega_2^{(\gamma)}$. Considering $z \in \Gamma_1^{(\gamma)}$ it implies $|z^2 + q_{m-1}z| \leq \sqrt{\eta_1^{(\gamma)}\eta_2^{(\gamma)}}$. Then we write $|z^2 + q_{m-1}z + q_{m-2} - q_{m-2}| \leq \sqrt{\eta_1^{(\gamma)}\eta_2^{(\gamma)}}$. This shows $|z^2 + q_{m-1}z + q_{m-2}| \leq |q_{m-2}| + \sqrt{\eta_1^{(\gamma)}\eta_2^{(\gamma)}}$ which implies $z \in \Omega_2^{(\gamma)}$. Thus $\Gamma_1^{(\gamma)} \subseteq \Omega_2^{(\gamma)}$. Now, let $z \in \Gamma_3^{(\gamma)}$. Then $|\omega(z)| |\omega(z) + q_{m-2}| \leq \eta_1^{(\gamma)}\beta_{m-1}^{(\gamma)}$. Thus

$$(|\omega(z) + q_{m-2}| - |q_{m-2}|)|\omega(z) + q_{m-2}| \le \eta_1^{(\gamma)}\beta_{m-1}^{(\gamma)}.$$

From the above discussion and by Lemma 3.15, we have

$$|\omega(z) + q_{m-2}| \le \eta_3^{(\gamma)}.$$

Thus, we obtain $z \in \Omega_2^{(\gamma)}$. Therefore $\Gamma_1^{(\gamma)} \cup \Gamma_2^{(\gamma)} \cup \Gamma_3^{(\gamma)} \subseteq \Omega_2^{(\gamma)}$.

To illustrate Theorem 3.16, we take Example 3.13. Substituting $\gamma = 1/2$ in Theorem 3.16, we have the following ovals of Cassini from Example 3.13:

$$\Omega_1^{(1/2)} = \{ z \in \mathbb{H} : |z| \ |z + \mathbf{j}| \le 6.1723 \}, \text{ and}$$
$$\Omega_2^{(1/2)} = \{ z \in \mathbb{H} : |z^2 + \mathbf{j}z + \mathbf{i}| \le 7.1723 \}.$$

Here it is also clear that all the zeros of $p_l(z)$ (defined in Example 3.13) are located in the ovals of Cassini $\Omega_1^{(1/2)}$ and $\Omega_2^{(1/2)}$.

Theorems 3.12 and 3.16 give the following corollary.

Corollary 3.17. Let $p_l(z)$ be a simple monic polynomial over \mathbb{H} of degree m. Then all the zeros of $p_l(z)$ are contained in ovals of Cassini $\mathcal{K}_1^{(\gamma)} \cap \Omega_1^{(\gamma)} \,\forall \gamma \in [0, 1]$ and also in ovals of Cassini $\mathcal{K}_2^{(\gamma)} \cap \Omega_2^{(\gamma)} \,\forall \gamma \in [0, 1].$ Inclusion regions for the zeros of the quaternionic polynomial $p_r(z)$: Since left eigenvalues of $C_{p_l}^T = C_{p_r}$ are the zeros of the simple monic polynomial $p_r(z)$. We also need some machinery for the development of inclusion regions for the zeros of $p_r(z)$. Now, let us take $\omega(z)$ as

$$\omega(z) := z^2 + \overline{q_{m-1}} \ z,$$

where $z, q_{m-1} \in \mathbb{H}$. To find $\omega_3(C_{p_l})$, we first construct $\overline{C_{p_l}}^2$:

$$\overline{C_{p_l}}^2 = \begin{bmatrix} 0 & 0 & 1 & & \\ & & \ddots & \\ & & & 1 \\ -\overline{q_0} & -\overline{q_1} & -\overline{q_2} & \dots & -\overline{q_{m-1}} \\ \overline{q_{m-1}} \ \overline{q_0} & \overline{q_{m-1}} \ \overline{q_1} - \overline{q_0} & \overline{q_{m-1}} \ \overline{q_2} - \overline{q_1} & \dots & \overline{q_{m-1}}^2 - \overline{q_{m-2}} \end{bmatrix}$$

Thus, we have

$$\overline{C_{p_l}}^2 + \overline{q_{m-1}} \ \overline{C_{p_l}} = \begin{bmatrix} 0 & \overline{q_{m-1}} & 1 & & \\ 0 & 0 & \overline{q_{m-1}} & \ddots & & \\ 0 & 0 & 0 & \ddots & 1 & \\ & & & \overline{q_{m-1}} & 1 \\ & & & & \overline{q_{m-1}} & 1 \\ -\overline{q_0} & -\overline{q_1} & -\overline{q_2} & \dots & -\overline{q_{m-2}} & 0 \\ 0 & -\overline{q_0} & -\overline{q_1} & \dots & -\overline{q_{m-3}} & -\overline{q_{m-2}} \end{bmatrix}$$

It is known from Lemma 2.1 that if λ is a left eigenvalue of a quaternionic matrix A, then $\overline{\lambda}$ is a left eigenvalue of A^H . Thus if λ is a left eigenvalue of the companion matrix C_{p_r} , then $\overline{\lambda}$ is a left eigenvalue of $C_{p_r}^H = \overline{C_{p_l}}$. Therefore $\overline{\lambda}^2 + \overline{q_{m-1}} \ \overline{\lambda}$ is a left eigenvalue of $\overline{C_{p_l}}^2 + \overline{q_{m-1}} \ \overline{C_{p_l}}$.

Remark 3.18. Almost similar results can be obtained for the zeros of the simple monic polynomial $p_r(z)$ with $\omega(z) := z^2 + \overline{q_{m-1}} z$ by using the ideas given in this section for the zeros of the simple monic polynomial $p_l(z)$.

3.4.1. Inclusion regions for zeros of complex polynomials

Case 3. Let us consider $\omega(z)$ as:

(3.21)
$$\omega(z) := \sum_{j \in S} \alpha_j z^j,$$

where $\alpha_j, z \in \mathbb{C}$ and S is defined as the set of integers when $A \in M_n(\mathbb{C})$ is invertible; otherwise S is the set of nonnegative integers. Throughout this section, we use the polynomial $\omega(z)$ (defined in (3.21)).

We take $p_l(z)$ and $p_r(z)$ over the complex field. Then $p_l(z) = p_r(z) = p(z)$ (say). Hence the complex polynomial is defined as

(3.22)
$$p(z) := q_m z^m + q_{m-1} z^{m-1} + \dots + q_1 z + q_0,$$

where q_j , $z \in \mathbb{C}$ $(0 \le j \le m)$. The polynomial (3.22) is called monic if $q_m = 1$. Then, the corresponding companion matrix of the monic polynomial p(z) is given by

$$C_p := \begin{bmatrix} 0 & 1 & & 0 \\ \vdots & & \ddots & \\ 0 & 0 & & 1 \\ -q_0 & -q_1 & \dots & -q_{m-1} \end{bmatrix}$$

Remark 3.19. Similar results to Theorems 2.3 and 2.8 can be obtained in the case of complex matrices.

Remark 3.20. Similar results to Theorems 3.2 and 3.3 can be obtained in the case of complex matrices with $\omega(z)$.

Remark 3.21. Similar results to Theorems 3.10, 3.11, 3.12, 3.14, and 3.16 can be obtained for the zeros of the complex monic polynomial p(z).

Now, we have the following corollary for the zeros of complex polynomials from Corollary 3.17.

Corollary 3.22. Let p(z) be a monic polynomial over \mathbb{C} of degree m. Then all the zeros of p(z) are contained in oval of Cassini $\mathcal{K}_1^{(\gamma)} \cap \Omega_1^{(\gamma)} \ \forall \gamma \in [0, 1]$ and also in oval of Cassini $\mathcal{K}_2^{(\gamma)} \cap \Omega_2^{(\gamma)} \ \forall \gamma \in [0, 1].$

We now state the following results from the above remarks which are available in the literature.

• Assuming $\gamma = 0$ in Theorem 3.10 for the case of complex monic polynomial p(z), we observe that all the zeros of p(z) are contained in each of the following sets:

$$T^{(0)} := \{ z \in \mathbb{C} : |z| \le \xi_1^{(0)} \} \cup \{ z \in \mathbb{C} : |z + q_{m-1}| \le 1 \},\$$

$$K^{(0)} := \{ z \in \mathbb{C} : |z| \le \sqrt{\xi_1^{(0)} \xi_2^{(0)}} \} \cup \{ z \in \mathbb{C} : |z| |z + q_{m-1}| \le \xi_1^{(0)} \}.$$

This can be seen in [37, Theorem 3.1].

• Substituting $\gamma = 0$ in Theorem 3.11 for the case of complex monic polynomial p(z), we obtain that all the zeros of p(z) are contained in $\mathcal{G}_1^{(0)} \cup \mathcal{G}_2^{(0)}$, where

$$\mathcal{G}_1^{(0)} := \{ z \in \mathbb{C} : |\omega(z)| \le \eta_1^{(0)} \}, \ \mathcal{G}_2^{(0)} := \{ z \in \mathbb{C} : |\omega(z) + q_{m-2}| \le \beta_{m-1}^{(0)} \}$$

with $\omega(z) := z^2 + q_{m-1}z, q_{m-1}, z \in \mathbb{C}$. This can be found in [37, Theorem 3.2].

• Let $\gamma = 0$ in Theorem 3.12 for the case of complex monic polynomial p(z), we see that all the zeros of p(z) are contained in the oval of Cassini defined by

$$\mathcal{K}_1^{(0)} := \{ z \in \mathbb{C} : |z| \ |z + q_{m-1}| \le \max\{\eta_1^{(0)}, |q_{m-2}| + \beta_{m-1}^{(0)}\} \},\$$

and also in the oval of Cassini defined as

$$\mathcal{K}_2^{(0)} := \{ z \in \mathbb{C} : |z^2 + q_{m-1}z + q_{m-2}| \le \max\{|q_{m-2}| + \eta_1^{(0)}, \beta_{m-1}^{(0)}\} \}.$$

This can be seen in [37, Theorem 3.3].

• Assuming $\gamma = 0$ in Theorem 3.14 for the case of complex monic polynomial p(z), we obtain that all the zeros of p(z) are contained in $\Gamma_1^{(0)} \cup \Gamma_2^{(0)} \cup \Gamma_3^{(0)}$, where

$$\Gamma_{1}^{(0)} := \{ z \in \mathbb{C} : |\omega(z)| \le \sqrt{\eta_{1}^{(0)} \eta_{2}^{(0)}} \},
\Gamma_{2}^{(0)} := \{ z \in \mathbb{C} : |\omega(z) + q_{m-2}| \le \sqrt{\beta_{m-1}^{(0)}} \},
\Gamma_{3}^{(0)} := \{ z \in \mathbb{C} : |\omega(z)| |\omega(z) + q_{m-2}| \le \eta_{1}^{(0)} \beta_{m-1}^{(0)} \} \\$$

with $\omega(z) := z^2 + q_{m-1}z, q_{m-1}, z \in \mathbb{C}$. This can be found in [37, Theorem 3.4].

• Assuming $\gamma = 0$ in Theorem 3.16, then all the zeros of the complex monic polynomial p(z) are contained in the oval of Cassini defined as

$$\Omega_1^{(0)} := \{ z \in \mathbb{C} : |z| \ |z + q_{m-1}| \le \max\{\sqrt{\eta_1^{(0)} \eta_2^{(0)}}, \eta_3^{(0)}, |q_{m-2}| + \sqrt{\beta_{m-1}^{(0)}} \} \},\$$

and also in the oval of Cassini given by

$$\Omega_2^{(0)} := \{ z \in \mathbb{C} : |z^2 + q_{m-1}z + q_{m-2}| \le \max\{|q_{m-2}| + \sqrt{\eta_1^{(0)}\eta_2^{(0)}}, \eta_3^{(0)}, \sqrt{\beta_{m-1}^{(0)}} \} \},$$

where
$$\eta_3^{(0)} := \frac{1}{2} \left(|q_{m-2}| + \sqrt{|q_{m-2}|^2 + 4\eta_1^{(0)}\beta_{m-1}^{(0)}} \right)$$
. This is stated in [37, Theorem 3.5].

Comparisons with the existing results: The inclusion regions $T^{(0)}, K^{(0)}, G_1^{(0)}, G_2^{(0)}, \mathcal{K}_1^{(0)}, \mathcal{K}_2^{(0)}, \Gamma_1^{(0)}, \Gamma_2^{(0)}, \Gamma_3^{(0)}, \Omega_1^{(0)}, \Omega_2^{(0)}$ are also given in [37]. Now, we compare these inclusion regions with $T^{(\gamma)}, G_1^{(\gamma)}, G_2^{(\gamma)}, \mathcal{K}_1^{(\gamma)}, \mathcal{K}_2^{(\gamma)}, \Gamma_1^{(\gamma)}, \Gamma_2^{(\gamma)}, \Gamma_3^{(\gamma)}, \Omega_1^{(\gamma)}, \Omega_2^{(\gamma)} \forall \gamma \in [0, 1]$, under certain conditions which are as follows:

- (a) Let $\xi_3^{(\gamma)} \leq 1 \ \forall \gamma \in [0,1]$. Since $\xi_1^{(\gamma)} \leq \xi_1^{(0)} \ \forall \gamma \in [0,1]$, then $T^{(\gamma)} \subseteq T^{(0)} \ \forall \gamma \in [0,1]$. Further, we have seen that $\xi_1^{(\gamma)} \xi_3^{(\gamma)} \leq \xi_1^{(0)}$ and $\xi_2^{(\gamma)} \leq \xi_2^{(0)} \ \forall \gamma \in [0,1]$, then $K^{(\gamma)} \subseteq K^{(0)} \ \forall \gamma \in [0,1]$.
- (b) If $1 + |q_{m-1}| \leq \eta_1^{(0)}$, then $\eta_1^{(\gamma)} \leq \eta_1^{(0)} \forall \gamma \in [0, 1]$. Further, if $\xi_4^{(\gamma)} \leq 1 \forall \gamma \in [0, 1]$, then $\beta_{m-1}^{(\gamma)} \leq \beta_{m-1}^{(0)}$. Consequently, $\mathcal{G}_1^{(\gamma)} \subseteq \mathcal{G}_1^{(0)}$ and $\mathcal{G}_2^{(\gamma)} \subseteq \mathcal{G}_2^{(0)} \forall \gamma \in [0, 1]$.
- (c) Considering $1 + |q_{m-1}| \leq \eta_1^{(0)}$ and $\xi_4^{(\gamma)} \leq 1 \ \forall \gamma \in [0,1]$, then $\max\{\eta_1^{(\gamma)}, |q_{m-2}| + \beta_{m-1}^{(\gamma)}\} \leq \max\{\eta_1^{(0)}, |q_{m-2}| + \beta_{m-1}^{(0)}\}$, and $\max\{|q_{m-2}| + \eta_1^{(\gamma)}, \beta_{m-1}^{(\gamma)}\} \leq \max\{|q_{m-2}| + \eta_1^{(0)}, \beta_{m-1}^{(0)}\} \ \forall \gamma \in [0,1]$. Thus $\mathcal{K}_1^{(\gamma)} \subseteq \mathcal{K}_1^{(0)}$ and $\mathcal{K}_2^{(\gamma)} \subseteq \mathcal{K}_2^{(0)} \ \forall \gamma \in [0,1]$.
- (d) Supposing $1 + |q_{m-1}| \leq \eta_1^{(0)}$ and $1 + |q_{m-1}| \leq \eta_2^{(0)}$, then $\eta_1^{(\gamma)} \leq \eta_1^{(0)}$ and $\eta_2^{(\gamma)} \leq \eta_2^{(0)} \forall \gamma \in [0, 1]$. Consequently, $\sqrt{\eta_1^{(\gamma)} \eta_2^{(\gamma)}} \leq \sqrt{\eta_1^{(0)} \eta_2^{(0)}} \forall \gamma \in [0, 1]$. Also if $\xi_4^{(\gamma)} \leq 1 \forall \gamma \in [0, 1]$, then $\xi_4^{(\gamma)} \sqrt{\beta_{m-1}^{(\gamma)}} \leq \sqrt{\beta_{m-1}^{(0)}}$ and $\eta_1^{(\gamma)} \beta_{m-1}^{(\gamma)} \leq \eta_1^{(0)} \beta_{m-1}^{(0)}$. Hence $\Gamma_1^{(\gamma)} \subseteq \Gamma_1^{(0)}$, $\Gamma_2^{(\gamma)} \subseteq \Gamma_2^{(0)}$, and $\Gamma_3^{(\gamma)} \subseteq \Gamma_3^{(0)} \forall \gamma \in [0, 1]$.
- (e) If $1 + |q_{m-1}| \leq \eta_1^{(0)}$, then $\eta_1^{(\gamma)} \leq \eta_1^{(0)}$. Similarly, if $\xi_4^{(\gamma)} \leq 1 \ \forall \ \gamma \in [0, 1]$, then $\beta_{m-1}^{(\gamma)} \leq \beta_{m-1}^{(0)}$. Consequently, $\eta_3^{(\gamma)} \leq \eta_3^{(0)}$. Hence $\Omega_1^{(\gamma)} \subseteq \Omega_1^{(0)}$ and $\Omega_2^{(\gamma)} \subseteq \Omega_2^{(0)} \ \forall \ \gamma \in [0, 1]$.

We see the following observations:

- If $1 + |q_{m-1}| \leq \eta_1^{(0)}$ and $\xi_4^{(\gamma)} \leq 1 \ \forall \gamma \in [0, 1]$, then $\mathcal{K}_1^{(\gamma)} \subseteq \mathcal{K}_1^{(0)}$, $\mathcal{K}_2^{(\gamma)} \subseteq \mathcal{K}_2^{(0)}$, $\Omega_1^{(\gamma)} \subseteq \Omega_1^{(0)}$, and $\Omega_2^{(\gamma)} \subseteq \Omega_2^{(0)} \ \forall \gamma \in [0, 1]$.
- If $\xi_3^{(\gamma)} \leq (1+\rho)^{\gamma} \forall \gamma \in (0,1]$, then our inclusion regions $K^{(\gamma)} \forall \gamma \in (0,1]$ are sharper than the inclusion region $K^{(0)}$ [37, Theorem 3.1].
- If $\xi_4^{(\gamma)} \leq (1+k\rho)^{\gamma} \ \forall \gamma \in (0,1]; k=2,3$, then ovals of Cassini $\mathcal{K}_1^{(\gamma)} \cap \Omega_1^{(\gamma)} \ \forall \gamma \in (0,1]$ and $\mathcal{K}_2^{(\gamma)} \cap \Omega_2^{(\gamma)} \ \forall \gamma \in (0,1]$ are sharper than the ovals of Cassini [37, Corollary 3.1].

Finally, we compare our results with A. Melman's results given in [37]. Then each of the sets $T^{(\gamma)}$ and $K^{(\gamma)}$ (for the case of complex polynomials) can be shown that they are smaller than the sets $T^{(0)}$ and $K^{(0)}$, respectively. For example, consider the complex polynomial

$$\begin{array}{rcl} (3.23) & p(z) = z^5 + (7 - 15i)z^4 + (14 + 7i)z^3 + 3z^2 + 35z + 2 + 3i. \\ T^{(0)} &= \{z \in \mathbb{C} : |z| \leq 35\} \cup \{z \in \mathbb{C} : |z + 7 - 15i| \leq 1\}, \\ K^{(0)} &= \{z \in \mathbb{C} : |z| \leq \sqrt{(35)(16.6526)}\} \cup \{z \in \mathbb{C} : |z| \mid z + 7 - 15i| \leq 35\}, \\ T^{(\gamma)} &= \{z \in \mathbb{C} : |z| \leq (35)^{1 - \gamma}\} \cup \{z \in \mathbb{C} : |z + 7 - 15i| \leq (58.2580)^{\gamma}\}, \\ K^{(\gamma)} &= \{z \in \mathbb{C} : |z| \leq \sqrt{(35)^{1 - \gamma}(16.6526)^{1 - \gamma}}\} \cup \{z \in \mathbb{C} : |z| \mid z + 7 - 15i| \leq \chi\} \\ \text{where} \quad \chi &= (35)^{1 - \gamma} (58.2580)^{\gamma}. \end{array}$$





FIGURE 3.1. Location of zeros of p(z).

CHAPTER 4

LOCALIZATION THEOREMS FOR QUATERNIONIC MATRIX PENCILS

In this chapter, a general framework for defining and analyzing the generalized right eigenvalues of a quaternionic matrix pencil are developed. Inclusion regions for the generalized right eigenvalues of a quaternionic matrix pencil and their applications are presented.

4.1. Introduction and preliminaries

Quaternionic matrix pencils and their corresponding canonical forms have been derived in [46,47]. In this chapter, we derive localization theorems for generalized right eigenvalues of a quaternionic matrix pencil with some properties. Location of zeros of quaternionic polynomials is presented.

Throughout this chapter, we adopt the following basic facts: Let $\mathbb{L}_1(M_n(\mathbb{H}))$ be the space of matrix pencils over a quaternion division algebra. $\mathbf{L}_1 \in \mathbb{L}_1(M_n(\mathbb{H}))$ is defined as

(4.1)
$$\mathbf{L}_1(\lambda) := A + \lambda B,$$

where $A, B \in M_n(\mathbb{H})$ and λ commutes with the quaternionic matrices. This matrix pencil over a quaternion division algebra can be found in [33, 46–48].

Now we define generalized right eigenvalue of $\mathbf{L}_1 \in \mathbb{L}_1(M_n(\mathbb{H}))$ of the form (4.1) which is as follows.

Definition 4.1. Let $\mathbf{L}_1 \in \mathbb{L}_1(M_n(\mathbb{H}))$ be as in (4.1) and let $\mu \in \mathbb{H}$. Then μ is called a generalized right eigenvalue of the matrix pencil \mathbf{L}_1 if

$$Ax = Bx\mu$$

for some nonzero $x \in \mathbb{H}^n$. Here x is called the right eigenvector corresponding to the generalized right eigenvalue μ . The set of generalized right eigenvalues of \mathbf{L}_1 is called right spectrum of \mathbf{L}_1 , denoted by $\Lambda_r(\mathbf{L}_1)$.

Regular matrix pencil over a quaternion division algebra is defined as follows.

Definition 4.2. Let $\mathbf{L}_1 \in \mathbb{L}_1(M_n(\mathbb{H}))$ be as in (4.1). Then the matrix pencil \mathbf{L}_1 is called regular if there exists $\alpha \in \mathbb{R}$ such that $A + \alpha B$ is an invertible matrix.

Let $\mathbb{P}_1(M_{2n}(\mathbb{C}))$ be the space of complex matrix pencils. $P_1 \in \mathbb{P}_1(M_{2n}(\mathbb{C}))$ is defined as

(4.2)
$$P_1(\mu) := \Psi_A + \mu \Psi_B,$$

where $A, B \in M_n(\mathbb{H})$ and $\mu \in \mathbb{C}$.

4.2. Location of generalized right eigenvalues

We give the following theorem with the help of the complex adjoint matrix.

Theorem 4.3. Let $\mathbf{L}_1 \in \mathbb{L}_1(M_n(\mathbb{H}))$ be as in (4.1). Then the matrix pencil \mathbf{L}_1 is a regular if and only if $P_1 \in \mathbb{P}_1(M_{2n}(\mathbb{C}))$ is a regular complex matrix pencil.

Proof. Consider $\mathbf{L}_1 \in \mathbb{L}_1(M_n(\mathbb{H}))$ is a regular matrix pencil, then there exists $\alpha \in \mathbb{R}$ such that $\mathbf{L}_1(\alpha) = A + \alpha B$ is an invertible matrix pencil. Now from Theorem 1.12, the corresponding complex matrix pencil $\Psi_{\mathbf{L}_1(\alpha)} = \Psi_A + \alpha \Psi_B$ is an invertible complex matrix pencil. Thus, $P_1(\mu)$ is a regular complex matrix pencil.

Conversely let $P_1(\mu)$ be a regular complex matrix pencil, then

det $[P_1(\eta)] \neq 0$ for some $\eta \in \mathbb{C}$, i.e., det $[\Psi_A + \eta \Psi_B] \neq 0$.

Thus, there exists a real number λ_0 (say) such that

$$\det[\Psi_A + \lambda_0 \Psi_B] \neq 0,$$

i.e., $\Psi_A + \lambda_0 \Psi_B = \Psi_{(A+\lambda_0 B)}$ is an invertible complex pencil. Then by Theorem 1.12, the matrix pencil $A + \lambda_0 B$ is invertible. Hence $\mathbf{L}_1 \in \mathbb{L}_1(M_n(\mathbb{H}))$ is a regular quaternionic matrix pencil.

Proposition 4.4. Let $\mathbf{L}_1 \in \mathbb{L}_1(M_n(\mathbb{H}))$ be as in (4.1) and let $\mu \in \mathbb{H}$ be a generalized right eigenvalue of the matrix pencil \mathbf{L}_1 . Let $0 \neq \rho \in \mathbb{H}$. Then $\rho^{-1}\mu\rho$ is a generalized right eigenvalue of \mathbf{L}_1 .

Proof. Suppose μ is a generalized right eigenvalue of the matrix pencil \mathbf{L}_1 corresponding to an eigenvector $x \in \mathbb{H}^n$, then

If $\rho \in \mathbb{H} \setminus \{0\}$, then from (4.3), we have

$$Ax\rho = Bx\rho(\rho^{-1}\mu\rho).$$

Thus $\rho^{-1}\mu\rho$ is also a generalized right eigenvalue of the matrix pencil \mathbf{L}_1 corresponding to an eigenvector $x\rho$.

Complex matrix pencils: Let $A, B \in M_n(\mathbb{C})$. Let $\mathcal{L}(z) := A - zB$ be complex matrix pencil. The complex eigenvalue problem $\mathcal{L}(\lambda)x = 0$, is called the generalized complex eigenvalue problem. Denote the set of $n \times n$ matrix pencils by $\tilde{\mathcal{L}}(M_n(\mathbb{C}))$. $\mathcal{L} \in \tilde{\mathcal{L}}(M_n(\mathbb{C}))$ is said to be regular if $\det(\mathcal{L}(\lambda)) \neq 0$ for some $\lambda \in \mathbb{C}$. The spectrum of a regular matrix pencil \mathcal{L} is given by

$$\Lambda(\mathcal{L}) := \left\{ \lambda \in \mathbb{C} : Ax = \lambda Bx = Bx\lambda \text{ for some nonzero } x \in \mathbb{C}^n \right\}.$$

We define generalized standard right eigenvalue of a regular matrix pencil $\mathbf{L}_1 \in \mathbb{L}_1(M_n(\mathbb{H}))$ as follows.

Definition 4.5. Let $\mathbf{L}_1 \in \mathbb{L}_1(M_n)(\mathbb{H})$ be as in (4.1). Then we define the set of the generalized standard right eigenvalues of a regular matrix pencil \mathbf{L}_1 as

$$\Lambda_s(\mathbf{L}) := \{ \alpha \in \mathbb{C}_{\infty} : Ax = Bx\alpha, \ 0 \neq x \in \mathbb{H}^n, \ \Im(\alpha) \ge 0 \}, \text{ where } \mathbb{C}_{\infty} = \mathbb{C} \cup \{\infty\}.$$

The above definition generalizes the definition of standard right eigenvalues of a single quaternionic matrix [5,27].

It is known that if *B* is not invertible, then Ψ_B is not invertible. Hence $P_1(\mu) = \Psi_A + \mu \Psi_B$ has an infinite eigenvalue. Consequently, $\mathbf{L}_1(\lambda) = A + \lambda B$ has an infinite right eigenvalue. However, if *B* is an invertible matrix, then $P_1(\mu) = \Psi_A + \mu \Psi_B$ has exactly 2n finite eigenvalues. Then, $\mathbf{L}_1(\lambda) = A + \lambda B$ has exactly 2n finite complex right eigenvalues. Thus, we have the following observations:

- If \mathbf{L}_1 is regular, then \mathbf{L}_1 has 2n right eigenvalues in \mathbb{C}_{∞} .
- If B is an invertible matrix, then \mathbf{L}_1 has exactly 2n finite complex right eigenvalues.

We now present the Gerschgorin-type localization of eigenvalues of complex matrix pencils as follows.

Theorem 4.6. [25] Let $\mathcal{L} \in \tilde{\mathcal{L}}$ $(M_n(\mathbb{C}))$ be a complex regular matrix pencil. Then all the eigenvalues of \mathcal{L} are contained in the union of n regions

$$\mathcal{G}_{i}(\mathcal{L}) := \left\{ z \in \mathbb{C} : |zb_{ii} - a_{ii}| \le \sum_{j=1, j \neq i}^{n} |a_{ij} - zb_{ij}| \right\} (1 \le i \le n), i.e.$$
$$\Lambda(\mathcal{L}) \subseteq \bigcup_{i=1}^{n} \mathcal{G}_{i}(\mathcal{L}).$$

Similarly, we define n regions over the skew field of quaternions as follows.

(4.4)
$$G_i(\mathbf{L}_1) := \left\{ z \in \mathbb{H} : |zb_{ii} - a_{ii}| \le \sum_{j=1, j \neq i}^n |a_{ij} - zb_{ij}| \right\} (1 \le i \le n),$$

where $\mathbf{L}_1(\lambda) := A + \lambda B$ is defined in (4.1).

But, in the case of quaternionic matrix pencil, a generalized right eigenvalue is not necessarily contained in the union of n regions $G_i(\mathbf{L}_1)$ as the following example suggest.

Example 4.7. A generalized right eigenvalue is not necessarily contained in a region. Consider the quaternionic matrix pencil $\mathbf{L}_1(\lambda) = A + \lambda B$, where

$$A = \begin{bmatrix} 2\mathbf{i} & 0\\ 0 & 1 \end{bmatrix}, \ B = \begin{bmatrix} 1 & 0\\ 1 & 0 \end{bmatrix}$$

and λ commutes with A and B. For finding generalized right eigenvalue, let

$$\det[\mu\Psi_B - \Psi_A] = 0,$$

where $\mu \in \mathbb{C}$. Then $(\mu + 2\mathbf{i})(\mu - 2\mathbf{i}) = 0$ has two zeros corresponding to the values $\mu = 2\mathbf{i}$ and $-2\mathbf{i}$ with multiplicity 1. From (4.4), we have the following regions:

$$G_1(\mathbf{L}_1) := \{ z \in \mathbb{H} : |z - 2\mathbf{i}| \le 0 \}, \text{ and}$$

 $G_2(\mathbf{L}_1) := \{ z \in \mathbb{H} : |-1| \le 1 \}.$

In particular, $-2\mathbf{i}$ is a generalized right eigenvalue of \mathbf{L}_1 . However it is not contained in any regions $G_i(\mathbf{L}_1)$, i = 1, 2.

Moreover, we present the generalized Gerschgorin type theorem for a quaternionic matrix pencil as follows.

Theorem 4.8. (Generalized Gerschgorin type theorem for generalized right eigenvalues) Let $\mathbf{L}_1 \in \mathbb{L}_1(M_n(\mathbb{H}))$ be a quaternionic regular matrix pencil. For every generalized right eigenvalue μ of \mathbf{L}_1 there exists a nonzero quaternion α such that $\alpha^{-1}\mu\alpha$ (which is also a generalized right eigenvalue) is contained in the union of n regions

$$D_{i}(\mathbf{L}_{1}) := \{ z \in \mathbb{H} : |b_{ii}z - a_{ii}| \le |z| r_{i}(B) + r_{i}(A) \} \ (1 \le i \le n), \ i.e., \\ \{ z^{-1}\mu z : 0 \ne z \in \mathbb{H} \} \cap \cup_{i=1}^{n} D_{i}(\mathbf{L}_{1}) \ne \emptyset.$$

In particular, when μ is real, then it is contained in the union of the regions $D_i(\mathbf{L}_1)$.

Proof. Let λ be a generalized right eigenvalue of the quaternionic matrix pencil \mathbf{L}_1 . Then $Ax = Bx\mu$ for some nonzero $x = [x_1, \ldots, x_n]^T \in \mathbb{H}^n$. Let x_t be an element of x such that $|x_t| \ge |x_i| \forall i \ (1 \le i \le n)$. Then $|x_t| > 0$. From the t-th equation of $Ax = Bx\mu$, we have

$$\sum_{j=1}^{n} a_{tj} x_{j} = \sum_{j=1}^{n} b_{tj} x_{j} \mu$$
$$a_{tt} x_{t} + \sum_{j=1, j \neq t}^{n} a_{tj} x_{j} = b_{tt} x_{t} \mu + \sum_{j=1, j \neq t}^{n} b_{tj} x_{j} \mu$$

Since $x_t \neq 0$, let $x_t \mu = \xi x_t$, i.e., ξ is similar to μ . Then

$$a_{tt}x_t - b_{tt}\xi x_t = \sum_{j=1, j \neq t}^n b_{tj}x_j\mu - \sum_{j=1, j \neq t}^n a_{tj}x_j$$
$$(a_{tt} - b_{tt}\xi)x_t = \sum_{j=1, j \neq t}^n b_{tj}x_j\mu - \sum_{j=1, j \neq t}^n a_{tj}x_j,$$

which yields, by the triangle inequality,

$$\begin{aligned} |a_{tt} - b_{tt}\xi||x_t| &= |\sum_{j=1, j \neq t}^n b_{tj}x_j\mu - \sum_{j=1, j \neq t}^n a_{tj}x_j| \\ &\leq \sum_{j=1, j \neq t}^n |b_{tj}x_j\mu| + \sum_{j=1, j \neq t}^n |a_{tj}x_j| \\ |a_{tt} - b_{tt}\xi||x_t| &\leq \sum_{j=1, j \neq t}^n (|b_{tj}\mu| + |a_{tj}|)|x_j| \\ &|a_{tt} - b_{tt}\xi| &\leq \sum_{j=1, j \neq t}^n (|b_{tj}\mu| + |a_{tj}|) \\ &|a_{tt} - b_{tt}\xi| &\leq |\mu| r_t(B) + r_t(A). \end{aligned}$$

Note that ξ is also a generalized right eigenvalue of \mathbf{L}_1 and $|\mu| = |\xi|$. Hence, ξ lies in the union of *n* regions $D_i(\mathbf{L}_1)$.

Theorem 4.9. Let $\mathbf{L}_1 \in \mathbb{L}_1(M_n(\mathbb{H}))$ be a quaternionic regular matrix pencil and let all the diagonal entries of quaternionic matrices A and B are real. Then all the generalized right eigenvalues of \mathbf{L}_1 are contained in the union of n regions

$$D_i(\mathbf{L}_1) := \{ z \in \mathbb{H} : |b_{ii}z - a_{ii}| \le |z| r_i(B) + r_i(A) \} \ (1 \le i \le n),$$

which means

$$\Lambda_r(\mathbf{L}_1) \subseteq \bigcup_{i=1}^n D_i(\mathbf{L}_1).$$

Proof. The proof is immediate from the proof method of Theorem 4.8 by using the fact that $ab = ba \ \forall a \in \mathbb{R}$ and $\forall b \in \mathbb{H}$, so we skip the proof.

Before going to locate the zeros of quaternionic polynomials, we first recall the quaternionic polynomials $p_l(z)$ and $p_r(z)$ from (1.20) and (1.21) as follows.

(4.5)
$$p_l(z) := q_m z^m + q_{m-1} z^{m-1} + \dots + q_1 z + q_0,$$

(4.6)
$$p_r(z) := z^m q_m + z^{m-1} q_{m-1} + \dots + zq_1 + q_0,$$

where $q_j, z \in \mathbb{H}, (0 \le j \le m)$. The polynomial $p_l(z)$ is associated with the quaternionic matrix pencil

(4.7)
$$T_1(\lambda) := \lambda E - C_{p_l},$$

where λ commutes with the quaternionic matrices

$$C_{p_l} := \begin{bmatrix} 0 & 1 & 0 \\ \vdots & \ddots & \\ 0 & 0 & 1 \\ -q_0 & -q_1 & \dots & -q_{m-1} \end{bmatrix} \text{ and } E := \begin{bmatrix} 1 & 0 & \dots & 0 \\ & \ddots & & \vdots \\ 0 & \dots & 1 & 0 \\ 0 & \dots & \dots & q_m \end{bmatrix}$$

This is known as a linearization of $p_l(z)$. We now turn to find the relation between the zeros of $p_l(z)$ and generalized right eigenvalues of the quaternionic matrix pencil $T_1(\lambda) = \lambda E - C_{p_l}$.

Theorem 4.10. Let $T_1(\lambda)$ be a quaternionic matrix pencil as in (4.7). For every generalized right eigenvalue μ of T_1 there exists a nonzero quaternion β such that $\beta^{-1}\mu\beta$ is a zero of the quaternionic polynomial $p_l(z)$. *Proof.* Let μ be a generalized right eigenvalue of T_1 . Then, there exists some nonzero $x \in \mathbb{H}^n$ such that $C_{p_l}x = Ex\mu$. Let $x := [x_1, \dots, x_m]^T \in \mathbb{H}^n$. Then

(4.8)
$$\begin{bmatrix} 0 & 1 & 0 \\ \vdots & \ddots & \\ 0 & 0 & 1 \\ -q_0 & -q_1 & \dots & -q_{m-1} \end{bmatrix} \begin{bmatrix} x_1 \\ x_2 \\ \vdots \\ x_m \end{bmatrix} = \begin{bmatrix} 1 & 0 & \dots & 0 \\ \ddots & & \vdots \\ 0 & \dots & 1 & 0 \\ 0 & \dots & \dots & q_m \end{bmatrix} \begin{bmatrix} x_1 \\ x_2 \\ \vdots \\ x_m \end{bmatrix} \mu.$$

(4.8) gives the following system of linear equations

$$x_{2} = x_{1}\mu,$$

$$x_{3} = x_{2}\mu,$$

$$\vdots$$

$$x_{m} = x_{m-1}\mu,$$

$$-q_{0}x_{1} - q_{1}x_{2} - q_{2}x_{3} - \dots - q_{m-2}x_{m-1} - q_{m-1}x_{m} = q_{m}x_{m}\mu.$$

By solving the above system of linear equations, we obtain

$$(4.9) \quad -q_0 x_1 - q_1 x_1 \mu - q_2 x_1 \mu^2 - \dots - q_{m-2} x_1 \mu^{m-2} - q_{m-1} x_1 \mu^{m-1} = q_m x_1 \mu^m.$$

Since x cannot be the zero quaternionic vector, $x_m \neq 0$ and hence multiplying by x_1^{-1} on the both sides of (4.9), we obtain

$$-q_0 - q_1 x_1 \mu x_1^{-1} - q_2 x_1 \mu^2 x_1^{-1} - \dots - q_{m-2} x_1 \mu^{m-2} x_1^{-1} - q_{m-1} x_1 \mu^{m-1} x_1^{-1} - q_m x_1 \mu^m x_1^{-1} = 0.$$

Putting $x_1 \mu x_1^{-1} = \rho$, we obtain

$$q_0 + q_1\rho + q_2\rho^2 + \dots + q_{m-2}\rho^{m-2} + q_{m-1}\rho^{m-1} + q_m\rho^m = 0.$$

Hence ρ is a zero of $p_l(z)$.

Let $C_{p_l} := (c_{ij}) \in M_n(\mathbb{H})$ and let $E := (e_{ij}) \in M_n(\mathbb{H})$. We define the following regions.

$$D_i(T_1) := \{ z \in \mathbb{H} : |e_{ii}z - c_{ii}| \le |z| r_i(E) + r_i(C_{p_l}) \} \ (1 \le i \le n),$$

where $T_1(\lambda) := \lambda E - C_{p_l}$ with λ commutes with the quaternionic matrices. Now, we have the following observations: • From Theorem 4.10, if μ is a generalized right eigenvalue of the quaternionic matrix pencil T_1 (associated with $p_l(z)$), then $\beta^{-1}\mu\beta$ ($0 \neq \beta \in \mathbb{H}$) is a zero of $p_l(z)$. From Theorem 4.8, we obtain

$$\beta^{-1}\mu\beta \in \bigcup_{i=1}^{n} D_{i}(T_{1}), \text{ i.e.},$$
$$\{z^{-1}\mu z: 0 \neq z \in \mathbb{H}\} \cap \bigcup_{i=1}^{n} D_{i}(T_{1}) \neq \emptyset.$$

• Let $q_m, q_{m-1} \in \mathbb{R}$ and let μ be a generalized right eigenvalue of T_1 . Then from Theorem 4.9, μ as well as $\beta^{-1}\mu\beta$ (zero of $p_l(z)$) are contained in the union of nregions $D_i(T_1)$.

Remark 4.11. Similar results can be obtained for $p_r(z)$ as well.

CHAPTER 5

BOUNDS FOR EIGENVALUES OF MATRIX POLYNOMIALS OVER A QUATERNION DIVISION ALGEBRA

In this chapter, the definitions of the left and right eigenvalues of quaternionic matrices are extended to quaternionic matrix polynomials. Localization theorems are discussed for the left and right eigenvalues of a quaternionic block matrix. Furthermore, bounds for the absolute values of the left and right eigenvalues of quaternionic matrix polynomials are devised and illustrated for the matrix *p*-norm, where $p = 1, 2, \infty$, and *F* (Frobenius). The above bounds generalize the bounds on the absolute values of the eigenvalues of complex matrix polynomials which give sharper bounds to the existing bounds for the case of 1, 2, and ∞ matrix norms.

5.1. Introduction and preliminaries

The bounds for the absolute values of the eigenvalues of complex matrix polynomials have been described in [15]. The location of the eigenvalues when computed with an iterative method, see e.g. [54], is of pivotal importance. The applications on matrices, matrix pencils, and matrix polynomials over a quaternion division algebra are discussed in [44, 47, 48]. The literature available on the theory of quaternionic matrix polynomials is limited [43, 44] and are restricted to the stability of various systems with quaternionic matrix coefficients. Therefore, it is of prime significance to carry out further research on quaternionic matrix polynomials to characterize the stability of systems with quaternionic matrix coefficients. The stability analysis of a given system depends on the behavior of right eigenvalues of quaternionic matrix polynomials. For understanding the stability of a given system it is required to analyze the location of the right eigenvalues of quaternion matrix polynomials. There are two types of eigenvalues for the case of matrix pencils and matrix polynomials over a quaternionic division algebra. In view of these facts, in this chapter, we have extended the ideas developed in Chapter 2 to obtain a general framework for matrix polynomials over a quaternion division algebra. A systematic procedure for finding the left and right eigenvalues and their corresponding eigenvectors for matrix polynomials over a quaternion division algebra is derived.

Besides, bounds for the absolute values of the left and right eigenvalues are obtained for quaternionic matrix polynomials which generalize bounds for the absolute values of the eigenvalues of complex matrix polynomials for 1, 2, and ∞ -matrix norms. Specifically, bounds for the absolute values of the left and right eigenvalues of quaternionic matrix polynomials are presented by using localization theorems for the left and right eigenvalues of a quaternionic block matrix.

Some of the results (given in Subsection 5.3.2) directly generalize the bounds for the zeros of the right quaternionic polynomials. In this work, we show that some of our bounds for the absolute values of the eigenvalues of complex matrix polynomials are sharper than the bounds given in [15, Lemma 2.3 (2.1), Corollary 2.4 (2.5) (2.6)]. Moreover, we develop an algorithm to derive the above results for the powers of the block companion matrices which give better and sharper results than the bounds obtained via block companion matrices for the left and right eigenvalues of quaternionic matrix polynomials. Our bounds so obtained via the powers of block companion matrices are better than the bounds illustrated in [15, Lemma 2.3].

Throughout this chapter, we adopt the following notation and terminology: For $A, B \in M_n(\mathbb{H})$, Kronecker product of A and B is defined as

$$A \otimes B := (a_{ij}B).$$

Let $A \in M_n(\mathcal{K}), \mathcal{K} := \{\mathbb{R}, \mathbb{C}, \mathbb{H}\}$, be partitioned into $k \times k$ complex/quaternionic blocks

(5.1)
$$A := (A_{ij}) := \begin{bmatrix} A_{11} & A_{12} & \dots & A_{1k} \\ A_{21} & A_{22} & \dots & A_{2k} \\ \vdots & \vdots & \ddots & \vdots \\ A_{k1} & A_{k2} & \dots & A_{kk} \end{bmatrix}_{n \times n}$$

where $A_{i,j} \in M_{n_i \times n_j}(\mathcal{K})$, $(1 \le i, j \le k)$, is the (i, j) block of A such that $n_1 + \dots + n_k = n$.

,

Let us define

$$r_i^{(p)}(A) := \sum_{j=1, j \neq i}^k \|A_{ij}\|_p, \quad c_i^{(p)}(A) := \sum_{j=1, j \neq i}^k \|A_{ji}\|_p,$$

$$R_i^{(p)}(A) := r_i^{(p)}(A) + \|A_{ii}\|_p, \quad C_i^{(p)}(A) := c_i^{(p)}(A) + \|A_{ii}\|_p,$$

where p = 1, 2 and ∞ .

5.2. Inclusion regions for the eigenvalues of quaternionic block

matrices

We first derive inclusion regions for the left and right eigenvalues of a quaternionic block matrix.

Theorem 5.1. Let $A := (A_{ij}) \in M_n(\mathbb{H})$ be a block matrix as in (5.1). Then all the left and right eigenvalues of A are contained in ball $\mathcal{E} := \bigcap_{p=1,2,\infty} \mathcal{E}^{(p)}$, where

(5.2)
$$\mathcal{E}^{(p)} := \bigcup_{i=1}^{k} \mathcal{E}_{i}^{(p)} \text{ with } \mathcal{E}_{i}^{(p)} := \{ z \in \mathbb{H} : |z| \le R_{i}^{(p)}(A) \},$$

similarly, all the left and right eigenvalues of A are contained in ball $\mathcal{F} := \bigcap_{p=1,2,\infty} \mathcal{F}^{(p)}$, where

(5.3)
$$\mathcal{F}^{(p)} := \bigcup_{i=1}^{k} \mathcal{F}_{i}^{(p)} \text{ with } \mathcal{F}_{i}^{(p)} := \{ z \in \mathbb{H} : |z| \le C_{i}^{(p)}(A) \},$$

 $p = 1, 2, \infty \text{ and } (1 \le i \le k).$

Proof. The proof of the first part is similar to that of the complex case , for p = 1, 2 and ∞ .

Second part: For p = 2, let λ be a right eigenvalue of the quaternionic block matrix A. Then from [47, Corollary 2.7], λ is also a right eigenvalue of A^H . Thus $A^H y = y\lambda$ with nonzero $y = [y_1^T, y_2^T, \dots, y_k^T]^T \in \mathbb{H}^n$, where $y_i \in \mathbb{H}^{n_i}$. Let y_t be an element of y such that $\|y_t\|_2 \ge \|y_i\|_2$ for all $i \ (1 \le i \le k)$. Then $A^H y = y\lambda$ implies

$$\sum_{j=1}^{k} A_{jt}^{H} y_j = y_t \lambda.$$

Taking 2-norm and applying $||y_t||_2 \ge ||y_i||_2$ for all $i (1 \le i \le k)$, $||A_{jt}^H||_2 = ||A_{jt}||_2$, we obtain

$$|\lambda| \le \sum_{j=1}^k \|A_{jt}\|_2.$$

Similar proofs can be obtained for p = 1 and ∞ .

Let λ be a left eigenvalue of a quaternionic matrix A. Then from Lemma 2.1, $\overline{\lambda}$ is a left eigenvalue of A^{H} . Now analogue proof can be obtained for the desired results.

Theorem 5.2. Let $A := (A_{ij}) \in M_n(\mathbb{H})$ be a block matrix as in (5.1) and let $\gamma \in [0, 1]$. Then all the left and right eigenvalues of A are contained in ball $\mathcal{T} := \bigcap_{p=1,2,\infty} \mathcal{T}^{(p)}$, where

$$\mathcal{T}^{(p)} := \bigcup_{i=1}^{k} \mathcal{T}_{i}^{(p)} \text{ with } \mathcal{T}_{i}^{(p)} := \{ z \in \mathbb{H} : |z| \le R_{i}^{(p)}(A)^{\gamma} C_{i}^{(p)}(A)^{1-\gamma} \}$$

 $p = 1, 2, \infty \text{ and } (1 \le i \le k).$

Proof. The proof is immediate from Theorem 5.1, so we skip the proof. \blacksquare

We now state an inclusion region for the left and right eigenvalues of a general quaternionic matrix which follows immediately from Theorem 5.2. Thus we have the following corollary.

Corollary 5.3. Let $A := (a_{ij}) \in M_n(\mathbb{H})$ and let $\gamma \in [0,1]$. Then all the left and right eigenvalues of A are contained in the union of n-balls $\mathcal{G}_i(A) := \{z \in \mathbb{H} : |z| \leq r'_i(A)^{\gamma} c'_i(A)^{1-\gamma}\}$ $(1 \leq i \leq n), i.e.,$

$$\Lambda_l(A), \Lambda_r(A) \subseteq \mathcal{G}(A) := \bigcup_{i=1}^n \mathcal{G}_i(A),$$

where $r'_i(A)$ and $c'_i(A)$ are defined in (1.16).

Now, we present a sufficient condition for the stability of the discrete-time quaternionic system

$$w(t+1) = Aw(t),$$

where $w : \mathbb{R} \to \mathbb{H}, t \in \mathbb{R}$ and $A \in M_n(\mathbb{H})$.

Proposition 5.4. Let $A = (a_{ij}) \in M_n(\mathbb{H})$. Assume that

$$\omega_i(A)^{\gamma} \tau_i(A)^{1-\gamma} < 1 \quad \forall i \ (1 \le i \le n).$$

Then A is stable.

Proof. The proof is immediate from Corollary 5.3 and Definition 1.25.

It is known that all the right eigenvalues of a quaternionic Hermitian matrix A are real. Thus $\lambda_s(A) = \Lambda_r(A)$. By applying this argument, we have the following result.

Theorem 5.5. Let $A = (a_{ij}) \in M_n(\mathbb{H})$ be a Hermitian matrix. Then

$$\operatorname{trace}(A) = \sum_{i=1}^{n} \lambda_i(A),$$

where $\lambda_i \in \Lambda_r(A)$, $(1 \le i \le n)$.

Proof. Let A be a quaternionic Hermitian matrix. Then from Theorem 1.29, we have

$$A = VDV^H,$$

where $V \in M_n(\mathbb{H})$ is an unitary matrix and D is a diagonal matrix with the standard right eigenvalues λ_i $(1 \le i \le n)$ of A. Applying mapping Ψ and taking trace, we obtain

trace
$$(\Psi_A)$$
 = trace (Ψ_{UDU^H}) = trace $(\Psi_U \Psi_D \Psi_{U^H})$ = trace (Ψ_D) .

If all the diagonal entries of $A \in M_n(\mathbb{H})$ are real, then $\operatorname{trace}(\Psi_A) = 2\operatorname{trace}(A)$.

Lemma 5.6. Let $A = (a_{ij}) \in M_n(\mathbb{H})$. Then $||A||_2 \le ||A||_F$.

Proof. By the definition of matrix 2-norm and from Theorem 5.5, we have

$$||A||_{2}^{2} = \rho_{r}(AA^{H}) \leq \sum_{i=1}^{n} \lambda_{i}(AA^{H}) = \operatorname{trace}(AA^{H}) = ||A||_{F}^{2}. \blacksquare$$

Lemma 5.7. Let $A = (a_{ij}) \in M_n(\mathbb{H})$. Then

$$\rho_l(A), \, \rho_r(A) \le \|A\|_F.$$

Proof. The proof is immediate from [62, Theorem 3] and Lemma 5.6, so we skip the proof. \blacksquare

Lemma 5.8. Let $A = (a_{ij}) \in M_n(\mathbb{H})$. Then

$$||A||_{2}^{2} = ||A^{H}||_{2}^{2} = ||A^{H}A||_{2} = ||AA^{H}||_{2}.$$

Proof. By the definition of the operator norm of A

$$||A||_{2}^{2} = \rho_{r} \left(A^{H}A\right) = \rho\left(\Psi_{(A^{H}A)}\right) = \sqrt{\rho\left(\Psi_{(A^{H}A)}\Psi_{(A^{H}A)}\right)} = \sqrt{\rho\left(\Psi_{(A^{H}A)(A^{H}A)}\right)} ||A||_{2}^{2} = \sqrt{\rho_{r}\left((A^{H}A)(A^{H}A)\right)} = ||A^{H}A||_{2}.$$
85

Analogue proof can be derived for the remaining parts. \blacksquare

Theorem 5.9. Let $A = (A_{ij}) \in M_n(\mathbb{H})$ be a block matrix as in (5.1). Then

$$\rho_l(A), \, \rho_r(A) \le \left(\|A\|_F^2 - \left[\max_{1 \le i \le k} |\xi_i(A) - \xi_i(A^T)| \right]^2 \right)^{1/2},$$
$$= \sum_{k=1}^k \|A_{ij}\|_F.$$

where $\xi_i(A) := \sum_{j=1, j \neq i} ||A_{ij}||_F$

Proof. We prove the equivalent statement

$$\rho_l(A), \, \rho_r(A) \le \left[\|A\|_F^2 - \left(\left| \xi_i(A) - \xi_i(A^T) \right| \right)^2 \right]^{1/2} \, (1 \le i \le k).$$

Let $\xi_i(A) \neq 0$, $\xi_i(A^T) \neq 0$ and let $W_w = \text{diag}(I_1, \ldots, I_{i-1}, W_i, I_{i+1}, \ldots, I_k)$ with

 $W_i = \operatorname{diag}(w, w, \ldots, w),$

where $0 \neq w \in \mathbb{R}^+$, in the *i*-th position. By [63, Lemma 2.3] and Proposition 2.22, we have

$$\rho_r(W_w A W_w^{-1}) = \rho_r(A) \text{ and } \rho_l(W_w A W_w^{-1}) = \rho_l(A).$$

Then by Lemma 5.7, we obtain the following.

(5.4)
$$[\rho_l(A)]^2, [\rho_r(A)]^2 \le \left\| W_w A W_w^{-1} \right\|_F^2 = \left\| A \right\|_F^2 - \xi_i(A)^2 - \xi_i(A^T)^2 + \Phi_1,$$

where $\Phi_1 = w^2 \xi_i(A)^2 + w^{-2} \xi_i(A^T)^2$. Suppose $f(w) := ||A||_F^2 - \xi_i(A)^2 - \xi_i(A^T)^2 + w^2 \xi_i(A)^2 + w^{-2} \xi_i(A^T)^2$, then its first and second derivatives are given by $f'(w) = 2w \xi_i(A)^2 - 2w^{-3} \xi_i(A^T)^2$ and $f''(w) = 2\xi_i(A)^2 + 6w^{-4} \xi_i(A^T)^2$, respectively. Then f'(w) = 0 yields

$$w^2 = \pm \left[\frac{\xi_i(A^T)}{\xi_i(A)}\right],$$

and f''(w) = +ve. Hence f(w) has minimum value at $w^2 = \left[\frac{\xi_i(A^T)}{\xi_i(A)}\right]$. Therefore

$$\rho_l(A), \, \rho_r(A) \le \left[\|A\|_F^2 - \left(|\xi_i(A) - \xi_i(A^T)| \right)^2 \right]^{1/2}.$$

Corollary 5.10. Let $A := (a_{ij}) \in M_n(\mathbb{H})$. Then

$$\rho_l(A), \, \rho_r(A) \le \left(\|A\|_F^2 - \left[\max_{1 \le i \le n} |M_i(A) - M_i(A^T)| \right]^2 \right)^{1/2}$$
where $M_i(A) := \left(\sum_{j=1, \ j \ne i}^n |a_{ij}|^2 \right)^{1/2}$.
5.3. Quaternionic matrix polynomials

Let $\mathcal{P}_m(M_n(\mathbb{C}))$ be the space of complex matrix polynomials. $\mathcal{P} \in \mathcal{P}_m(M_n(\mathbb{C}))$ is defined by

(5.5)
$$\mathcal{P}(\lambda) := \sum_{i=0}^{m} \lambda^{i} A_{i},$$

where $A_i \in M_n(\mathbb{C})$ $(0 \le i \le m)$ and $\lambda \in \mathbb{C}$. Then the eigenvalue problem $\mathcal{P}(\lambda)x = 0$ is referred as a complex polynomial eigenvalue problem. The polynomial $\mathcal{P} \in \mathcal{P}_m(M_n(\mathbb{C}))$ is said to be regular if $\det(\mathcal{P}(\lambda)) \ne 0$ for some $\lambda \in \mathbb{C}$. The spectrum of a regular polynomial \mathcal{P} is denoted by $\Lambda(\mathcal{P})$, and is defined by

$$\Lambda(\mathcal{P}) := \{\lambda \in \mathbb{C} : \det(\mathcal{P}(\lambda)) = 0\}.$$

The above space of complex matrix polynomials can be extended to the space of matrix polynomials over a quaternion division algebra. Quaternionic matrix polynomials are derived in [44, 46–48], but for the case of right eigenvalues, where the matrix coefficients commutes with the variable identity of quaternionic matrix polynomial. Now in this section, we define a general framework for matrix polynomials over a quaternion division algebra to discuss the left and right eigenvalues of quaternionic matrix polynomials.

Let $\mathbb{L}'_m(M_n(\mathbb{H}))$ be the space of matrix polynomials over a quaternion division algebra. $\mathbf{L}' \in \mathbb{L}'_m(M_n(\mathbb{H}))$ is defined as

(5.6)
$$\mathbf{L}'(\xi) := \sum_{i=0}^{m} \xi^i A_i,$$

where $A_i \in M_n(\mathbb{H})$ $(0 \le i \le m)$ and $\xi \in \mathbb{H}$. Throughout this chapter we consider the following three cases:

- **Case 1**: when $\xi \in \mathbb{R}$ and $A_i \in M_n(\mathbb{H}) \ (0 \le i \le m)$,
- **Case 2**: when $\xi \in \mathbb{H}$ and $A_i \in M_n(\mathbb{H}) \ (0 \le i \le m)$,

Case 3: when $\xi \in \mathbb{C}$ and $A_i \in M_n(\mathbb{C})$ $(0 \le i \le m)$.

Case 1. Let $\mathbb{L}_m(M_n(\mathbb{H}))$ be the space of matrix polynomials over a quaternion division algebra. $\mathbf{L} \in \mathbb{L}_m(M_n(\mathbb{H}))$ is defined as

(5.7)
$$\mathbf{L}(\lambda) := \sum_{i=0}^{m} A_i \lambda^i,$$

where $A_i \in M_n(\mathbb{H})$ $(0 \le i \le m)$ and λ commutes with the quaternionic coefficients of the matrix polynomial. This polynomial over a quaternion division algebra can be found in [46–48].

Now we turn to define the right eigenvalue of $\mathbf{L} \in \mathbb{L}_m(M_n(\mathbb{H}))$ of the form (5.7) which is as follows.

Definition 5.11. Let $\mathbf{L} \in \mathbb{L}_m(M_n(\mathbb{H}))$ be as in (5.7) and let $\mu \in \mathbb{H}$. Then μ is called a right eigenvalue of \mathbf{L} if

$$A_0 x + A_1 x \mu + A_2 x \mu^2 + \dots + A_m x \mu^m = 0$$

for some nonzero $x \in \mathbb{H}^n$. Here x is called the right eigenvector corresponding to right eigenvalue μ . The set of right eigenvalues of **L** is called right spectrum of **L**, denoted by $\Lambda_r(\mathbf{L})$.

Case 2. We define the left eigenvalue of $\mathbf{L}' \in \mathbb{L}'_m(M_n(\mathbb{H}))$ which is as follows.

Definition 5.12. Let $\mathbf{L}' \in \mathbb{L}'_m(M_n(\mathbb{H}))$ be of the form (5.6) and let $\mu \in \mathbb{H}$. Then μ is called a left eigenvalue of \mathbf{L}' if

$$A_0 x + \mu A_1 x + \mu^2 A_2 x + \dots + \mu^m A_m x = 0$$

for some nonzero $x \in \mathbb{H}^n$. Here x is called the left eigenvector corresponding to left eigenvalue μ . The set of left eigenvalues of \mathbf{L}' is called left spectrum of \mathbf{L}' , denoted by $\Lambda_l(\mathbf{L}')$.

5.3.1. Right eigenvalues of quaternionic matrix polynomials and their bounds

In this subsection, first we give the linearization form of the quaternionic matrix polynomial \mathbf{L} (defined in (5.7)) by using the standard linearization technique given in [10] for the right and left eigenvalues of \mathbf{L} .

• For the right eigenvalues: The polynomial $\mathbf{L} \in \mathbb{L}_m(M_n(\mathbb{H}))$ be as in (5.7) can be written in the form:

$$C_{\mathbf{L}} + \lambda X,$$

where $C_{\mathbf{L}}, X \in M_{mn}(\mathbb{H})$ are of the forms

$$C_{\mathbf{L}} := \begin{bmatrix} 0 & I_n & 0 & \dots & 0 \\ 0 & 0 & I_n & & 0 \\ \vdots & \vdots & & \ddots & \vdots \\ 0 & 0 & 0 & \dots & I_n \\ -A_0 & -A_1 & -A_2 & \dots & -A_{m-1} \end{bmatrix}, \quad X := \begin{bmatrix} I_n & 0 & 0 & \dots & 0 \\ 0 & I_n & 0 & \dots & 0 \\ 0 & 0 & I_n & & 0 \\ \vdots & \vdots & & \ddots & \vdots \\ 0 & 0 & 0 & \dots & A_m \end{bmatrix}$$

and λ commutes with the quaternionic coefficients of the matrix polynomial. When $A_m = I_n$, the identity matrix, then the matrix polynomial (5.7) is said to be monic matrix polynomial and linearization form of it is given by

$$C_{\mathbf{L}} + \lambda E$$
, where $E := I_{nm}$.

Let $\mathbb{P}_m(M_{2n}(\mathbb{C}))$ be the space of complex matrix polynomials. $P \in \mathbb{P}_m(M_{2n}(\mathbb{C}))$ is given by

(5.8)
$$P(\mu) := \sum_{i=0}^{m} \Psi_{A_i} \mu^i,$$

where $A_i \in M_n(\mathbb{H}), \Psi_{A_i} \in M_{2n}(\mathbb{C}) \ (0 \le i \le m)$ and $\mu \in \mathbb{C}$.

Theorem 5.13. Let $\mathbf{L} \in \mathbb{L}_m(M_n(\mathbb{H}))$ be as in (5.7) and let $(C_{\mathbf{L}}, X)$ be a matrix pencil obtained from the linearization of \mathbf{L} . Then they have the same right eigenvalues.

Proof. Let $\mu \in \mathbb{H}$ be a right eigenvalue of the quaternionic matrix pencil $(C_{\mathbf{L}}, X)$, then there exists $x := [x_1^T, x_2^T, \dots, x_m^T]^T \in \mathbb{H}^n$ with $x_i \in \mathbb{H}^n$, $i = 1, 2, \dots, m$ such that

$$(5.9) \begin{bmatrix} 0 & I_n & 0 & \dots & 0 \\ 0 & 0 & I_n & & 0 \\ \vdots & \vdots & & \ddots & \vdots \\ 0 & 0 & 0 & \dots & I_n \\ -A_0 & -A_1 & -A_2 & \dots & -A_{m-1} \end{bmatrix} \begin{bmatrix} x_1 \\ x_2 \\ \vdots \\ x_{m-1} \\ x_m \end{bmatrix} = \begin{bmatrix} I_n & 0 & 0 & \dots & 0 \\ 0 & I_n & 0 & \dots & 0 \\ 0 & 0 & I_n & & 0 \\ \vdots & \vdots & & \ddots & \vdots \\ 0 & 0 & 0 & \dots & A_m \end{bmatrix} \begin{bmatrix} x_1 \\ x_2 \\ \vdots \\ x_{m-1} \\ x_m \end{bmatrix} \mu,$$

then (5.9) implies

$$x_2 = x_1 \mu,
 x_3 = x_2 \mu,
 \vdots
 x_m = x_{m-1} \mu,
 -A_0 x_1 - A_1 x_2 - \dots - A_{m-1} x_m = A_m x_m \mu.$$

Consecutive substitutions of the $(m-1)^{st}$ system of equations give

(5.10)
$$x_j := x_1 \mu^{j-1}, \quad j = 2, 3, \dots, m.$$

Substituting (5.10) in the last system of equation and solving, we obtain

$$A_0x_1 + A_1x_1\mu + \dots + A_{m-2}x_1\mu^{m-2} + A_{m-1}x_1\mu^{m-1} + A_mx_1\mu^m = 0.$$

Hence μ is a right eigenvalue of the quaternionic matrix polynomial **L**.

Regular matrix pencil over the quaternion skew filed has been defined in [33] and now, we extend it for matrix polynomial over the quaternion skew field.

Definition 5.14. Let $\mathbf{L} \in \mathbb{L}_m(M_n(\mathbb{H}))$ be as in (5.7). Then the matrix polynomial \mathbf{L} is called regular if there exists $\alpha \in \mathbb{R}$ such that $A_0 + \alpha A_1 + \cdots + \alpha^m A_m$ is an invertible matrix.

Theorem 5.15. Let $\mathbf{L} \in \mathbb{L}_m(M_n(\mathbb{H}))$ be of the form (5.7). Then the matrix polynomial \mathbf{L} is a regular polynomial if and only if $P \in \mathbb{P}_m(M_{2n}(\mathbb{C}))$ is a regular complex matrix polynomial.

Proof. Consider $\mathbf{L} \in \mathbb{L}_m(M_n(\mathbb{H}))$ is a regular matrix polynomial, then there exists $\alpha \in \mathbb{R}$ such that $\mathbf{L}(\alpha) = A_0 + \alpha A_1 + \cdots + \alpha^m A_m$ is an invertible matrix polynomial. From Theorem 1.12, the corresponding complex matrix polynomial $\Psi_{\mathbf{L}(\alpha)} = \Psi_{A_0} + \alpha \Psi_{A_1} + \cdots + \alpha^m \Psi_{A_m}$ is an invertible complex matrix polynomial. Thus, $P(\mu)$ is a regular complex matrix polynomial.

Conversely, let $P(\mu)$ be a regular complex matrix polynomial, then

det $[P(\eta)] \neq 0$, for some $\eta \in \mathbb{C}$, i.e., det $[\Psi_{A_0} + \eta \Psi_{A_1} + \dots + \eta^m \Psi_{A_m}] \neq 0$. Thus, there exists a real number λ_0 (say) such that

$$\det[\Psi_{A_0} + \lambda_0 \Psi_{A_1} + \dots + \lambda_0^m \Psi_{A_m}] \neq 0,$$

i.e., $\Psi_{A_0} + \lambda_0 \Psi_{A_1} + \cdots + \lambda_0^m \Psi_{A_m} = \Psi_{(A_0 + \lambda_0 A_1 + \cdots + \lambda_0^m A_m)}$ is an invertible complex matrix polynomial. Then by Theorem 1.12, the matrix polynomial $A_0 + \lambda_0 A_1 + \cdots + \lambda_0^m A_m$ is an invertible quaternionic matrix polynomial. Hence $\mathbf{L} \in \mathbb{L}_m(M_n(\mathbb{H}))$ is a regular quaternionic matrix polynomial.

Proposition 5.16. Let $\mathbf{L} \in \mathbb{L}_m(M_n(\mathbb{H}))$ be as in (5.7) and let $\mu \in \mathbb{H}$ be a right eigenvalue of \mathbf{L} . Assume that $0 \neq \rho \in \mathbb{H}$, then $\rho^{-1}\mu\rho$ is also a right eigenvalue of \mathbf{L} .

Proof. Suppose x is a quaternionic eigenvector of the matrix polynomial \mathbf{L} corresponding to the right eigenvalue μ of the matrix polynomial \mathbf{L} , then

(5.11)
$$A_0 x + A_1 x \mu + A_2 x \mu^2 + \dots + A_m x \mu^m = 0.$$

If $\rho \in \mathbb{H} \setminus \{0\}$, then from (5.11), we have

$$A_0 x \rho + A_1 x \rho(\rho^{-1} \mu \rho) + A_2 x \rho(\rho^{-1} \mu^2 \rho) + \dots + A_m x \rho(\rho^{-1} \mu^m \rho) = 0.$$

Thus $\rho^{-1}\mu\rho$ is also a right eigenvalue of the matrix polynomial **L** corresponding to an eigenvector $x\rho$.

Theorem 5.17. Let $\mathbf{L} \in \mathbb{L}_m(M_n(\mathbb{H}))$ be as in (5.7). Then

(5.12)
$$\Lambda_r(\mathbf{L}) \cap \mathbb{C} = \Lambda(P),$$

(5.13)
$$\Lambda_r(\mathbf{L}) = \{ \rho^{-1} \mu \rho \in \mathbb{H} : \mu \in \Lambda(P), \ 0 \neq \rho \in \mathbb{H} \}.$$

Proof. Consider $A_i \in M_n(\mathbb{H})$ for i = 0, 1, ..., m and $x \in \mathbb{H}^n$. We write A_i and x as $A_i = A_i' + A_i'' \mathbf{j}, 0 \le i \le m$ and $x = x_1 + x_2 \mathbf{j}$, where $A_i', A_i'' \in M_n(\mathbb{C})$ and $x_1, x_2 \in \mathbb{C}^n$. If $\mu \in \Lambda_r(\mathbf{L}) \cap \mathbb{C}$, then there exists $x \in \mathbb{H}^n$ such that $A_0 x + A_1 x \mu + A_2 x \mu^2 + \cdots + A_m x \mu^m = 0$ which is equivalent to the complex system. Hence from Theorem 1.12, we obtain

$$(5.14) \begin{bmatrix} A'_0 & A''_0 \\ -\overline{A''_0} & \overline{A'_0} \end{bmatrix} \begin{bmatrix} x_1 \\ -\overline{x_2} \end{bmatrix} + \mu \begin{bmatrix} A'_1 & A''_1 \\ -\overline{A''_1} & \overline{A_1'} \end{bmatrix} \begin{bmatrix} x_1 \\ -\overline{x_2} \end{bmatrix} + \dots + \mu^m \Phi_2 \begin{bmatrix} x_1 \\ -\overline{x_2} \end{bmatrix} = 0,$$
91

where
$$\Phi_2 = \begin{bmatrix} A'_m & A''_m \\ -\overline{A''_m} & \overline{A'_m} \end{bmatrix}$$
 and also
 $\begin{bmatrix} A'_0 & A''_0 \\ -\overline{A''_0} & \overline{A'_0} \end{bmatrix} \begin{bmatrix} x_2 \\ \overline{x_1} \end{bmatrix} + \overline{\mu} \begin{bmatrix} A'_1 & A''_1 \\ -\overline{A''_1} & \overline{A_1'} \end{bmatrix} \begin{bmatrix} x_2 \\ \overline{x_1} \end{bmatrix} + \dots + \overline{\mu}^m \begin{bmatrix} A'_m & A''_m \\ -\overline{A''_m} & \overline{A'_m} \end{bmatrix} \begin{bmatrix} x_2 \\ \overline{x_1} \end{bmatrix} = 0;$

thus $\mu, \ \overline{\mu} \in \Lambda(P)$.

Conversely, suppose $\mu \in \Lambda(P)$, then there exists $x := [x_1^T, x_2^T]^T \in \mathbb{C}^{2n}$ such that

$$\begin{bmatrix} A'_0 & A''_0 \\ -\overline{A''_0} & \overline{A'_0} \end{bmatrix} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} + \mu \begin{bmatrix} A'_1 & A''_1 \\ -\overline{A''_1} & \overline{A_1'} \end{bmatrix} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} + \dots + \mu^m \begin{bmatrix} A'_m & A''_m \\ -\overline{A''_m} & \overline{A'_m} \end{bmatrix} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} = 0,$$

which is equivalent to the following systems

(5.15)
$$\begin{cases} A'_0 x_1 + A''_0 x_2 + \mu A'_1 x_1 + \mu A''_1 x_2 + \dots + \mu^m A'_m x_1 + \mu^m A''_m x_2 = 0, \\ -\overline{A''_0} x_1 + \overline{A'_0} x_2 - \mu \overline{A''_1} x_1 + \mu \overline{A'_1} x_2 + \dots - \mu^m \overline{A''_m} x_1 + \mu^m \overline{A'_m} x_2 = 0. \end{cases}$$

Suppose that $x = x_1 - \overline{x_2}$ **j**, then (5.15) is equivalent to $A_0 x + A_1 x \mu + A_2 x \mu^2 + \cdots + A_m x \mu^m = 0$, hence $\mu \in \Lambda_r(\mathbf{L}) \cap \mathbb{C}$.

For (5.13), consider $\mu \in \Lambda(\Psi_{C_{\mathbf{L}}}, \Psi_X) = \Lambda(P)$, then for every $0 \neq \rho \in \mathbb{H}$, it is clear that $\rho^{-1}\mu\rho \in \Lambda_r(C_{\mathbf{L}}, X) = \Lambda_r(\mathbf{L})$. Conversely; consider $\lambda \in \Lambda_r(C_{\mathbf{L}}, X) = \Lambda_r(\mathbf{L})$, then by [5, Lemma 9], there exists $0 \neq \rho \in \mathbb{H}$ such that $\rho\mu\rho^{-1} = \mu_1 \in \Lambda_r(C_{\mathbf{L}}, X) \cap \mathbb{C}$. Applying (5.12), we have $\mu_1 \in \Lambda(P)$ and also $\mu = \rho^{-1}\mu_1\rho$.

We now state two corollaries to Theorem 5.17 which are also available in [33].

Corollary 5.18. Let $\mathbf{L} \in \mathbb{L}_1(M_n(\mathbb{H}))$ be of the form $\mathbf{L}(\lambda) := A + \lambda B$, where λ commutes with the quaternionic matrices and let $P \in \mathbb{P}_1(M_{2n}(\mathbb{C}))$ be of the form $P(\mu) := \Psi_A + \mu \Psi_B$, where $\mu \in \mathbb{C}$. Then

(5.16)
$$\begin{cases} \Lambda_r(\mathbf{L}) \cap \mathbb{C} = \Lambda(P), \\ \Lambda_r(\mathbf{L}) = \{\rho^{-1}\lambda\rho : \lambda \in \Lambda(P), \ 0 \neq \rho \in \mathbb{H}\}. \end{cases}$$

Proof. Assume that m = 1 in Theorem 5.17, then we have the desired result.

Also we give another corollary of Theorem 5.17 which can be found from [33].

Corollary 5.19. Let $A \in M_n(\mathbb{H})$. Then

$$\Lambda_r(A) \cap \mathbb{C} = \Lambda(\Psi_A),$$

$$\Lambda_r(A) = \{\rho^{-1}\lambda\rho : \lambda \in \Lambda(\Psi_A), \ 0 \neq \rho \in \mathbb{H}\}.$$

92

Proof. Assume that $B = I_n$ in Corollary 5.18, we have the desired result.

Remark 5.20. From (5.14), it is also clear that $\mu \in \mathbb{H}$ is a right eigenvalue of the quaternionic matrix polynomial **L** if and only if det $(P(\mu)) = 0$.

Let $\mathbb{P}_m(M_{2n}(\mathbb{C}))$ be the space of complex matrix polynomials of degree $(\leq m)$, and let $P \in \mathbb{P}_m(M_{2n}(\mathbb{C}))$ be of the form (5.8). If A_m is invertible, then P has 2mn finite eigenvalues, i.e., \mathbf{L} does not have any infinite (∞) right eigenvalue as there is one one correspondence between the space of matrix polynomials of degree m over a quaternion division algebra and the space of its conjugate matrix polynomial P of degree m over the complex field. Also, if A_m is not invertible, then P has at least one infinite (∞) eigenvalue. In general, if A_m is not invertible, then the quaternionic matrix polynomial \mathbf{L} has at least one infinite right eigenvalue, whereas if A_0 is not invertible, then 0 is a right eigenvalue of the quaternionic matrix polynomial \mathbf{L} . Therefore, for upper bounds of the absolute values of the right eigenvalues of \mathbf{L} require A_m to be invertible and the lower bounds of the absolute values of the right eigenvalues of \mathbf{L} require A_0 to be invertible. Hence, now onwards, we assume A_0 and A_m are invertible quaternionic matrices.

Let d be the degree of det $[P(\mu)]$, where $\mu \in \mathbb{H}$. Then the d zeros of det $[P(\mu)]$ are called the finite eigenvalues of P. If d < 2mn, then we say that $P(\mu)$ has 2mn - dinfinite eigenvalues. Let $\mathbf{L} \in \mathbb{L}_m(M_n(\mathbb{H}))$ be a regular polynomial and if A_m is not invertible, then the corresponding adjoint complex matrix Ψ_{A_m} is not invertible. Thus, the complex matrix polynomial $P(\mu)$ has at least one infinite (∞) eigenvalue and hence the quaternionic matrix polynomial \mathbf{L} has at least one infinite (∞) right eigenvalue.

Example 5.21. Consider the quaternionic matrix pencil $\mathbf{L}(\lambda) = A + \lambda B$, where λ commutes with the quaternionic matrices and $A, B \in M_n(\mathbb{H})$ of the forms

$$A = \begin{bmatrix} \mathbf{i} + \mathbf{j} & 2 + \mathbf{k} \\ 1 - \mathbf{j} & 3 - \mathbf{i} - \mathbf{j} \end{bmatrix}, B = \begin{bmatrix} 0 & \mathbf{j} + \mathbf{k} \\ 0 & 0 \end{bmatrix}.$$

Since B is not invertible, hence Ψ_B is also not invertible. The complex matrix pencil $\Psi_A + \mu \Psi_B, \ \mu \in \mathbb{C}$ has at least one infinite (∞) eigenvalue. Hence **L** has an infinite (∞) right eigenvalue. In this case, the set of right eigenvalues of **L** is $\{[-1 + 2.5495\mathbf{i}], \infty\}$.

Now, consider A_m is invertible. Then from [5, 27] the block companion matrix C_V has exactly mn right eigenvalues which are complex numbers with non-negative imaginary

parts. Also, we observe that if A_m is not invertible and \mathbf{L} is regular, then $P \in \mathbb{P}_m(M_{2n}(\mathbb{C}))$ has the corresponding matrix polynomial has exactly mn right eigenvalues which belong to $\mathbb{C}_{\infty} := \mathbb{C} \cup \{\infty\}$ with non-negative imaginary parts. Those right eigenvalues are said to be the standard right eigenvalues of \mathbf{L} . Thus, we define the standard right eigenvalues of $\mathbf{L} \in \mathbb{L}_m(M_n(\mathbb{H}))$ as follows.

Definition 5.22. Let $\mathbf{L} \in \mathbb{L}_m(M_n)(\mathbb{H})$ be as in (5.7). Then we define the set of standard right eigenvalues of a regular matrix polynomial \mathbf{L} as

$$\Lambda_s(\mathbf{L}) := \{ \alpha \in \mathbb{C}_{\infty} : A_0 x + A_1 x \alpha + \dots + A_m x \alpha^m = 0, \ 0 \neq x \in \mathbb{H}^n, \ \Im(\alpha) \ge 0 \},\$$
$$\mathbb{C}_{\infty} = \mathbb{C} \cup \{\infty\}.$$

The above definition generalizes the definition of standard right eigenvalues of a single quaternionic matrix [5,27].

Now, we state a framework to find bounds for the right eigenvalues of quaternionic matrix polynomials.

• To find bounds of the right eigenvalues of the quaternionic matrix polynomial $\mathbf{L} \in \mathbb{L}_m(M_n(\mathbb{H}))$, we introduce two new quaternionic matrix polynomials associated with $\mathbf{L} \in \mathbb{L}_m(M_n(\mathbb{H}))$:

(5.17)
$$\mathbf{L}_{V}(\lambda) := \lambda^{m} I_{n} + \sum_{i=0}^{m-1} \lambda^{i} V_{i},$$

where $V_i := A_m^{-1} A_i$, $i = 0, 1, \dots, m-1$, so that $\mathbf{L}(\lambda) = A_m \mathbf{L}_V(\lambda)$, and

(5.18)
$$\mathbf{L}_{S}(\lambda) := \lambda^{m} I_{n} + \sum_{i=1}^{m} S_{i} \lambda^{m-i},$$

where $S_i := A_0^{-1}A_i$, i = 1, ..., m, now $A_0^{-1}\lambda^m \mathbf{L}(1/\lambda) = \mathbf{L}_S(\lambda)$. The polynomials \mathbf{L} and \mathbf{L}_V have the same right eigenvalues, whereas the right eigenvalues of the polynomial \mathbf{L}_S are the reciprocal of the right eigenvalues of the polynomial \mathbf{L} . The block companion matrices C_V and C_S corresponding to the monic matrix polynomials \mathbf{L}_V and \mathbf{L}_S are given by

$$C_V := \begin{array}{c|c} n & n(m-1) \\ \hline 0 & I \\ \hline -V_0 & -\Delta_V \end{array} \right] \text{ and } C_S := \begin{array}{c|c} n & n(m-1) \\ \hline 0 & I \\ \hline -S_m & -\Delta_S \end{array} \right],$$
94

respectively, where $\Delta_V, \Delta_S \in M_{n \times n(m-1)}(\mathbb{H})$ are of the forms

$$\Delta_V = \begin{bmatrix} V_1 & V_2 & \dots & V_{m-1} \end{bmatrix}, \ \Delta_S = \begin{bmatrix} S_{m-1} & S_{m-2} & \dots & S_1 \end{bmatrix}.$$

Moreover, the right eigenvalues of the monic matrix polynomials $\mathbf{L}_V(\lambda), \mathbf{L}_S(\lambda)$ and the right eigenvalues of the block companion matrices C_V, C_S are same, respectively.

Bounds for the right eigenvalues of quaternionic matrix polynomials: Now onwards, we define

$$V := \begin{bmatrix} V_0 & V_1 & \dots & V_{m-1} \end{bmatrix} \in M_{n \times mn}(\mathbb{H}), \quad S := \begin{bmatrix} S_m & S_{m-1} & \dots & S_1 \end{bmatrix} \in M_{n \times mn}(\mathbb{H}),$$

where V_i , $S_j \in M_n(\mathbb{H})$, $(0 \le i \le m-1)$, $(1 \le j \le m)$.

Theorem 5.23. Let $\mathbf{L} \in \mathbb{L}_m(M_n(\mathbb{H}))$ be as in (5.7). Then for any right eigenvalue μ of the polynomial L satisfies the following inequalities:

(5.19)
$$\frac{1}{\beta_1} \le |\mu| \le \alpha_1,$$

(5.20)
$$\frac{1}{\beta_2} \le |\mu| \le \alpha_2,$$

(5.21)
$$\frac{1}{\beta_3} \le |\mu| \le \alpha_3,$$

where $\gamma \in [0, 1]$,

$$\begin{aligned} \alpha_1 &:= \min_{p=1,2,\infty} \left\{ \max\left(\|V_0\|_p, 1 + \max_{1 \le j \le m-1} (\|V_j\|_p) \right) \right\}, \\ \beta_1 &:= \min_{p=1,2,\infty} \left\{ \max\left(\|S_m\|_p, 1 + \max_{1 \le j \le m-1} (\|S_j\|_p) \right) \right\}, \\ \alpha_2 &:= \min_{p=1,2,\infty} \left\{ \max\left(1, \sum_{j=0}^{m-1} \|V_j\|_p \right) \right\}, \\ \beta_2 &:= \min_{p=1,2,\infty} \left(\max\left\{ 1, \sum_{j=1}^m \|S_j\|_p \right\} \right), \end{aligned}$$

$$\alpha_{3} := \min_{p=1,2,\infty} \left\{ \max_{1 \le i \le m} \left(R_{i}^{(p)}(C_{V})^{\gamma} C_{i}^{(p)}(C_{V})^{1-\gamma} \right) \right\}, and$$

$$\beta_{3} := \min_{p=1,2,\infty} \left\{ \max_{1 \le i \le m} \left(R_{i}^{(p)}(C_{S})^{\gamma} C_{i}^{(p)}(C_{S})^{1-\gamma} \right) \right\}.$$

Proof. The proofs are immediate from Theorems 5.1 and 5.2, so we skip the proofs. \blacksquare

It is known that similar quaternionic matrices have the same right eigenvalues. Therefore, we can apply a similarity transformation to obtain different and potentially tighter bounds. So, we now define the following matrices.

$$C'_V := WC_V W^{-1}, \ C'_S := WC_S W^{-1}, \ W := w_1 I_n \oplus (W' \otimes I_n) = w_1 I_n \oplus \cdots \oplus w_m I_n,$$

 $W' := \operatorname{diag}(w_2 I_n, w_3 I_n, \ldots, w_m I_n)$, where w_i are positive integers. The matrix C'_V can also be written as

with
$$W := \begin{pmatrix} n(m-1) & n \\ C_1 & 0 \\ \hline C_2 & C_3 \end{pmatrix}$$
 and $W^{-1} := \begin{pmatrix} n(m-1) & n \\ C_1' & 0 \\ \hline C_2' & C_3' \end{pmatrix}$,

where

$$C_1 := \begin{bmatrix} w_1 I_n & 0 & \dots & 0 \end{bmatrix}, \ C'_1 := \begin{bmatrix} \frac{1}{w_1} I_n & 0 & \dots & 0 \end{bmatrix},$$

$$C_{2} := \begin{bmatrix} 0 & w_{2}I_{n} & 0 & \dots & 0 \\ \vdots & 0 & w_{3}I_{n} & & 0 \\ \vdots & \vdots & & \ddots & 0 \\ 0 & 0 & 0 & \dots & w_{m-1}I_{n} \\ 0 & 0 & 0 & \dots & 0 \end{bmatrix}, C_{2}' := \begin{bmatrix} 0 & \frac{1}{w_{2}}I_{n} & 0 & \dots & 0 \\ \vdots & 0 & \frac{1}{w_{3}}I_{n} & & 0 \\ \vdots & \vdots & & \ddots & 0 \\ 0 & 0 & 0 & \dots & \frac{1}{w_{m-1}}I_{n} \\ 0 & 0 & 0 & \dots & 0 \end{bmatrix},$$

 $C_3 := \begin{bmatrix} 0\\0\\\vdots\\w_m I_n \end{bmatrix}, \text{ and } C'_3 := \begin{bmatrix} 0\\0\\\vdots\\\frac{1}{w_m} I_n \end{bmatrix}.$

By using above arguments and, Theorems 5.1 and 5.2, we have the following bounds.

Theorem 5.24. Let $w_j \in \mathbb{R}^+$, $(1 \le j \le m)$, with $w_m = 1$. Then for any right eigenvalue μ of $\mathbf{L} \in \mathbb{L}_m(M_n(\mathbb{H}))$ satisfies the following inequalities:

(5.23)
$$\frac{1}{\beta_1} \le |\mu| \le \alpha_1,$$

(5.24)
$$\frac{1}{\beta_2} \le |\mu| \le \alpha_2,$$

(5.25)
$$\frac{1}{\beta_3} \le |\mu| \le \alpha_3,$$

where $\gamma \in [0,1]$,

$$\begin{aligned} \alpha_{1} &:= \min_{p=1,2,\infty} \left\{ \max\left(\frac{\|V_{0}\|_{p}}{w_{1}}, \max_{1 \leq j \leq m-1} \left(\frac{w_{j}}{w_{j+1}} + \frac{\|V_{j}\|_{p}}{w_{j+1}}\right)\right) \right\}, \\ \beta_{1} &:= \min_{p=1,2,\infty} \left\{ \max\left(\frac{\|S_{m}\|_{p}}{w_{1}}, \max_{1 \leq j \leq m-1} \left(\frac{w_{j}}{w_{j+1}} + \frac{\|S_{m-j}\|_{p}}{w_{j+1}}\right)\right) \right\}, \\ \alpha_{2} &:= \min_{p=1,2,\infty} \left\{ \max\left(\max\left(\max_{1 \leq j \leq m-1} \left(\frac{w_{j}}{w_{j+1}}\right), \sum_{i=0}^{m-1} \frac{\|V_{i}\|_{p}}{w_{i+1}}\right) \right\}, \\ \beta_{2} &:= \min_{p=1,2,\infty} \left\{ \max\left(\max\left(\max_{1 \leq j \leq m-1} \left(\frac{w_{j}}{w_{j+1}}\right), \sum_{i=0}^{m-1} \frac{\|S_{m-i}\|_{p}}{w_{i+1}}\right) \right\}, \\ \alpha_{3} &:= \min_{p=1,2,\infty} \left\{ \max\left(R_{i}^{(p)}(C_{V}')^{\gamma} C_{i}^{(p)}(C_{V}')^{1-\gamma}\right) \right\}, and \\ \beta_{3} &:= \min_{p=1,2,\infty} \left\{ \max\left(R_{i}^{(p)}(C_{S}')^{\gamma} C_{i}^{(p)}(C_{S}')^{1-\gamma}\right) \right\}. \end{aligned}$$

Proof. The block companion matrix of the monic matrix polynomial $\mathbf{L}_{V}(\lambda)$ is given by

$$C_V := \begin{array}{c|c} n & n(m-1) \\ \hline 0 & I \\ \hline -V_0 & -\Delta_V \end{array} \right].$$

Now (5.22) yields

$$C'_V := W C_V W^{-1}$$

$$C_{V}' = \begin{array}{c|c} n(m-1) & n \\ \hline C_{1}'' & 0 \\ \hline C_{2}'' & C_{3}'' \\ \end{array} \right], \ where \ C_{1}'' = \begin{bmatrix} 0 & \frac{w_{1}}{w_{2}}I_{n} & 0 & \dots & 0 \end{bmatrix},$$
97

$$C_{2}'' = \begin{bmatrix} 0 & 0 & \frac{w_{2}}{w_{3}}I_{n} & 0 & \dots & 0 \\ \vdots & \vdots & 0 & \frac{w_{3}}{w_{4}}I_{n} & & 0 \\ \vdots & \vdots & \vdots & \ddots & 0 \\ 0 & 0 & 0 & 0 & \dots & \frac{w_{m-2}}{w_{m-1}}I_{n} \\ -\frac{w_{m}}{w_{1}}V_{0} & -\frac{w_{m}}{w_{2}}V_{1} & -\frac{w_{m}}{w_{3}}V_{2} & -\frac{w_{m}}{w_{4}}V_{3} & \dots & -\frac{w_{m}}{w_{m-1}}V_{m-2} \end{bmatrix}, \text{ and } C_{3}'' := \begin{bmatrix} 0 \\ 0 \\ \vdots \\ \frac{w_{m-1}}{w_{m}}I_{n} \\ -V_{m-1} \end{bmatrix}.$$

Upper bounds of the right eigenvalues of the matrix polynomial \mathbf{L} are obtained by applying Theorems 5.1 and 5.2 to the matrix C'_V .

Also, lower bounds of the right eigenvalues of the matrix polynomial \mathbf{L} are obtained by applying Theorems 5.1 and 5.2 to the matrix C'_S .

Substituting $w_j = ||V_j||_p$, $(p = 1, 2, \infty)$, in the part (1) of Theorem 5.24, we obtain

(5.26)
$$|\mu| \le \min_{p=1,2,\infty} \left\{ \max\left(\frac{\|V_0\|_p}{\|V_1\|_p}, \ 2\max_{1\le j\le m-1} \frac{\|V_j\|_p}{\|V_{j+1}\|_p}\right) \right\}$$

Now we discuss the different bounds for the right eigenvalues of quaternionic matrix polynomials.

Theorem 5.25. Let $\mathbf{L} \in \mathbb{L}_m(M_n(\mathbb{H}))$ be as in (5.7). Then for any right eigenvalue μ of \mathbf{L} satisfies the following inequality:

$$\frac{1}{\eta^{C_S}} \le |\mu| \le \eta^{C_V},$$

where

$$\eta^{C_{\alpha}} := \left[\|C_{\alpha}\|_{F}^{2} - \left\{ \max_{1 \le i \le m} \left| \xi_{i} \left(C_{\alpha} \right) - \xi_{i} \left(C_{\alpha}^{T} \right) \right| \right\}^{2} \right]^{1/2}, \alpha \in \{V, S\}.$$

Proof. The proof is immediate from Theorem 5.9, so we omit the proof. \blacksquare

Theorem 5.26. Let $\mathbf{L} \in \mathbb{L}_m(M_n(\mathbb{H}))$ be as in (5.7). Then for any right eigenvalue μ of \mathbf{L} satisfies the following inequality:

$$\frac{1}{\eta^{C_S}} \le |\mu| \le \eta^{C_V}, \text{ where}$$
$$\eta^{C_\alpha} := \left[\|C_\alpha\|_F^2 - \left\{ \max_{1 \le i \le m} \left| M_i(C_\alpha) - M_i(C_\alpha^T) \right| \right\}^2 \right]^{1/2}, \alpha \in \{V, S\}.$$

Proof. The proof is immediate from Corollary 5.10, so we omit the proof. \blacksquare

Next, we give upper bounds for the left and right spectral radius of quaternionic matrices in term of quaternionic matrix norms.

Theorem 5.27. Let $A \in M_n(\mathbb{H})$. If $\|.\|_{\beta}$, $(\beta = 1, 2, \infty, F)$, are the quaternionic matrix norms, then

$$\rho_l(A), \, \rho_r(A) \le \|A\|_{\beta}.$$

Proof. The proof is similar to that of the complex case. \blacksquare

By applying Theorem 5.27, we extend bounds for the eigenvalues of a complex matrix polynomial [15] to bounds for the right eigenvalues of a quaternionic matrix polynomial which are as follows.

Theorem 5.28. Let $\mathbf{L} \in \mathbb{L}_m(M_n(\mathbb{H}))$ be as in (5.7). Then for any right eigenvalue μ of \mathbf{L} satisfies the following inequalities:

1.
$$\max\left(\|S_m\|_1, 1 + \max_{1 \le i \le m-1} (\|S_i\|_1)\right)^{-1} \le |\mu| \le \max\left(\|V_0\|_1, 1 + \max_{1 \le i \le m-1} (\|V_i\|_1)\right),$$

2.
$$\max\left(1, \|S\|_{\infty}\right)^{-1} \le |\mu| \le \max\left(1, \|V\|_{\infty}\right),$$

3.
$$\left(\|I_n + SS^H\|_2\right)^{-1/2} \le |\mu| \le \left(\|I_n + VV^H\|_2\right)^{1/2}.$$

Proof. The first two bounds are follows from Theorem 5.27 (for $\beta = 1, \infty$). Now for the third bound, applying Theorem 5.27 ($\beta = 2$), we have

$$(||C_S||_2)^{-1} \le |\mu| \le ||C_V||_2.$$

To find $||C_V||_2$, we write

$$M := \begin{bmatrix} 0 & I_n & 0 & \dots & 0 \\ 0 & 0 & I_n & & 0 \\ \vdots & \vdots & \vdots & \ddots & \vdots \\ 0 & 0 & 0 & \dots & I_n \\ 0 & 0 & 0 & \dots & 0 \end{bmatrix} \text{ and } N := \begin{bmatrix} 0 & 0 & 0 & \dots & 0 \\ 0 & 0 & 0 & & 0 \\ \vdots & \vdots & \vdots & \ddots & \vdots \\ 0 & 0 & 0 & \dots & 0 \\ -V_0 & -V_1 & -V_2 & \dots & -V_{m-1} \end{bmatrix}$$

Then $M^H N = N^H M = 0$. Thus by Lemma 5.8, we have

$$||C_V||_2^2 = ||C_V^H C_V||_2 = ||(M+N)^H (M+N)||_2 = ||M^H M + N^H N||_2$$

$$\leq ||I_{mn} + N^H N||_2 = ||I_{mn} + NN^H||_2 = ||I_n + VV^H||_2.$$

For finding $||C_S||_2$ analogue proof can be given, so we skip the proof.

Theorem 5.29. Let $w_j \in \mathbb{R}^+$, $(1 \le j \le m)$, with $w_m = 1$. Then for any right eigenvalue μ of $\mathbf{L} \in \mathbb{L}_m(M_n(\mathbb{H}))$ satisfies the following inequalities:

1.
$$\left[\max_{1 \le j \le m-1} \left\{ \frac{\|S_m\|_1}{w_1}, \frac{w_j}{w_{j+1}} + \frac{\|S_{m-j}\|_1}{w_{j+1}} \right\} \right]^{-1} \le |\mu| \le \max_{1 \le j \le m-1} \left\{ \frac{\|V_0\|_1}{w_1}, \frac{w_j}{w_{j+1}} + \frac{\|V_j\|_1}{w_{j+1}} \right\}.$$
2.
$$\frac{1}{\beta_1} \le |\mu| \le \alpha_1,$$

where

3.

$$\alpha_{1} := \max_{1 \le j \le m-1} \left(\frac{w_{j}}{w_{j+1}}, \left\| \left[\frac{V_{0}}{w_{1}}, \dots, \frac{V_{m-2}}{w_{m-1}}, V_{m-1} \right] \right\|_{\infty} \right),$$

$$\beta_{1} := \max_{1 \le j \le m-1} \left(\frac{w_{j}}{w_{j+1}}, \left\| \left[\frac{S_{m}}{w_{1}}, \dots, \frac{S_{2}}{w_{m-1}}, S_{1} \right] \right\|_{\infty} \right).$$

$$\left[\max_{1 \le j \le m-1} \left(\frac{w_{j}^{2}}{w_{j+1}^{2}} \right) + \sum_{j=0}^{m-1} \frac{\|S_{m-j}\|_{2}^{2}}{w_{j+1}^{2}} \right]^{-1/2} \le |\mu| \le \left[\max_{1 \le j \le m-1} \left(\frac{w_{j}^{2}}{w_{j+1}^{2}} \right) + \sum_{j=0}^{m-1} \frac{\|V_{j}\|_{2}^{2}}{w_{j+1}^{2}} \right]^{1/2}.$$

Proof. The proofs are immediate from the proof method of Theorem 5.28 and using Lemma 5.27 on the matrices C'_V and C'_S .

The following theorem is a direct application of Corollary 5.3 which gives bounds for the right eigenvalues of quaternionic matrix polynomials.

Theorem 5.30. Let $\mathbf{L} \in \mathbb{L}_m(M_n(\mathbb{H}))$ be as in (5.7) and let $\gamma \in [0, 1]$. Then for any right eigenvalue μ of \mathbf{L} satisfies the following inequality:

$$\left[\max_{1 \le i \le mn} \left\{ r'_i(C_S)^{\gamma} t'_i(C_S)^{1-\gamma} \right\} \right]^{-1} \le |\mu| \le \max_{1 \le i \le mn} \left\{ r'_i(C_V)^{\gamma} c'_i(C_V)^{1-\gamma} \right\}$$

By the definition of the right spectral radius, we have for any $A \in M_n(\mathbb{H})$

$$||A||_{2}^{2} = \rho_{r}(A^{H}A) \leq ||A^{H}A||_{1} \leq ||A^{H}||_{1} ||A||_{1}$$
$$||A||_{2}^{2} \leq ||A||_{\infty} ||A||_{1}.$$

Now, we have two cases as follows.

Case 1: If $||A||_1 \le ||A||_{\infty}$, then $||A||_2 \le ||A||_{\infty}$.

Case 2: If $||A||_{\infty} \le ||A||_1$, then $||A||_2 \le ||A||_1$.

Thus if $A \in \mathbb{S} := \{$ symmetric, skew symmetric, Hermitian, skew-Hermitian, η -Hermitian, η -anti-Hermitian $\}$, then we have

$$||A||_2 \le ||A||_{\infty} = ||A||_1 \,.$$
100

It is clear that if $V_j (0 \le j \le m-1) \in \mathbb{S}$, then $\|V_j\|_2 \le \|V_j\|_q (\beta = 1, \infty)$. Thus from Theorem 5.23, we obtain

$$\min_{p=1,2,\infty} \left\{ \max\left(\|V_0\|_p, 1 + \max_{1 \le j \le m-1} (\|V_j\|_p) \right) \right\} = \max\left(\|V_0\|_2, 1 + \max_{1 \le j \le m-1} (\|V_j\|_2) \right),$$

$$\min_{p=1,2,\infty} \left\{ \max\left(1, \sum_{j=0}^{m-1} \|V_j\|_p \right) \right\} = \max\left(1, \sum_{j=0}^{m-1} \|V_j\|_2 \right),$$

$$\min_{p=1,2,\infty} \left\{ \max\left(R_i^{(p)}(C_V)^{\gamma} C_i^{(p)}(C_V)^{1-\gamma} \right) \right\} = \max\left(R_i^{(2)}(C_V)^{\gamma} C_i^{(2)}(C_V)^{1-\gamma} \right).$$

Other bounds can also be written in reduced form for the above structured matrices.

Bounds for the right eigenvalues of quaternionic matrix polynomials using powers of block companion matrix: We first define the colon notation. Let A := $(A_{ij}) \in M_n(\mathbb{H})$ be a block matrix as in (5.1), then A[s,:] designates the s^{th} block row of the block matrix A as in (5.1) as follows:

$$A[s,:] = \begin{bmatrix} A_{s1} & \dots & A_{sk} \end{bmatrix} \in M_{n \times nk}(\mathbb{H}),$$

and the s^{th} block column of the block matrix A as in (5.1) is specified by

$$A[:,s] := \begin{bmatrix} A_{1s} \\ \vdots \\ A_{ks} \end{bmatrix} \in M_{nk \times n}(\mathbb{H}),$$

where $1 \le s \le k$. Consider the integers p, q, r satisfy $1 \le p \le q \le k$, $1 \le r \le k$. Define

$$A[r, p:q] := \begin{bmatrix} A_{rp} & \dots & A_{rq} \end{bmatrix} \in M_{n \times n(q-p+1)}(\mathbb{H}).$$

Similarly if $1 \le p \le q \le k$, $1 \le c \le k$, then we define

$$A[p:q,c] := \begin{bmatrix} A_{pc} \\ \vdots \\ A_{qc} \end{bmatrix} \in M_{n(q-p+1)\times n}(\mathbb{H}).$$

Now, we present a framework to find the powers of the quaternionic block companion matrix $C_{\mathbf{L}}$ with the help of above colon notation as follows.

Theorem 5.31. Consider $C_{\mathbf{L}} = {n(m-1) \choose n} \left[{0 \qquad I \over -A_0 \qquad -\Delta} \right]$ and let t < m be a positive integer, then

(5.27)
$$C_{\mathbf{L}}^{t} = \begin{array}{c|c} nt & n(m-t) \\ \hline 0 & I \\ \hline C & D \end{array} \right].$$

If $t \geq m$, then

(5.28)
$$C_{\mathbf{L}}^{t} = \begin{bmatrix} C_{\mathbf{L}}^{t-(m-1)}[m, 1:m] \\ C_{\mathbf{L}}^{t-(m-2)}[m, 1:m] \\ \vdots \\ C_{\mathbf{L}}^{t-1}[m, 1:m] \\ C_{\mathbf{L}}^{t}[m, 1:m] \end{bmatrix}_{nm \times nm},$$

where

$$C = \begin{bmatrix} C_{\mathbf{L}}[m, 1:t] \\ C_{\mathbf{L}}^{2}[m, 1:t] \\ \vdots \\ C_{\mathbf{L}}^{t}[m, 1:t] \end{bmatrix}_{nt \times nt} , \quad D = \begin{bmatrix} C_{\mathbf{L}}[m, t+1:m] \\ C_{\mathbf{L}}^{2}[m, t+1:m] \\ \vdots \\ C_{\mathbf{L}}^{t}[m, t+1:m] \end{bmatrix}_{nt \times n(m-t)} ,$$
$$\Delta = \begin{bmatrix} A_{1} \quad A_{2} \quad \dots \quad A_{m} \end{bmatrix}, \quad C_{\mathbf{L}}^{t}[m, 1] := C_{\mathbf{L}}^{t-1}[m, m]C_{\mathbf{L}}[m, 1], \text{ and} \\ C_{\mathbf{L}}^{t}[m, 2:m] := C_{\mathbf{L}}^{t-1}[m, 1:m-1] + C_{\mathbf{L}}^{t-1}[m, m]C_{\mathbf{L}}[m, 2:m].$$

Proof. The proof is similar to the proof method of Theorem 2.35 and by using above notation. \blacksquare

We can now derive the following results similar to the complex case. Consider $A \in M_n(\mathbb{H})$, then

$$\rho_r(A) = \rho_r(A^t)^{1/t} \le ||A^t||_{\beta}^{1/t} \le ||A||_{\beta}, \ t = 1, 2, \dots,$$

where $\|.\|_{\beta}$ is any quaternionic matrix norm. Also we have

$$\rho_r(A) = \lim_{t \to \infty} \left\| A^t \right\|_{\beta}^{1/t}.$$

The following bound for the right eigenvalues of a quaternionic matrix polynomial is derived by using powers of the corresponding quaternionic block companion matrices. **Theorem 5.32.** Let $\mathbf{L} \in \mathbb{L}_m(M_n(\mathbb{H}))$ be as in (5.7) and let $\gamma \in [0, 1]$. Then for any right eigenvalue μ of \mathbf{L} satisfies the following inequality:

$$\frac{1}{\xi_1'} \le |\mu| \le \xi_1,$$

where

$$\begin{aligned} \xi_1 &= \min_{p=1,2,\infty} \left\{ \max_{1 \le i \le m} \left(\left(R_i^{(p)}(C_V^t) \right)^{\gamma/t} \left(C_i^{(p)}(C_V^t) \right)^{\frac{1-\gamma}{t}} \right) \right\}, \\ \xi_1' &= \min_{p=1,2,\infty} \left\{ \max_{1 \le i \le m} \left(\left(R_i^{(p)}(C_S^t) \right)^{\gamma/t} \left(C_i^{(p)}(C_S^t) \right)^{\frac{1-\gamma}{t}} \right) \right\}. \end{aligned}$$

Proof. The proof follows from Theorem 5.2, so we skip the proof. \blacksquare

Next, we derive bounds for the right eigenvalues of quaternionic matrix polynomials by using the matrix 1, 2 and ∞ -norms and powers of corresponding quaternionic block companion matrices which are as follows.

Theorem 5.33. Let $\mathbf{L} \in \mathbb{L}_m(M_n(\mathbb{H}))$ be as in (5.7). Then for any right eigenvalue μ of \mathbf{L} satisfies the following inequality:

(5.29)
$$(\|C_S^t\|_1)^{-1/t} \le |\mu| \le (\|C_V^t\|_1)^{1/t}$$

(5.30)
$$(\|C_S^t\|_{\infty})^{-1/t} \le |\mu| \le (\|C_V^t\|_{\infty})^{1/t},$$

(5.31)
$$(\|C_S^t\|_2)^{-1/t} \le |\mu| \le (\|C_V^t\|_2)^{1/t}.$$

Proof. Consider λ is a right eigenvalue of the block companion matrix C_V , then λ^t ($t \ge 2$ is a positive integer) is a right eigenvalue of the matrix C_V^t . Thus by proof method of Theorem 5.27, we have

(5.32)
$$\rho_r(C_V) \le \|C_V^t\|_{\beta}^{1/t}, \quad (\beta = 1, 2, \infty)$$

Similarly, we have also for C_S

(5.33)
$$\rho_r(C_S) \le \|C_S^t\|_{\beta}^{1/t}, \quad (\beta = 1, 2, \infty).$$

Thus, from (5.32) and (5.33), we have the required results.

Now, we define

$$U := \begin{bmatrix} U_0 & U_1 & \dots & U_{m-1} \end{bmatrix}, \quad \mathcal{L} := \begin{bmatrix} \mathcal{L}_m & \mathcal{L}_{m-1} & \dots & \mathcal{L}_1 \end{bmatrix},$$

where $U_j = V_{m-1}V_j - V_{j-1}$ and $\mathcal{L}_{j+1} = S_1S_{j+1} - S_{j+2}$, $(0 \le j \le m-1)$ with $V_{-1} = S_{m+1} = 0$.

Substituting t = 2 in Theorem 5.33 and applying APPENDIX-B, we obtain the following corollaries.

Corollary 5.34. Let $\mathbf{L} \in \mathbb{L}_m(M_n(\mathbb{H}))$ be as in (5.7). Then for any right eigenvalue μ of \mathbf{L} satisfies the following inequalities:

(5.34)
$$\frac{1}{\beta_4} \le |\mu| \le \alpha_4,$$

(5.35)
$$\max\left(1, \|S\|_{\infty}, \|\mathcal{L}\|_{\infty}\right)^{-1/2} \le |\mu| \le \max\left(1, \|V\|_{\infty}, \|U\|_{\infty}\right)^{1/2},$$

where

$$\alpha_{4} := \left[\max \left(\|V_{0}\|_{1} + \|U_{0}\|_{1}, \|V_{1}\|_{1} + \|U_{1}\|_{1}, 1 + \max_{2 \le i \le m-1} \left(\|V_{i}\|_{1} + \|U_{i}\|_{1} \right) \right) \right]^{1/2},$$

$$\beta_{4} := \left[\max \left(\|S_{m}\|_{1} + \|\mathcal{L}_{m}\|_{1}, \|S_{m-1}\|_{1} + \|\mathcal{L}_{m-1}\|_{1}, 1 + \max_{1 \le i \le m-2} \left(\|S_{i}\|_{1} + \|\mathcal{L}_{i}\|_{1} \right) \right) \right]^{1/2}.$$

Corollary 5.35. Let $\mathbf{L} \in \mathbb{L}_m(M_n(\mathbb{H}))$ be as in (5.7). Then for any right eigenvalue μ of \mathbf{L} satisfies the following inequality:

$$\left(\left\| I_{2n} + TT^{H} \right\|_{2} \right)^{-1/4} \le \left| \mu \right| \le \left(\left\| I_{2n} + DD^{H} \right\|_{2} \right)^{1/4},$$

where $D = \begin{bmatrix} V & 0 \\ 0 & U \end{bmatrix}$ and $T = \begin{bmatrix} S & 0 \\ 0 & \mathcal{L} \end{bmatrix}.$

Proof. Let us consider $P = C_V^2 = X + Y + Z$, where

$$X := \begin{bmatrix} 0 & 0 & \dots & 0 \\ \vdots & \vdots & \ddots & \vdots \\ 0 & 0 & \dots & 0 \\ U_0 & U_1 & \dots & U_{m-1} \end{bmatrix}, Y := \begin{bmatrix} 0 & 0 & \dots & 0 \\ \vdots & \vdots & \ddots & \vdots \\ -V_0 & -V_1 & \dots & -V_{m-1} \\ 0 & 0 & \dots & 0 \end{bmatrix}, Z := \begin{bmatrix} 0 & 0 & I_n & 0 \\ \vdots & \vdots & \ddots & \vdots \\ 0 & 0 & 0 & \dots & I_n \\ 0 & 0 & 0 & \dots & 0 \\ 0 & 0 & 0 & \dots & 0 \end{bmatrix}.$$

Now, we have $||XY^H||_2 = ||XZ^H||_2 = ||YX^H||_2 = ||YZ^H||_2 = ||ZX^H||_2 = ||ZY^H||_2 = 0$. Thus

$$\|P\|_{2}^{2} = \|PP^{H}\|_{2} \leq \|XX^{H} + YY^{H} + ZZ^{H}\|_{2}$$

$$\leq \|I_{mn} + XX^{H} + YY^{H}\|_{2} = \|I + DD^{H}\|_{2}$$

$$\|P\|_{2} \leq (\|I_{2n} + DD^{H}\|_{2})^{1/2}.$$

104

If μ is any right eigenvalue of C_U , then

$$|\mu| \le ||C_V^2||_2^{1/2} \le (||I_{2n} + DD^H||_2)^{1/4}$$

Analogue proof can be given for the lower bound. \blacksquare

We give the following example to illustrate our theory.

Example 5.36. Consider the quaternionic matrix polynomial $\mathbf{L} \in \mathbb{L}_2(M_2(\mathbb{H}))$ of the form

$$\mathbf{L}(\lambda) := A_2 \lambda^2 + A_1 \lambda + A_0$$
, where

$$A_{0} = \begin{bmatrix} 1 + \mathbf{i} & 1 + \mathbf{i} \\ 1 & 1 - \mathbf{i} \end{bmatrix}, A_{1} = \begin{bmatrix} 11 + 2\mathbf{i} + 3\mathbf{j} + \mathbf{k} & 7 + 3\mathbf{i} - \mathbf{j} + 8\mathbf{k} \\ 1 - 8\mathbf{i} - 3\mathbf{j} - 7\mathbf{k} & 9 + 5\mathbf{i} + \mathbf{j} - \mathbf{k} \end{bmatrix}, A_{2} = \begin{bmatrix} 2 + \mathbf{i} & 2 - \mathbf{i} \\ 2 & 1 - \mathbf{i} \end{bmatrix}.$$

Then

Then

$$V_{0} = A_{2}^{-1}A_{0} = \begin{bmatrix} 0.5 - 0.5\mathbf{i} & 1 - 2\mathbf{i} \\ -0.5 + 0.5\mathbf{i} & -2 + \mathbf{i} \end{bmatrix},$$

$$V_{1} = A_{2}^{-1}A_{1} = \begin{bmatrix} -5.5 - 13.5\mathbf{i} - 4\mathbf{j} - 13\mathbf{k} & 4 + 9\mathbf{i} + 2\mathbf{j} - 6\mathbf{k} \\ -3.5 + 15.5\mathbf{i} - 7\mathbf{j} + 12\mathbf{k} & 7 - 6\mathbf{i} - 8\mathbf{j} + 5\mathbf{k} \end{bmatrix},$$

$$S_{1} = A_{0}^{-1}A_{1} = \begin{bmatrix} 3 + \mathbf{i} - 4\mathbf{j} + 4\mathbf{k} & 12 - 6\mathbf{i} + 7\mathbf{k} \\ 3.5 - 5.5\mathbf{i} + 6\mathbf{j} - 5\mathbf{k} & -7 + 4\mathbf{i} + 3.5\mathbf{j} - 2.5\mathbf{k} \end{bmatrix},$$

$$S_{2} = A_{0}^{-1}A_{2} = \begin{bmatrix} 2 - \mathbf{i} & 1 - 2\mathbf{i} \\ -0.5 + 0.5\mathbf{i} & -0.5 + 0.5\mathbf{i} \end{bmatrix}.$$

The right spectrum of $\mathbf{L}(\lambda)$ is

$$\Lambda_r(\mathbf{L}) = [1.5719 + 23.2242\mathbf{i}] \cup [-3.0550 + 4.1765\mathbf{i}] \cup [-0.0806 - 0.0730\mathbf{i}] \cup [0.0637 + 0.0421\mathbf{i}].$$

Also
$$\max_{\lambda_i \in \Lambda_r(\mathbf{L})} |\lambda_i| = 23.2773$$
 and $\min_{\lambda_i \in \Lambda_r(\mathbf{L})} |\lambda_i| = 0.0764$

Let LB:= Lower Bound, UB:= Upper Bound. Thus Theorem 5.23 is verified. Also we have verified our results for the case of the 1-norm, 2-norm and the ∞ -norm.

5.3.2. Left eigenvalues of quaternionic matrix polynomials and their bounds

In this subsection, we state that if the variable of a matrix polynomial is a quaternion identity, then we always be able to find the bounds for the left eigenvalues of quaternionic matrix polynomials. So, if we consider any left matrix polynomial \mathbf{L}' as given in (5.6)

Example 5.36	LB	\mathbf{UB}
Theorem 5.23		
(5.19)	0.0464	34.7804
(5.20)	0.0419	36.9768
$(5.21), \gamma = 1/4$	0.0452	35.3178
Theorem 5.25	0.0469	34.1084
Theorem 5.26	0.0608	24.2378
Theorem 5.28		
(1)	0.0412	42.0443
(2)	0.0383	35.7120
(3)	0.0480	33.9391

TABLE 5.1. Lower and upper bounds for right eigenvalues of $L(\lambda)$.

including the one define in (5.7), then the similar bounds as in the previous section for the right eigenvalues of \mathbf{L} can be derived for the left eigenvalues of \mathbf{L}' .

Let $\mathbf{L}' \in \mathbb{L}'_m(M_n(\mathbb{H}))$ of the form (5.6). Then, we have the following linearization form of the matrix polynomial \mathbf{L}' .

• For the left eigenvalues: The matrix polynomial $\mathbf{L}' \in \mathbb{L}'_m(M_n(\mathbb{H}))$ of the form (5.6) can be written in the linearization form:

$$C_{\mathbf{L}'} + \xi X,$$

where $\xi \in \mathbb{H}, C_{\mathbf{L}'}, X \in M_{mn}(\mathbb{H})$ are of the forms

$$C_{\mathbf{L}'} := \begin{bmatrix} 0 & 0 & 0 & \dots & -A_0 \\ I_n & 0 & 0 & \dots & -A_1 \\ & \ddots & \vdots & \dots & \vdots \\ 0 & 0 & I_n & -A_{m-2} \\ 0 & 0 & \dots & I_n & -A_{m-1} \end{bmatrix}, \quad X := \begin{bmatrix} I_n & 0 & 0 & \dots & 0 \\ 0 & I_n & 0 & \dots & 0 \\ 0 & 0 & I_n & 0 \\ \vdots & \vdots & \ddots & \vdots \\ 0 & 0 & 0 & \dots & A_m \end{bmatrix}.$$

When $A_m = I_n$, the identity matrix, then the matrix polynomial $\mathbf{L}' \in \mathbb{L}'_m(M_n(\mathbb{H}))$ of the form (5.6) is said to be monic matrix polynomial and the corresponding linearization form of it is given by

$$C_{\mathbf{L}'} + \xi E$$
, where $E := I_{nm}$.

Theorem 5.37. The quaternionic matrix polynomial \mathbf{L}' of the form (5.6) and the pencil $(C_{\mathbf{L}'}, X)$ have same left eigenvalues.

Proof. Let $\mu \in \mathbb{H}$ be a left eigenvalue of the quaternionic matrix pencil $(C_{\mathbf{L}'}, X)$. Then there exists $x := [x_1^T, x_2^T, \dots, x_m^T]^T$ with $x_i \in \mathbb{H}^n$, $i = 1, 2, \dots, m$ such that

$$(5.36) \qquad \begin{bmatrix} 0 & 0 & 0 & \dots & -A_0 \\ I_n & 0 & 0 & \dots & -A_1 \\ & \ddots & \vdots & & \vdots \\ 0 & 0 & I_n & & -A_{m-2} \\ 0 & 0 & \dots & I_n & -A_{m-1} \end{bmatrix} \begin{bmatrix} x_1 \\ x_2 \\ \vdots \\ x_{m-1} \\ x_m \end{bmatrix} = \mu \begin{bmatrix} I_n & 0 & 0 & \dots & 0 \\ 0 & I_n & 0 & \dots & 0 \\ \vdots & \vdots & & \ddots & \vdots \\ 0 & 0 & 0 & \dots & A_m \end{bmatrix} \begin{bmatrix} x_1 \\ x_2 \\ \vdots \\ x_{m-1} \\ x_m \end{bmatrix},$$

then (5.36) implies

$$-A_{0}x_{m} = \mu x_{1},$$

$$x_{1} - A_{1}x_{m} = \mu x_{2},$$

$$x_{2} - A_{2}x_{m} = \mu x_{3},$$

$$\vdots$$

$$x_{m-2} - A_{m-2}x_{m} = \mu x_{m-1},$$

$$x_{m-1} - A_{m-1}x_{m} = \mu A_{m}x_{m}$$

By solving the m^{st} system of equations, we obtain

$$A_0 x_m + \mu A_1 x_m + \mu^2 A_2 x_m + \dots + \mu^m A_m x_m = 0.$$

Hence μ is a left eigenvalue of the quaternionic matrix polynomial L'.

Theorem 5.38. Let $\mathbf{L}' \in \mathbb{L}_m(M_n(\mathbb{H}))$ be of the form (5.6). Then $\mu \in \mathbb{H}$ is a left eigenvalue of \mathbf{L}' if and only if $\det[\Psi_{(A_0+\mu A_1+\dots+\mu^m A_m)}] = 0$.

Proof. Consider μ is a left eigenvalue of **L**, then there exists a nonzero $x \in \mathbb{H}^n$ such that $A_0x + \mu A_1x + \mu^2 A_2x + \cdots + \mu^m A_mx = 0$. Applying the mapping Ψ , we have

$$\Psi_{(A_0+\mu A_1+\dots+\mu^m A_m)}\Psi_x=0.$$

It follows that μ is a left eigenvalue of \mathbf{L}' if and only if

$$\det[\Psi_{(A_0+\mu A_1+\dots+\mu^m A_m)}] = 0. \blacksquare$$

We now state a framework to find bounds for the left eigenvalues of quaternionic matrix polynomials.

To find the bounds of the left eigenvalues of the quaternionic matrix polynomial L', we again introduce the new quaternionic matrix polynomials associated with L':

(5.37)
$$\mathbf{L}'_{V''}(\xi) := \xi^m I_n + \sum_{i=0}^{m-1} \xi^i V_i'',$$

where $V_i'' := A_i A_m^{-1}$, i = 0, 1, ..., m - 1, so that $\mathbf{L}'(\xi) = \mathbf{L}'_{V''}(\xi) A_m$, and

(5.38)
$$\mathbf{L}'_{S''}(\xi) := \xi^m I_n + \sum_{i=1}^m \xi^{m-i} S''_i,$$

where $S_i'' := A_i A_0^{-1}$, i = 1, ..., m, now $\xi^m \mathbf{L}'(1/\xi) A_0^{-1} = \mathbf{L}'_{S''}(\xi)$. The matrix polynomials \mathbf{L}' and $\mathbf{L}'_{V''}$ have the same left eigenvalues, whereas the left eigenvalues of the matrix polynomial $\mathbf{L}'_{S''}$ are the reciprocal of the left eigenvalues of the matrix polynomial $\mathbf{L}'_{S''}$ are the reciprocal of the left eigenvalues of the matrix polynomial \mathbf{L}' . The block companion matrices $C_{V''}$ and $C_{S''}$ corresponding to the monic matrix polynomials $\mathbf{L}'_{V''}$ and $\mathbf{L}'_{S''}$ are given by

$$C_{V''} := \binom{n(m-1)}{n} \frac{n}{I} - \frac{-V_0''}{I} \text{ and } C_{S''} := \binom{n}{n(m-1)} \left[\begin{array}{c|c} 0 & -S_m''\\ \hline I & -\Delta_{V''} \end{array} \right],$$

respectively, where

$$\Delta_{V''} = \begin{bmatrix} V_1''^T & V_2''^T & \dots & V_{m-1}''^T \end{bmatrix}^T \in M_{n(m-1)\times n}(\mathbb{H}),$$

$$\Delta_{S''} = \begin{bmatrix} S_{m-1}'' & S_{m-2}'' & \dots & S_1''^T \end{bmatrix}^T \in M_{n(m-1)\times n}(\mathbb{H}).$$

Moreover, the left eigenvalues of the monic matrix polynomials $\mathbf{L}'_{V''}(\xi)$, $\mathbf{L}'_{S''}(\xi)$ and the left eigenvalues of the block companion matrices $C_{V''}$, $C_{S''}$ are same, respectively. **Theorem 5.39.** Consider $C_{\mathbf{L}'} = \binom{n}{n(m-1)} \begin{bmatrix} 0 & | & -A_0 \\ \hline I & | & -\Delta \end{bmatrix}$ and let t < m be a positive integer, then

(5.39)
$$C_{\mathbf{L}'}^t = \frac{nt}{n(m-t)} \begin{bmatrix} 0 & C \\ \hline I & D \end{bmatrix}.$$

If $t \geq m$, then we write

$$C_{\mathbf{L}'}^{t} = \left[\begin{array}{cc} C_{\mathbf{L}'}^{t-(m-1)}[1:m,m] & C_{\mathbf{L}'}^{t-(m-2)}[1:m,m] & \dots & C_{\mathbf{L}'}^{t-1}[1:m,m] & C_{\mathbf{L}'}^{t}[1:m,m] \end{array} \right],$$

$$\begin{split} \text{where } C_{\mathbf{L}'}^t \ \text{are } nm \times nm \ \text{matrices,} \\ \Delta &:= \begin{bmatrix} A_1^T & A_2^T & \dots & A_m^T \end{bmatrix}^T, \ C := \begin{bmatrix} C_{\mathbf{L}'}[1:t,m] & C_{\mathbf{L}'}^2[1:t,m] & \dots & C_{\mathbf{L}'}^t[1:t,m] \end{bmatrix}, \\ D &:= \begin{bmatrix} C_{\mathbf{L}'}[t+1:m,m] & C_{\mathbf{L}'}^2[t+1:m,m] & \dots & C_{\mathbf{L}'}^t[t+1:m,m] \end{bmatrix}, \\ C_{\mathbf{L}'}^t[1,m] &:= & C_{\mathbf{L}'}[1,m] \ C_{\mathbf{L}'}^{t-1}[m,m], \ \text{ and} \\ C_{\mathbf{L}'}^t[2:m,m] &:= & C_{\mathbf{L}'}^{t-1}[1:m-1,m] + C_{\mathbf{L}'}[2:m,m] \ C_{\mathbf{L};}^{t-1}[m,m]. \end{split}$$

Proof. The proof is similar to the proof method of Theorem 2.37 and by using colon notation of Subsection 5.3.1. \blacksquare

Remark 5.40. Similar results can be obtained for the left eigenvalues of the quaternionic matrix polynomial $\mathbf{L}' \in \mathbb{L}'_m(M_n(\mathbb{H}))$.

Next, we define the right quaternionic polynomial as

(5.40)
$$p_r(z) := z^m q_m + z^{m-1} q_{m-1} + \dots + zq_1 + q_0,$$

where $z, q_j \in \mathbb{H} \ (0 \le j \le m)$. The polynomial (5.40) is called monic if $q_m = 1$. Then we write the following remark.

Remark 5.41. If we take n = 1, then the quaternionic matrix polynomial L' will be reduced into the right quaternionic polynomial $p_r(z)$.

Now we have the following observations.

- By Remark 5.41, Theorems 5.25 and 5.26 for left eigenvalue of the matrix polynomial L' yield the bounds for the zeros of the right monic quaternionic polynomial p_r(z).
- Theorem 5.30 for left eigenvalue of the matrix polynomial \mathbf{L}' yields the bounds for the zeros of the right monic quaternionic polynomial $p_r(z)$ which can be found in Theorem 2.26 for the zero of the right monic quaternionic polynomial $p_r(z)$.

The following corollary is a particular case of Theorem 5.28 for the bounds of the left eigenvalues of the quaternionic matrix polynomial $\mathbf{L}' \in \mathbb{L}'_m(M_n(\mathbb{H}))$ which generalizes the bounds for the zeros of right quaternionic polynomials and these can be seen in Corollary 2.27.

Corollary 5.42. Let $\mathbf{L}' \in \mathbb{L}'_m(M_n(\mathbb{H}))$ be as in (5.6). Then for any left eigenvalue μ of \mathbf{L}' satisfies the following inequalities:

(5.41)
$$\left(1 + \max_{1 \le i \le m} \|S_i''\|_1\right)^{-1} \le |\mu| \le 1 + \max_{0 \le i \le m-1} \|V_i''\|_1$$

(5.42)
$$\max\left(1, \sum_{i=1}^{m} \|S_i''\|_{\infty}\right)^{-1} \le |\mu| \le \max\left(1, \sum_{i=0}^{m-1} \|V_i''\|_{\infty}\right).$$

(5.43)
$$\left(1 + \sum_{i=1}^{m} \|S_i''\|_2^2\right)^{-1/2} \le |\mu| \le \left(1 + \sum_{i=0}^{m-1} \|V_i''\|_2^2\right)^{1/2}$$

5.4. Bounds for eigenvalues of complex matrix polynomials

We show that the bounds obtained in Subsections 5.3.1 and 5.3.2 for the left and right eigenvalues are similar to the bounds for the eigenvalues of complex matrix polynomials. Our framework for the bounds of the eigenvalues of quaternionic matrix polynomials are same with the bounds on the eigenvalue of complex matrix polynomials. That is the bounds for the left and right eigenvalues of quaternionic matrix polynomials are same with the bounds for the eigenvalues of a complex matrix polynomial. Bounds for the eigenvalues of a complex matrix polynomial. Bounds for the eigenvalues of a complex matrix polynomial and their applications are given in [4, 15, 38]. If A_m is not invertible in the matrix polynomial $\mathcal{P} \in \mathcal{P}_m(M_n(\mathbb{C}))$ given in (5.5), then $\mathcal{P}(z)$ has an infinite eigenvalue. While if A_0 is not invertible, then 0 is an eigenvalue of the matrix polynomial $\mathcal{P}(z)$. Therefore, for upper bounds of the absolute values of the eigenvalues of $\mathcal{P}(z)$ require A_m to be invertible and lower bounds of the absolute values of eigenvalues of $\mathcal{P}(z)$ require A_0 to be invertible.

For finding the upper and lower bounds for the absolute values of the eigenvalues of the complex matrix polynomial $\mathcal{P}(z)$, we state the following remarks from Section 5.2 and Subsection 5.3.1.

Remark 5.43. It is seen that Theorems 5.1 and 5.2 are true for the eigenvalues of complex block matrices. Then Theorems 5.23 and 5.24 are true for the eigenvalues of complex matrix polynomials.

From Remark 5.43, it is clear that bounds (5.19) and (5.20) for the absolute values of the eigenvalues of complex matrix polynomials are tighter than the bounds given in [15, Lemma 2.3 (2.1), Corollary 2.4 (2.5) (2.6)]. Now we have the following observations from Section 5.2 and Subsection 5.3.1.

- From Remark 5.43 it shows that Theorems 5.32 is true for the eigenvalues of complex matrix polynomials.
- Theorem 5.33 is true for the eigenvalues of complex matrix polynomials for 1,2 and ∞-matrix norms over the complex field.
- It is obvious that Theorem 5.9 is also true for complex block matrices. Hence Theorem 5.25 holds for the eigenvalues of complex matrix polynomials.
- It is also seen that Corollary 5.10 is true for complex matrices. Hence, Theorem 5.26 is valid for the eigenvalues of complex matrix polynomials.

If we take bounds for the absolute values of the eigenvalues of complex matrix polynomials by using powers of block companion matrix, then the bounds (5.29), (5.30) and (5.31) for complex matrix polynomials are sharper than the bounds given in [15, Lemma 2.3].

Example 5.44. Consider the complex matrix polynomial

$$\mathcal{P}(z) := A_5 z^5 + A_4 z^4 + A_3 z^3 + A_2 z^2 + A_1 z + A_0,$$

where

$$A_{0} = \begin{bmatrix} 2+i & 13-2i \\ -2+13i & 4 \end{bmatrix}, A_{1} = \begin{bmatrix} 23i & 7+3i \\ -7+3i & 31+i \end{bmatrix}$$
111

$$A_{2} = \begin{bmatrix} 2 & 1+2i \\ 1-2i & 23i \end{bmatrix}, A_{3} = \begin{bmatrix} 1 & 9-7i \\ 7+9i & 2 \end{bmatrix},$$
$$A_{4} = \begin{bmatrix} 7+i & 3i \\ 3 & 25 \end{bmatrix}, A_{5} = \begin{bmatrix} 11+2i & 10+8i \\ 8-10i & 6-7i \end{bmatrix}.$$

The spectrum of $\mathcal{P}(z)$ is given by

$$\begin{split} \Lambda(\mathcal{P}) &= \begin{cases} 3.3319 - 1.0719i, 1.0701 - 1.3366i, -0.6994 - 1.0580i, -1.3205 + 0.0574i, \\ &-0.8939 + 0.5348i, 0.1973 + 1.1657i, 0.7705 - 0.5866i, 0.2632 + 0.6958i, \\ &0.4865 + 0.0727i, -0.4532 + 0.0166i \\ &\\ &\\ &\sum_{\lambda_i \in \Lambda(\mathcal{P}(z))} |\lambda_i| = 3.5001, \text{ and } \min_{\lambda_i \in \Lambda(\mathcal{P}(z))} |\lambda_i| = 0.4535. \end{split}$$

As we have already explained in this section that Theorems 5.23, 5.32 and 5.33 are true for the eigenvalues of complex matrix polynomials. Thus we now give the following tables for the absolute values of the eigenvalues of $(\mathcal{P}(z))$.

Example 5.44	\mathbf{LB}	UB
Theorem 5.23		
(5.19)	0.2962	8.6818
(5.20)	0.1105	23.1505
(5.21) , $\gamma=1/4$	0.2280	7.6143

TABLE 5.2. Lower and upper bounds for eigenvalues of $\mathcal{P}(z)$.

Example 5.44	\mathbf{LB}	UB
[15, Lemma 2.3]		
(2.1)	0.2501	8.7210
(2.2)	0.1049	23.0150
(2.3)	0.2407	11.3218
Theorem 5.33, $t=2$		
(5.29)	0.3299	6.0610
(5.30)	0.2304	9.3571
(5.31)	0.3473	6.6054
Theorem 5.33, $t=3$		
(5.29)	0.3528	5.2268
(5.30)	0.2954	6.9081
(5.31)	0.3709	5.4567
Theorem 5.33, $t=4$		
(5.29)	0.3853	4.7508
(5.30)	0.3315	5.8152
(5.31)	0.4086	4.8856

TABLE 5.3. Lower and upper bounds for eigenvalues of $\mathcal{P}(z)$.

CHAPTER 6

PERTURBATION BOUNDS FOR RIGHT EIGENVALUES OF QUATERNIONIC MATRICES AND THEIR APPLICATIONS

This chapter deals with the concept of perturbation bounds for right eigenvalues/generalized right eigenvalues of a quaternionic matrix/quaternionic matrix pencil. In particular, Bauer-Fike type theorems for right eigenvalues/generalized right eigenvalues of a diagonalizable quaternionic matrix/diagonalizable quaternionic matrix pencil are derived. Other perturbation bounds for right eigenvalues of a quaternionic matrix are discussed. Furthermore, the location of standard right eigenvalues of a quaternionic matrix and a sufficient condition for the stability of a perturbed quaternionic matrix are given. Perturbation bounds for the zeros of quaternionic polynomials are derived. Finally, we give numerical examples to illustrate our results.

6.1. Introduction

The goal of this chapter is to derive Bauer-Fike type theorems for right eigenvalues/generalized right eigenvalues, perturbation analysis for quaternionic matrices, location of right eigenvalues of perturbed quaternionic matrices, and perturbation bounds for the zeros of quaternionic polynomials.

Bauer-Fike theorems are the standard results in the perturbation theory over the complex field. The applications of the Bauer-Fike theorems over the complex field are given in [7,11,35,36,51]. In general, quaternionic matrix similarity is meaningless for left eigenvalues. However, there are many literatures on matrix similarity and diagonalization for right eigenvalues of a quaternionic matrix. Many results, like the Jordan canonical form, Schur decomposition, singular-value decomposition, diagonalizable of a quaternionic matrix have been extended from the complex field to the skew field of quaternions by different authors. For instance, see [9,21,46,49,61]. However, an extension of Bauer-Fike theorem, perturbation analysis on quaternionic matrices, perturbation analysis of zeros of

quaternionic polynomials have not yet been studied. Perturbation analysis over the skew field of quaternions is important in quantum physics, control theory, and mechanics (see, for example, [1, 26, 43, 49]).

In this chapter, we extend the Bauer-Fike theorems over the complex field to the skew field of quaternions. Specifically, Bauer-Fike type theorems for right eigenvalues/generalized right eigenvalues of a diagonalizable quaternionic matrix/diagonalizable quaternionic matrix pencil are derived. Other perturbation results for right eigenvalues of a quaternionic matrix are given via Jordan canonical form and block diagonal decomposition of a quaternionic matrix. Meanwhile, localization theorems for right eigenvalues of a quaternionic matrix and a sufficient condition for the stability of a perturbed quaternionic matrix are derived. Perturbation for the zeros of the quaternionic polynomials $p_l(z)$ and $p_r(z)$ (defined in Subsection 1.2.4) are given.

6.2. Perturbation analysis on quaternionic matrices

We derive the following lemmas for the development of our theory.

Lemma 6.1. Let $a, b \in \mathbb{C}^+$. Then $|a - b| \leq |\overline{a} - b|$.

Proof. Suppose a = x + iy and b = p + iq, where $x, p \in \mathbb{R}$ and $y, q \in \mathbb{R}^+ \cup \{0\}$. We have

(6.1)
$$|a-b| = |x+iy-(p+iq)| = |(x-p)+i(y-q)|$$
, and

(6.2)
$$|\overline{a} - b| = |x - iy - (p + iq)| = |(x - p) + i(y + q)|$$

It is known that if $y, q \in \mathbb{R}^+ \cup \{0\}$, then $y + q \ge y - q$. Hence from (6.1) and (6.2), we have $|a - b| \le |\overline{a} - b|$.

Lemma 6.2. Consider $\alpha_k, \beta_k, \gamma_k \in \mathbb{R}, (k = 1, 2)$, such that $\alpha_1^2 + \beta_1^2 + \gamma_1^2 = 1$ and $\alpha_2^2 + \beta_2^2 + \gamma_2^2 = 1$. Then

$$\alpha_1 \alpha_2 + \beta_1 \beta_2 + \gamma_1 \gamma_2 \le 1.$$

Proof. We have that

$$(\alpha_{1} - \alpha_{2})^{2} + (\beta_{1} - \beta_{2})^{2} + (\gamma_{1} - \gamma_{2})^{2} = 2 - 2(\alpha_{1}\alpha_{2} + \beta_{1}\beta_{2} + \gamma_{1}\gamma_{2}) \ge 0, i.e.,$$

$$\alpha_{1}\alpha_{2} + \beta_{1}\beta_{2} + \gamma_{1}\gamma_{2} \le 1. \blacksquare$$

Lemma 6.3. Let $\lambda, \mu \in \mathbb{C}^+$ and let $0 \neq \rho, \eta \in \mathbb{H}$. Then

$$|\lambda - \mu| \le \left| \rho^{-1} \lambda \rho - \eta^{-1} \mu \eta \right|.$$

Proof. Consider $\lambda = \lambda' + \lambda'' \mathbf{i}$, $\mu = \mu' + \mu'' \mathbf{i}$, $\rho^{-1} \mathbf{i} \rho = \alpha_1 \mathbf{i} + \beta_1 \mathbf{j} + \gamma_1 \mathbf{k}$, and $\eta^{-1} \mathbf{i} \eta = \alpha_2 \mathbf{i} + \beta_2 \mathbf{j} + \gamma_2 \mathbf{k}$, where $\alpha_k, \beta_k, \gamma_k \in \mathbb{R}$ (k = 1, 2) with $\alpha_1^2 + \beta_1^2 + \gamma_1^2 = \alpha_2^2 + \beta_2^2 + \gamma_2^2 = 1$. Then

(6.3)
$$P = |\lambda - \mu|^2 = (\lambda' - \mu')^2 + (\lambda'')^2 + (\mu'')^2 - 2\lambda''\mu''.$$

Also, $Q = |\rho^{-1}\lambda\rho - \eta^{-1}\mu\eta|^2 = (\lambda' - \mu')^2 + (\lambda''\alpha_1 - \mu''\alpha_2)^2 + (\lambda''\beta_1 - \mu''\beta_2)^2 + (\lambda''\gamma_1 - \mu''\gamma_2)^2 = (\lambda' - \mu')^2 + (\lambda'')^2[\alpha_1^2 + \beta_1^2 + \gamma_1^2] + (\mu'')^2[\alpha_2^2 + \beta_2^2 + \gamma_2^2] - 2\lambda''\mu''[\alpha_1\alpha_2 + \beta_1\beta_2 + \gamma_1\gamma_2].$ From Lemma 6.2, we have

$$\alpha_1\alpha_2 + \beta_1\beta_2 + \gamma_1\gamma_2 < 1.$$

Thus $P \leq Q$, i.e.,

$$|\lambda - \mu| \le \left| \rho^{-1} \lambda \rho - \eta^{-1} \mu \eta \right|.$$

Lemma 6.4. Let $A \in M_n(\mathbb{H})$ such that $A^t = 0_n$ for any positive integer t, where 0_n is the $n \times n$ zero matrix. Then $\Psi_A^t = 0_{2n}$, where 0_{2n} is the $2n \times 2n$ zero matrix.

Proof. Consider $A \in M_n(\mathbb{H})$ such that $A^t = 0_n$ for any positive integer t. Then by taking the complex adjoint matrix of $A^t = 0_n$ and by applying Theorem 1.12, we have

$$\Psi_{A^t} = \Psi_{0_n} \Rightarrow \Psi_A^t = 0_{2n}. \blacksquare$$

Lemma 6.5. Let $T \in M_n(\mathbb{H})$ be partitioned as follows:

$$T = {\begin{array}{*{20}c} p & q \\ p & T_{11} & T_{12} \\ q & T_{22} \end{array}} \right].$$

Define the linear transformation $\phi: M_{p \times q}(\mathbb{H}) \to M_{p \times q}(\mathbb{H})$ by

$$\phi(X) = T_{11}X - XT_{22},$$

where $X \in M_{p \times q}(\mathbb{H})$. Then ϕ is invertible if and only if $\Lambda_r(T_{11}) \cap \Lambda_r(T_{22}) = \emptyset$. If ϕ is invertible and $Y \in M_n(\mathbb{H})$ is defined by

$$Y = \begin{bmatrix} I_p & Z \\ 0 & I_q \end{bmatrix}, \quad \phi(Z) = -T_{12}$$

then $Y^{-1}TY = \text{diag}(T_{11}, T_{22}).$

Proof. Let $X \in M_{p \times q}(\mathbb{H})$. Then singular value decomposition of X is given by

(6.4)
$$U^{H}XV = {\begin{array}{*{20}c}r & q-r\\ p-r & \Sigma_{r} & 0\\ 0 & 0 \\ \end{array}},$$

where $\Sigma_r = \text{diag}(\sigma_i)$ and r = rank(X). Assuming $\phi(X) = 0$ for $X \neq 0$. Substituting (6.4) into the quaternionic matrix equation $T_{11}X = XT_{22}$, we have

$$\begin{bmatrix} A_{11} & A_{12} \\ A_{21} & A_{22} \end{bmatrix} \begin{bmatrix} \Sigma_r & 0 \\ 0 & 0 \end{bmatrix} = \begin{bmatrix} \Sigma_r & 0 \\ 0 & 0 \end{bmatrix} \begin{bmatrix} B_{11} & B_{12} \\ B_{21} & B_{22} \end{bmatrix},$$

where $U^H T_{11}U := (A_{ij})$ and $V^H T_{22}V := (B_{ij})$. By comparing blocks we see that $A_{21} = 0$, $B_{12} = 0$, and $\Lambda_r(A_{11}) = \Lambda_r(B_{11})$. Consequently,

$$\emptyset \neq \Lambda_r(A_{11}) = \Lambda_r(B_{11}) \subseteq \Lambda_r(T_{11}) \cap \Lambda_r(T_{22}).$$

On the other hand, if $\lambda \in \Lambda_r(T_{11}) \cap \Lambda_r(T_{22})$, then there exist nonzero $x, y \in \mathbb{H}^n$ such that

$$T_{11}x = x\lambda, \quad T_{22}^Hy = y\lambda^H.$$

Then $y^H T_{22} = \lambda y^H$ and hence $\phi(xy^H) = 0$. Finally, if ϕ is invertible, then the quaternionic matrix Z exists and

$$Y^{-1}TY = \begin{bmatrix} I & -Z \\ 0 & I \end{bmatrix} \begin{bmatrix} T_{11} & T_{12} \\ 0 & T_{22} \end{bmatrix} \begin{bmatrix} I & Z \\ 0 & I \end{bmatrix}$$
$$= \begin{bmatrix} T_{11} & T_{11}Z - ZT_{22} + T_{12} \\ 0 & T_{22} \end{bmatrix} = \begin{bmatrix} T_{11} & 0 \\ 0 & T_{22} \end{bmatrix}. \blacksquare$$

We next establish block diagonal decomposition of a quaternionic matrix.

Theorem 6.6. (Block diagonal decomposition) Suppose

$$U^{H}AU = T = \begin{bmatrix} T_{11} & T_{12} & \dots & T_{1q} \\ 0 & T_{22} & \dots & T_{2q} \\ \vdots & \vdots & \ddots & \vdots \\ 0 & 0 & \dots & T_{qq} \end{bmatrix}$$
118

is a Schur decomposition of $A \in M_n(\mathbb{H})$ and assume that the T_{ii} are square. If $\Lambda_r(T_{ii}) \cap \Lambda_r(T_{jj}) = \emptyset$ whenever $i \neq j$, then there exists an invertible matrix $Y \in M_n(\mathbb{H})$ such that

$$(QY)^{-1}A(QY) = \operatorname{diag}(T_{11}, \dots, T_{qq})$$

Proof. The proof is immediate by applying Lemma 6.5 and induction. \blacksquare

6.2.1. Bauer-Fike type theorem for the right eigenvalues

We first derive Bauer-Fike type theorem for diagonalizable quaternionic matrices which is as follows.

Theorem 6.7. Let $A \in M_n(\mathbb{H})$ be a diagonalizable matrix, i.e., $A = Y\Lambda Y^{-1}$, $\Lambda = \text{diag}(\lambda_1, \ldots, \lambda_n)$, where λ_i are the standard right eigenvalues of A and Y be an invertible quaternionic matrix. Let $\Delta A \in M_n(\mathbb{H})$. If μ is a standard right eigenvalue of $A + \Delta A$, then

$$\operatorname{dist}(\mu, \Lambda_s(A)) := \min_{\lambda_i \in \Lambda_s(A)} \left\{ |\lambda_i - \mu| \right\} \le K_2(Y) \, \|\Delta A\|_2.$$

Moreover, we have

$$\operatorname{dist}(\xi, \Lambda_r(A)) := \inf_{\eta_j \in \Lambda_r(A)} \left\{ |\eta_j - \xi| \right\} \le K_2(Y) \, \|\Delta A\|_2,$$

where $\xi \in \Lambda_r(A + \Delta A)$ and $K_2(\cdot)$ is the condition number with respect to the matrix 2-norm.

Proof. Let $\lambda_i \neq \mu$ for any *i*. Since μ is a standard right eigenvalue of $A + \Delta A$, then there exists $x \neq 0 \in \mathbb{H}^n$ such that $(A + \Delta A)x = x\mu$. This system is equivalent to the complex system

$$\Psi_{A+\Delta A}\psi_x = \mu\psi_x$$

which implies

$$(\Psi_{A+\Delta A} - \mu I_{2n})\psi_x = 0.$$

From this we have

$$(\Psi_A + \Psi_{\Delta A} - \mu I_{2n})\psi_x = 0.$$

The above system can be written as $(\mu I_{2n} - \Psi_{Y\Lambda Y^{-1}})\psi_x = \Psi_{\Delta A}\psi_x$. Further we can write $\Psi_Y(\mu I_{2n} - \Psi_\Lambda)\Psi_{Y^{-1}}\psi_x = \Psi_{\Delta A}\psi_x$. Then $(\mu I_{2n} - \Psi_\Lambda)(\Psi_Y)^{-1}\psi_x = (\Psi_Y)^{-1}\Psi_{\Delta A}\psi_x$. Thus 119

 $(\Psi_Y)^{-1}\psi_x = (\mu I_{2n} - \Psi_\Lambda)^{-1}[(\Psi_Y)^{-1}\Psi_{\Delta A}\Psi_Y](\Psi_Y)^{-1}\psi_x$. Taking matrix 2-norm (operator norm) on both sides, we get

$$\begin{aligned} \|(\Psi_Y)^{-1}\psi_x\|_2 &\leq \|(\mu I_{2n} - \Psi_\Lambda)^{-1}\|_2 \|(\Psi_Y)^{-1}\Psi_{\Delta A}\Psi_Y\|_2 \|(\Psi_Y)^{-1}\psi_x\|_2 \\ 1 &\leq \|(\mu I_{2n} - \Psi_\Lambda)^{-1}\|_2 \|(\Psi_Y)^{-1}\|_2 \|\Psi_{\Delta A}\|_2 \|\Psi_Y\|_2 \\ 1 &\leq \max_{\lambda_i \in \Lambda_s(A)} \left\{ \frac{1}{|\lambda_i - \mu|}, \frac{1}{|\overline{\lambda_i} - \mu|} \right\} \|(\Psi_Y)^{-1}\|_2 \|\Psi_{\Delta A}\|_2 \|\Psi_Y\|_2. \end{aligned}$$

From Lemma 6.1, we have

$$1 \leq \max_{\lambda_{i} \in \Lambda_{s}(A)} \left\{ \frac{1}{|\lambda_{i} - \mu|} \right\} \| (\Psi_{Y})^{-1} \|_{2} \| \Psi_{\Delta A} \|_{2} \| \Psi_{Y} \|_{2}$$

$$1 \leq \frac{1}{\min_{\lambda_{i} \in \Lambda_{s}} (A) \{ |\lambda_{i} - \mu| \}} K_{2}(\Psi_{Y}) \| \Psi_{\Delta A} \|_{2}$$

$$\min_{\lambda_{i} \in \Lambda_{s}(A)} \{ |\lambda_{i} - \mu| \} \leq K_{2}(\Psi_{Y}) \| \Psi_{\Delta A} \|_{2}.$$

Now from Lemma 1.13, we obtain

$$\min_{\lambda_i \in \Lambda_s(A)} \left\{ |\lambda_i - \mu| \right\} \le K_2(Y) \|\Delta A\|_2.$$

Thus Lemma 6.3 yields

$$\min_{\lambda_i \in \Lambda_s(A)} \{ |\lambda_i - \mu| \} = \inf_{\eta_j \in \Lambda_r(A)} \{ |\eta_j - \xi| \},$$

where $\xi \in \Lambda_r(A + \Delta A)$. Hence

$$\inf_{\eta_j \in \Lambda_r(A)} \{ |\eta_j - \xi| \} \le K_2(Y) \|\Delta A\|_2. \blacksquare$$

In particular, when $A \in M_n(\mathbb{H})$ is normal, Theorem 6.7 leads to the following corollary.

Corollary 6.8. Let $A \in M_n(\mathbb{H})$ be a normal matrix and let μ be a standard right eigenvalue of the perturbed quaternionic matrix $A + \Delta A$. Then

$$\operatorname{dist}(\mu, \Lambda_s(A)) := \min_{\lambda_i \in \Lambda_s(A)} \left\{ |\lambda_i - \mu| \right\} \le \|\Delta A\|_2.$$

Moreover, we have

$$\operatorname{dist}(\xi, \Lambda_r(A)) := \min_{\eta_j \in \Lambda_r(A)} \left\{ |\eta_j - \xi| \right\} \le \|\Delta A\|_2,$$

where $\xi \in \Lambda_r(A + \Delta A)$.

We next have the following theorem for a relative perturbation bound.

Theorem 6.9. Let $A \in M_n(\mathbb{H})$ be diagonalizable and invertible, i.e., $A = Y\Lambda Y^{-1}$, $\Lambda = \text{diag}(\lambda_1, \ldots, \lambda_n)$, where λ_i are the standard right eigenvalues of A and Y be an invertible quaternionic matrix. Let $\Delta A \in M_n(\mathbb{H})$. If μ is a standard right eigenvalue of $A + \Delta A$, then

$$\operatorname{dist}(\mu, \Lambda_s(A)) := \min_{\lambda_i \in \Lambda_s(A)} \left\{ \frac{|\lambda_i - \mu|}{|\lambda_i|} \right\} \le K_2(Y) \, \|A^{-1} \Delta A\|_2.$$

Moreover, we have

$$\operatorname{dist}(\xi, \Lambda_r(A)) := \inf_{\eta_j \in \Lambda_r(A)} \left\{ \frac{|\eta_j - \xi|}{|\eta_j|} \right\} \le K_2(Y) \, \|A^{-1} \Delta A\|_2$$

where $\xi \in \Lambda_r(A + \Delta A)$ and $K_2(Y) = ||Y||_2 ||Y^{-1}||_2$ is the condition number of Y with respect to the matrix 2-norm.

Proof. Let $\lambda_i \neq \mu$ for any *i*. Since μ is a standard right eigenvalue of $A + \Delta A$, there exists $x \neq 0 \in \mathbb{H}^n$ such that $(A + \Delta A)x = x\mu$. This system is equivalent to the complex system

$$\Psi_{A+\Delta A}\psi_x = \mu\psi_x.$$

Since A is an invertible matrix, multiplying by $-\Psi_{A^{-1}}$ from left, we obtain

$$-\Psi_{A^{-1}} (\Psi_A + \Psi_{\Delta A}) \psi_x = -\mu \Psi_{A^{-1}} \psi_x$$
$$(-I_{2n} - \Psi_{A^{-1}\Delta A}) \psi_x = -\mu \Psi_{A^{-1}} \psi_x$$
$$(\mu \Psi_{A^{-1}} - I_{2n}) = \Psi_{A^{-1}\Delta A} \psi_x.$$

The matrix A is diagonalizable, i.e., $A = Y\Lambda Y^{-1}$, we obtain

$$(\mu \Psi_{Y\Lambda^{-1}Y^{-1}} - I_{2n})\psi_x = \Psi_{A^{-1}\Delta A}\psi_x$$
$$\Psi_Y(\mu \Psi_{\Lambda^{-1}} - I_{2n})\Psi_{Y^{-1}}\psi_x = \Psi_{A^{-1}\Delta A}\psi_x$$

Thus by calculation, we get

$$(\Psi_Y)^{-1}\psi_x = (\mu\Psi_{\Lambda^{-1}} - I_{2n})^{-1}[(\Psi_Y)^{-1}\Psi_{A^{-1}\Delta A}\Psi_Y](\Psi_Y)^{-1}\psi_x.$$

By applying the proof method of Theorem 6.7, we have

$$\operatorname{dist}(\mu, \Lambda_s(A)) := \min_{\lambda_i \in \Lambda_s(A)} \left\{ \frac{|\lambda_i - \mu|}{|\lambda_i|} \right\} \le K_2(Y) \, \|A^{-1} \Delta A\|_2$$

As well as

$$\operatorname{dist}(\xi, \Lambda_r(A)) := \inf_{\eta_j \in \Lambda_r(A)} \left\{ \frac{|\eta_j - \xi|}{|\eta_j|} \right\} \le K_2(Y) \, \|A^{-1} \Delta A\|_2,$$

where $\xi \in \Lambda_r(A + \Delta A)$.

Before going to present a localization theorem of standard right eigenvalues of the perturbed quaternionic matrix $A + \Delta A$. We see that a right eigenvalue of the perturbed quaternionic matrix $A + \Delta A$ is not necessarily contained in the union of *n*-discs

(6.5)
$$\Omega_i(A) := \{ z \in \mathbb{C} : |z - \lambda_i| \le K_2(Y) \|\Delta A\|_2 \} \ (1 \le i \le n),$$

where A is diagonalizable, i.e., $A = Y \operatorname{diag}(\lambda_i) Y^{-1}$ and $\lambda_i \in \Lambda_s(A)$. For example, consider a quaternionic matrix $A = \begin{bmatrix} 1 + \mathbf{i} & 0 \\ 0 & \mathbf{i} \end{bmatrix}$. Let $\Delta A = \begin{bmatrix} \epsilon & 0 \\ 0 & \epsilon \end{bmatrix}$. Then $A + \Delta A = \begin{bmatrix} 1 + \mathbf{i} & 0 \\ 0 & \epsilon \end{bmatrix}$ and $\|\Delta A\|_2 = \epsilon$. Since A is normal, $K_2(Y) = 1$. We set $\epsilon = 10^{-3}$. From (6.5), we get the following two discs:

$$\Omega_1(A) = \{ z \in \mathbb{C} : |z - 1 - \mathbf{i}| \le 10^{-3} \} \text{ and } \Omega_2(A) = \{ z \in \mathbb{C} : |z - \mathbf{i}| \le 10^{-3} \}.$$

The perturbed quaternionic matrix $A + \Delta A$ has two standard right eigenvalues $1 + \mathbf{i} + \epsilon$ and $\mathbf{i} + \epsilon$. In particular, $1 - \mathbf{i} + \epsilon$ is also a right eigenvalue of $A + \Delta A$. However it is not contained in any discs. Fortunately, we can show that all the standard right eigenvalues of $A + \Delta A$ are contained in the union of *n*-discs $\Omega_i(A)$ which is as follows.

Theorem 6.10. Let $A \in M_n(\mathbb{H})$ be a diagonalizable matrix, i.e., $A = Y\Lambda Y^{-1}$, $\Lambda = \text{diag}(\lambda_1, \ldots, \lambda_n)$, where λ_i are the standard right eigenvalues of A and Y be an invertible quaternionic matrix. Let $\Delta A \in M_n(\mathbb{H})$. Then all the standard right eigenvalues of the perturbed matrix $A + \Delta A$ are contained in the union of n-discs $\Omega_i(A) := \{z \in \mathbb{C} : |z - \lambda_i| \leq K_2(Y) \|\Delta A\|_2\}$ $(1 \leq i \leq n)$, i.e.,

$$\Lambda_s(A + \Delta A) \subseteq \Omega(A) := \bigcup_{i=1}^n \Omega_i(A).$$

Proof. The proof follows direct from Theorem 6.7.

The following result is a sufficient condition for the stability of the perturbed quaternionic matrix $A + \Delta A$.

Proposition 6.11. Let $A \in M_n(\mathbb{H})$ be a diagonalizable, i.e., $A = Y\Lambda Y^{-1}$, $\Lambda = \text{diag}(\lambda_1, \ldots, \lambda_n)$, where λ_i are the standard right eigenvalues of A and Y be an invertible quaternionic matrix. Let $\Delta A \in M_n(\mathbb{H})$. Assume that

(6.6)
$$Re(\lambda_i) + K_2(Y) \|\Delta A\|_2 < 0 \quad \forall \ i \ (1 \le i \le n).$$

Then the perturbed quaternionic matrix $A + \Delta A$ is stable.
Proof. Let λ be any standard right eigenvalue of A. Then

$$\operatorname{Re}(\lambda) = \operatorname{Re}(\rho^{-1}\lambda\rho) \ \forall \ \rho \in \mathbb{H},$$

i.e., the real part of the standard right eigenvalue λ and the real part of corresponding non standard right eigenvalues are same. Thus, the proof follows from the proof method of Proposition 2.17 and using Theorem 6.10 with Definition 1.24.

We next turn to prove the Bauer-Fike type theorem for central closed matrices.

Theorem 6.12. Let $A \in M_n(\mathbb{H})$ be a central closed matrix, i.e., $A = Y\Lambda Y^{-1}$, $\Lambda = \text{diag}(\lambda_1, \ldots, \lambda_n)$, where λ_i are the standard right eigenvalues of A. Let $\Delta A \in M_n(\mathbb{H})$. If μ is a standard right eigenvalue of $A + \Delta A$, then

$$\operatorname{dist}(\mu, \Lambda_s(A)) := \min_{\lambda_i \in \Lambda_s(A)} \left\{ |\lambda_i - \mu| \right\} \le K_2(Y) \|\Delta A\|_2.$$

Moreover,

$$\operatorname{dist}(\xi, \Lambda_r(A)) := \inf_{\lambda_i \in \Lambda_r(A)} \left\{ |\lambda_i - \xi| \right\} \le K_2(Y) \|\Delta A\|_2,$$

where ξ is a right eigenvalue of $A + \Delta A$ and $K_2(\cdot)$ is the condition number with respect to the matrix 2-norm.

Proof. Let $\lambda_i \neq \mu$ for any *i*. Then it is not trivial. Since μ is a standard right eigenvalue of $A + \Delta A$, there exists $x \neq 0 \in \mathbb{H}^n$ such that $(A + \Delta A)x = x\mu$. By the proof method of Theorem 6.7, we have

$$1 \leq \|(\mu I_{2n} - \Psi_{\Lambda})^{-1}\|_{2} \|(\Psi_{Y})^{-1}\|_{2} \|\Psi_{\Delta A}\|_{2} \|\Psi_{Y}\|_{2}.$$

From Definition 1.8, λ_i $(1 \le i \le n)$ are real. Hence

$$1 \leq \max_{\lambda_{i} \in \Lambda_{s}(A)} \left\{ \frac{1}{|\lambda_{i} - \mu|} \right\} \| (\Psi_{Y})^{-1} \|_{2} \| \Psi_{\Delta A} \|_{2} \| \Psi_{Y} \|_{2}$$

$$1 \leq \frac{1}{\min_{\lambda_{i} \in \Lambda_{s}(A)} \{ |\lambda_{i} - \mu| \}} K_{2}(\Psi_{Y}) \| \Psi_{\Delta A} \|_{2}$$

$$\min_{\lambda_{i} \in \Lambda_{s}(A)} \{ |\lambda_{i} - \mu| \} \leq K_{2}(\Psi_{Y}) \| \Psi_{\Delta A} \|_{2}.$$

By Lemma 1.13, we obtain

$$\min_{\lambda_i \in \Lambda_s(A)} \left\{ |\lambda_i - \mu| \right\} \le K_2(Y) \|\Delta A\|_2.$$

Since $|\lambda_i - \mu| = |\lambda_i - \rho^{-1}\mu\rho|, \ 0 \neq \rho \in \mathbb{H}$. Hence

$$\inf_{\lambda_i \in \Lambda_r(A)} \{ |\lambda_i - \eta| \} \le K_2(Y) \|\Delta A\|_2,$$

where $\eta \in \Lambda_r(A + \Delta A)$.

The following theorem is a direct consequence of Theorem 6.12.

Theorem 6.13. Let $A \in M_n(\mathbb{H})$ be a central closed matrix, i.e., $A = Y\Lambda Y^{-1}$, $\Lambda = \text{diag}(\lambda_1, \ldots, \lambda_n)$, where λ_i are the standard right eigenvalues of A. Let $\Delta A \in M_n(\mathbb{H})$. Then all the right eigenvalues of $A + \Delta A$ are contained in the union of n-balls $\mathcal{G}_i(A) := \{z \in \mathbb{H} : |z - \lambda_i| \leq K_2(Y) \|\Delta A\|_2\}$ $(1 \leq i \leq n), i.e.,$

$$\Lambda_r(A + \Delta A) \subseteq \mathcal{G}(A) := \bigcup_{i=1}^n \mathcal{G}_i(A).$$

Proof. The proof follows from Theorem 6.12. \blacksquare

Since real numbers commute with quaternions. Therefore, real left eigenvalues of a matrix $A \in M_n(\mathbb{H})$ are also right eigenvalues of A and vice versa. By applying this argument, we have the following result.

Theorem 6.14. Let $A \in M_n(\mathbb{H})$ be a Hermitian matrix. For some $\tilde{\mu} \in \mathbb{R}$ and $\tilde{x} \in \mathbb{H}^n$ with $\|\tilde{x}\|_2 = 1$, define residual vector $r = A\tilde{x} - \tilde{\mu}\tilde{x}$. Then $|\tilde{\mu} - \mu| \leq \|r\|_2$ for some $\mu \in \Lambda_r(A)$.

Proof. Since $\mu \in \Lambda_r(A)$, $\mu \in \Lambda_l(A)$. Now if $\tilde{\mu} \notin \Lambda_r(A)$, then $\tilde{\mu} \notin \Lambda_l(A)$. Hence $(A - \tilde{\mu}I_n)^{-1}$ exists. So we can write $r = A\tilde{x} - \tilde{\mu}\tilde{x}$ as follows:

$$\tilde{x} = (A - \tilde{\mu}I_n)^{-1}r.$$

Since A is a Hermitian matrix. Therefore, from Theorem 1.29, A is unitarily diagonalizable, i.e., $V^{-1}AV = \text{diag}(\mu_i)$ $(1 \le i \le n)$, where V is a quaternionic unitary matrix. Now, by applying the proof method of Theorem 6.7, we have the required result

$$\min_{\mu_i \in \Lambda_{\mathbf{s}}(A)} \left\{ |\tilde{\mu} - \mu_i| \right\} \le ||r||_2. \blacksquare$$

6.2.2. Perturbation bounds for non-diagonalizable quaternionic matrices

First, in this subsection, we derive a perturbation result on quaternionic matrices via block diagonal decomposition of quaternionic matrices. **Theorem 6.15.** Let $A \in M_n(\mathbb{H})$. Consider $A = YTY^{-1}$, $T = \text{diag}(V_1, \ldots, V_k)$, where $V_i = \Lambda_i + N_i \in M_{n_i}(\mathbb{H})$ is upper triangular, Λ_i is diagonal and N_i is strict upper triangular for $i = 1, 2, \ldots, k$. If μ is a standard right eigenvalue of $A + \Delta A$, then there exists $\lambda_j \in \Lambda_s(A)$ such that

$$|\lambda_j - \mu| \le \max\left(\chi, \chi^{1/n_j}\right),$$

where $\chi = \|Y^{-1}\Delta AY\|_2 \sum_{t=0}^{n_j-1} \|N_j\|_2^t$, and $N_j^{n_j} = 0_{n_j}$ with $N_j^{n_j-1} \neq 0_{n_j}$.

Proof. Let $\mu \notin \Lambda_s(A)$. Since μ is a standard right eigenvalue of $A + \Delta A$, there exists $x \neq 0 \in \mathbb{H}^n$ such that $(A + \Delta A)x = x\mu$. Then from the proof method of Theorem 6.7, we have

(6.7)
$$1 \leq \|(\mu I_{2n} - \Psi_T)^{-1}\|_2 \|Y^{-1} \Delta A Y\|_2$$

We have that

(6.8)
$$\frac{1}{\|(\mu I_{2n} - \Psi_T)^{-1}\|_2} = \frac{1}{\max_{1 \le i \le k} \|(\mu I_{2n_i} - \Psi_{V_i})^{-1}\|_2} = \frac{1}{\|(\mu I_{2n_j} - \Psi_{V_j})^{-1}\|_2}$$

Define

(6.9)
$$\tau = \frac{1}{\|(\mu I_{2n_j} - \Psi_{\Lambda_j})^{-1}\|_2} = \frac{1}{\max\left\{\frac{1}{|\mu - \lambda_j|}, \frac{1}{|\mu - \overline{\lambda_j}|}\right\}} = |\mu - \lambda_j|.$$

Since $N_j^{n_j} = 0$, from Lemma 6.4, we have $\Psi_{N_j}^{n_j} = 0$. As N_j is a strict upper triangular quaternionic matrix, Ψ_{N_j} is also a strict upper triangular. Hence $(\mu I_{2n_j} - \Psi_{\Lambda_j}) \Psi_{N_j} = \Psi_{N_j} (\mu I_{2n_j} - \Psi_{\Lambda_j})$. Therefore,

$$\left[\left(\mu I_{2n_j} - \Psi_{\Lambda_j}\right)^{-1} \Psi_{N_j}\right]^{n_j} = 0.$$

Consequently, we get

(6.10)
$$\left(\mu I_{2n_j} - \Psi_{V_j}\right)^{-1} = \sum_{t=0}^{n_j-1} (-1)^t \left[\left(\mu I_{2n_j} - \Psi_{\Lambda_j}\right)^{-1} \Psi_{N_j} \right]^t \left(\mu I_{2n_j} - \Psi_{\Lambda_j}\right)^{-1}$$

Then, we have

(6.11)
$$\| (\mu I_{2n} - \Psi_T)^{-1} \|_2 = \| (\mu I_{2n_j} - \Psi_{V_j})^{-1} \|_2 \le \frac{1}{\tau} \sum_{t=0}^{n_j-1} \left[\frac{\|N_j\|_2}{\tau} \right]^t.$$

If $\tau > 1$, then

(6.12)
$$\| \left(\mu I_{2n_j} - \Psi_{V_j} \right)^{-1} \|_2 \le \frac{1}{\tau} \sum_{t=0}^{n_j - 1} \| N_j \|_2^t.$$

From (6.7), (6.9), and (6.12), we obtain

(6.13)
$$1 \le \frac{1}{\tau} \sum_{t=0}^{n_j-1} \|N_j\|_2^t \left(\|Y^{-1}\Delta AY\|_2\right).$$

Setting $\chi = \|Y^{-1}\Delta AY\|_2 \sum_{t=0}^{n_j-1} \|N_j\|_2^t$. Thus $\tau \leq \chi$. If $\tau \leq 1$, then

(6.14)
$$\| (\mu I_{2n} - \Psi_T)^{-1} \|_2 = \| (\mu I_{2n_j} - \Psi_{V_j})^{-1} \|_2 \le \frac{1}{\tau^{n_j}} \sum_{t=0}^{n_j-1} \| N_j \|_2^t.$$

Hence, (6.7), (6.9), and (6.14) yield

(6.15)
$$1 \le \frac{1}{\tau^{n_j}} \sum_{t=0}^{n_j-1} \|N_j\|_2^t \left(\|Y^{-1}\Delta AY\|_2\right).$$

Thus $\tau^{n_j} \leq \chi$ and hence $\tau \leq \chi^{\frac{1}{n_j}}$. From the above it is clear that

$$|\lambda_j - \mu| \le \max\left(\chi, \chi^{1/n_j}\right).$$

where $\chi = \|Y^{-1} \Delta AY\|_2 \sum_{t=0}^{n_j - 1} \|N_j\|_2^t$.

Next, we present a perturbation result on quaternionic matrices via the Jordan canonical form of a quaternionic matrix.

Theorem 6.16. Let $A \in M_n(\mathbb{H})$ with $Y^{-1}AY = J = \text{diag}(J_{m_i}(\lambda_i))$, where $J_{m_i}(\lambda_i)$ $(1 \le i \le t)$ are Jordan blocks of A. Let $\Delta A \in M_n(\mathbb{H})$. If μ is a standard right eigenvalue of $A + \Delta A$, then

$$\min_{1 \le i \le t} \left\{ \frac{1}{\| (J_{m_i}(\lambda_i) - \mu I_{m_i})^{-1} \|_2}, \frac{1}{\| (\overline{J_{m_i}(\lambda_i)} - \mu I_{m_i})^{-1} \|_2} \right\} \le K_2(Y) \| \Delta A \|_2,$$

where $K_2(\cdot)$ is the condition number with respect to the matrix 2-norm.

Proof. If μ is not a standard right eigenvalue of any Jordan block matrices $J_{m_i}(\lambda_i)$, then it is not trivial. Since μ is a standard right eigenvalue of $A + \Delta A$, there exists $x \neq 0 \in \mathbb{H}^n$ such that

$$(A + \Delta A)x = x\mu,$$
126

and this system is equivalent to the complex system

$$\Psi_{A+\Delta A}\psi_x = \mu\psi_x \Leftrightarrow (\Psi_{A+\Delta A} - \mu I_{2n})\psi_x = 0 \Leftrightarrow (\Psi_A + \Psi_{\Delta A} - \mu I_{2n})\psi_x = 0.$$

Since A has a Jordan canonical form via the invertible quaternionic matrix Y. Hence

$$(\mu I_{2n} - \Psi_{Y \operatorname{diag}(J_{m_i}(\lambda_i))Y^{-1}})\psi_x = \Psi_{\Delta A}\psi_x$$
$$\Psi_Y(\mu I_{2n} - \Psi_{\operatorname{diag}(J_{m_i}(\lambda_i))})\Psi_{Y^{-1}}\psi_x = \Psi_{\Delta A}\psi_x$$
$$(\mu I_{2n} - \Psi_{\operatorname{diag}(J_{m_i}(\lambda_i))})(\Psi_Y)^{-1}\psi_x = (\Psi_Y)^{-1}\Psi_{\Delta A}\psi_x.$$

This implies that $(\Psi_Y)^{-1}\psi_x = (\mu I_{2n} - \Psi_{\text{diag}(J_{m_i}(\lambda_i))})^{-1}[(\Psi_Y)^{-1}\Psi_{\Delta A}\Psi_Y](\Psi_Y)^{-1}\psi_x.$

Taking matrix 2-norm on both sides of the above equation, we obtain

$$\|(\Psi_Y)^{-1}\psi_x\|_2 \leq \|(\mu I_{2n} - \Psi_{\operatorname{diag}(J_{m_i}(\lambda_i))})^{-1}\|_2 \|[(\Psi_Y)^{-1}\Psi_{\Delta A}\Psi_Y]\|_2 \|(\Psi_Y)^{-1}\psi_x\|_2$$

$$1 \leq \|(\mu I_{2n} - \Psi_{\operatorname{diag}(J_{m_i}(\lambda_i))})^{-1}\|_2 \|(\Psi_Y)^{-1}\|_2 \|\Psi_{\Delta A}\|_2 \|\Psi_Y\|_2$$

$$1 \leq \max_{1 \leq i \leq t} \{ \| (J_{m_i}(\lambda_i) - \mu I_{m_i})^{-1} \|_2, \| (\overline{J_{m_i}(\lambda_i)} - \mu I_{m_i})^{-1} \|_2 \} \| (\Psi_Y)^{-1} \|_2 \| \Psi_{\Delta A} \|_2 \| \Psi_Y \|_2$$

$$1 \leq \max_{1 \leq i \leq t} \{ \| (J_{m_i}(\lambda_i) - \mu I_{m_i})^{-1} \|_2, \| (\overline{J_{m_i}(\lambda_i)} - \mu I_{m_i})^{-1} \|_2 \} \| (\Psi_Y)^{-1} \|_2 \| \Psi_{\Delta A} \|_2 \| \Psi_Y \|_2.$$

The above inequality can be rewritten as

$$\frac{1}{\max_{1\leq i\leq t}\left\{\|(J_{m_i}(\lambda_i)-\mu I_{m_i})^{-1}\|_2, \|(\overline{J_{m_i}(\lambda_i)}-\mu I_{m_i})^{-1}\|_2\right\}} \leq \|(\Psi_Y)^{-1}\|_2 \|\Psi_{\Delta A}\|_2 \|\Psi_Y\|_2,$$

i.e.,

$$\min_{1 \le i \le t} \left\{ \frac{1}{\|(J_{m_i}(\lambda_i) - \mu I_{m_i})^{-1}\|_2}, \frac{1}{\|(\overline{J_{m_i}(\lambda_i)} - \mu I_{m_i})^{-1}\|_2} \right\} \le K_2(\Psi_Y) \|\Psi_{\Delta A}\|_2$$

Then, from Lemma 1.13, we have

$$\min_{1 \le i \le t} \left\{ \frac{1}{\|(J_{m_i}(\lambda_i) - \mu I_{m_i})^{-1}\|_2}, \frac{1}{\|(\overline{J_{m_i}(\lambda_i)} - \mu I_{m_i})^{-1}\|_2} \right\} \le K_2(Y) \|\Delta A\|_2. \quad \blacksquare$$

Remark 6.17. Obviously from Lemma 5.6, all the results in the 2-norm (operator norm) imply that for the Frobenius norm.

We give the following localization theorem for standard right eigenvalues of the perturbed quaternionic matrix $A + \Delta A$ when A is not a diagonalizable matrix. **Theorem 6.18.** Let $A \in M_n(\mathbb{H})$ with $Y^{-1}AY = J = \text{diag}(J_{m_i}(\lambda_i))$, where $J_{m_i}(\lambda_i)$ $(1 \le i \le t)$ are Jordan blocks of A. Let $\Delta A \in M_n(\mathbb{H})$. Then all the standard right eigenvalues of $A + \Delta A$ are contained in the union of t-sets $P_i(A) := T_i(A) \cup K_i(A)$, where

$$T_{i}(A) := \{ z \in \mathbb{C} : \| (J_{m_{i}}(\lambda_{i}) - \mu I_{m_{i}})^{-1} \|_{2}^{-1} \leq K_{2}(Y) \| \Delta A \|_{2} \},$$

$$K_{i}(A) := \{ z \in \mathbb{C} : \| (\overline{J_{m_{i}}(\lambda_{i})} - \mu I_{m_{i}})^{-1} \|_{2}^{-1} \leq K_{2}(Y) \| \Delta A \|_{2} \}, i.e.,$$

$$\Lambda_{s}(A + \Delta A) \subseteq P(A) := (\cup_{i=1}^{t} T_{i}(A)) \cup (\cup_{i=1}^{t} K_{i}(A)).$$

Proof. The proof follows from Theorem 6.16. \blacksquare

6.3. Perturbation bounds for the zeros of quaternionic polyno-

mials

Theorem 6.19. Let $p_l(z) = z^m + \sum_{k=0}^{m-1} q_k z^k$ be a quaternionic simple monic polynomial. Let $C_{p_l} = Y \operatorname{diag}(V_1, V_2, \ldots, V_t) Y^{-1}$ with $V_i = \Lambda_i + N_i \in M_{n_i}(\mathbb{H})$ is upper triangular, Λ_i is diagonal and N_i is strict upper triangular for $i = 1, 2, \ldots, t$. Assume that $\hat{p}_l(z) = z^m + \sum_{k=0}^{m-1} \hat{q}_k z^k$ is a perturbation of $p_l(z)$ with $\hat{q}_k = q_k + \Delta q_k$, $|\Delta q_k| \leq \epsilon$, $(0 \leq k \leq m-1)$. Then for any complex zero $\hat{z}_k \in Z_{\mathbb{C}}(\hat{p}_l(z))$, there exists a complex zero $z_j \in Z_{\mathbb{C}}(p_l(z))$ such that

$$|\widehat{z_k} - z_j| \le \max(\chi, \chi^{1/n_j}),$$

where $\chi := \|Y^{-1}\Delta C_{p_l}Y\|_2 \sum_{\eta=0}^{n_j-1} \|N_j\|_2^{\eta}$, and $\Delta C_{p_l} := -e_m[\Delta q_0, \dots, \Delta q_{m-1}]$ with $e_m := [0, \dots, 0, 1]^T \in \mathbb{R}^m$.

Proof. Let us consider the corresponding companion matrix C_{p_l} to the simple monic polynomial $p_l(z)$ such that $C_{p_l} = Y \operatorname{diag}(V_1, V_2, \ldots, V_t) Y^{-1}$. Since $\Delta C_{p_l} = -e_m[\Delta q_0, \ldots, \Delta q_{m-1}]$, therefore

$$C_{p_l} + \Delta C_{p_l} = \begin{bmatrix} 0 & 1 & 0 & \dots & 0 \\ 0 & 0 & 1 & & 0 \\ \vdots & \vdots & \vdots & \ddots & \vdots \\ 0 & 0 & 0 & \dots & 1 \\ -\widehat{q_0} & -\widehat{q_1} & -\widehat{q_2} & \dots & -\widehat{q_{m-1}} \end{bmatrix}.$$
128

It is known that the left eigenvalues of C_{p_l} and the zeros of $p_l(z)$ are same. Also the left spectrum of C_{p_l} falls in the right spectrum of C_{p_l} . Thus by the proof method of Theorem 6.15, we get the desired result.

Theorem 6.20. Let $p_l(z) = z^m + \sum_{k=0}^{m-1} q_k z^k$ be a quaternionic simple monic polynomial and its companion matrix $C_{p_l} = YDY^{-1}, D = \text{diag}(\lambda_1, \ldots, \lambda_m), \lambda_t \in \Lambda_s(A), t = 1, 2, \ldots, m$. Assume that $\hat{p}_l(z) = z^m + \sum_{k=0}^{m-1} \hat{q}_k z^k$ is a perturbation of $p_l(z)$ with $\hat{q}_k = q_k + \Delta q_k, |\Delta q_k| \le \epsilon, (0 \le k \le m-1)$. Then the zeros of $p_l(z)$ and $\hat{p}_l(z)$ can be given as

$$\inf_{\widehat{z}_i \in Z_{\mathbb{H}}(\widehat{p}_l(z))} |\widehat{z}_i - z_j| \le K_2(Y) \|\Delta C_{p_l}\|_2,$$
$$z_j \in Z_{\mathbb{H}}(p_l(z))$$

where $\Delta C_{p_l} := -e_m[\Delta q_0, \dots, \Delta q_{m-1}]$ with $e_m := [0, \dots, 0, 1]^T \in \mathbb{R}^m$.

Proof. By applying Theorem 6.7 and the proof method of Theorem 6.19, we get the desired result. ■

Remark 6.21. Similar results can be obtained for the zeros of $p_r(z)$ as well.

6.4. Bauer-Fike type theorem for generalized right eigenvalues

Let $\mathbb{L}_1(M_n(\mathbb{H}))$ be the space of matrix pencils over a quaternion division algebra. $\mathbb{L}_1 \in \mathbb{L}_1(M_n(\mathbb{H}))$ is defined as

(6.16)
$$\mathbf{L}_1(\lambda) := A + \lambda B,$$

where $A, B \in M_n(\mathbb{H})$ and λ commutes with the quaternionic matrices. The generalized right eigenvalue of $\mathbf{L}_1 \in \mathbb{L}_1(M_n(\mathbb{H}))$ of the form (6.16) is defined as follows.

Definition 6.22. Let $\mathbf{L}_1 \in \mathbb{L}_1(M_n(\mathbb{H}))$ be as in (6.16) and let $\mu \in \mathbb{H}$. Then μ is called a generalized right eigenvalue of \mathbf{L}_1 if

$$Ax = Bx\mu$$

for some nonzero $x \in \mathbb{H}^n$. Here x is called the right eigenvector corresponding to the generalized right eigenvalue μ . The set of generalized right eigenvalues of \mathbf{L}_1 is called generalized right spectrum of \mathbf{L}_1 , denoted by $\Lambda_r(\mathbf{L}_1)$.

Definition 6.23. Let $\mathbf{L}_1 \in \mathbb{L}_1(M_n(\mathbb{H}))$ be as in (6.16). Then the matrix pencil \mathbf{L}_1 is called regular if there exists $\alpha \in \mathbb{R}$ such that $A + \alpha B$ is an invertible matrix.

Next, we give the definition of generalized standard right eigenvalues of quaternionic matrix pencils.

Definition 6.24. Let $\mathbf{L}_1 \in \mathbb{L}_1(M_n)(\mathbb{H})$ be as in (6.16). Then we define a set of generalized standard right eigenvalues of a regular matrix pencil \mathbf{L}_1 as

$$\Lambda_s(\mathbf{L}_1) := \{ \alpha \in \mathbb{C}_\infty : Ax = Bx\alpha, \ 0 \neq x \in \mathbb{H}^n, \ \Im(\alpha) \ge 0 \}, \ \mathbb{C}_\infty = \mathbb{C} \cup \{\infty\}.$$

We have the quaternionic generalized right eigenvalue problem

$$Ax = Bx\mu.$$

If B is an invertible matrix, then $B^{-1}Ax = x\mu$. Moreover, $AB^{-1}(Bx) = (Bx)\mu$. Putting $Bx = y \in \mathbb{H}^n$, $AB^{-1}y = y\mu$. Hence,

$$\Lambda_r(\mathbf{L}_1) = \Lambda_r(A^{-1}B) = \Lambda_r(AB^{-1}).$$

Theorem 6.25. Let $A + \lambda B$ (defined in (6.16)) be a quaternionic matrix pencil and let $\mu \in \mathbb{H}$ be a generalized right eigenvalue of $A + \lambda B$ with eigenvector $x \in \mathbb{H}^n$. Then μ is also a generalized right eigenvalue of the quaternionic matrix pencil $P^HAQ + \lambda P^HBQ$ with eigenvector $Q^{-1}x \in \mathbb{H}^n$, where $P, Q \in M_n(\mathbb{H})$ are invertible matrices.

Proof. Let μ be a generalized right eigenvalue of the quaternionic matrix pencil $A + \lambda B$. Then

$$(6.17) Ax = Bx\mu$$

for some nonzero $x \in \mathbb{H}^n$. If P and Q are invertible quaternionic matrices, then (6.17) is equivalent to the following generalized right eigenvalue problem

$$P^H A Q(Q^{-1}x) = P^H B Q(Q^{-1}x)\mu. \blacksquare$$

Theorem 6.26. Let $\mathbf{L}_1 \in \mathbb{L}_1(M_n(\mathbb{H}))$ be a quaternionic matrix pencil such that A and B are Hermitian matrices and B be a positive definite. Then there exists an invertible quaternionic matrix Q such that $Q^H A Q = \operatorname{diag}(\mu_i)$ with μ_i are the generalized standard right eigenvalues of \mathbf{L}_1 and $Q^H B Q = I_n$.

Proof. Let λ be a generalized right eigenvalue of \mathbf{L}_1 . Then there exists some nonzero $x \in \mathbb{H}^n$ such that

Since B is a Hermitian positive definite matrix, B has a positive definite square root $B^{\frac{1}{2}}$. Then the generalized right eigenvalue problem (6.18) is equivalent to the following quaternionic eigenvalue problem.

(6.19)
$$B^{-\frac{1}{2}}AB^{-\frac{1}{2}}y = y\lambda,$$

where $y = (B^{-\frac{1}{2}}x) \in \mathbb{H}^n$. Also, we can see that

$$(B^{-\frac{1}{2}}AB^{-\frac{1}{2}})^{H} = (B^{-\frac{1}{2}})^{H}A^{H}(B^{-\frac{1}{2}})^{H} = B^{-\frac{1}{2}}AB^{-\frac{1}{2}}.$$

Hence $B^{-\frac{1}{2}}AB^{-\frac{1}{2}}$ is a Hermitian matrix. From Theorem 1.29, there exists a unitary matrix $V \in M_n(\mathbb{H})$ such that

$$B^{-\frac{1}{2}}AB^{-\frac{1}{2}} = V \operatorname{diag}(\mu_i)V^H,$$

where $\mu_i \in \Lambda_r(B^{-\frac{1}{2}}AB^{-\frac{1}{2}})$. Setting $Q = B^{-\frac{1}{2}}V$, $Q^H A Q = \text{diag}(\mu_i)$ and $Q^H B Q = I$. Thus from (6.19) and Theorem 6.25, we have the required result.

Now, we develop Bauer-Fike type theorem for diagonalizable regular quaternionic matrix pencils.

Theorem 6.27. Let $\mathbf{L}_1 \in \mathbb{L}_1(M_n(\mathbb{H}))$ be a diagonalizable regular quaternionic matrix pencil with $\Lambda_s(\mathbf{L}_1) = \{\mu_1, \mu_2, \dots, \mu_n\}$, i.e., $\mathbf{L}_1(\lambda) := A + \lambda B = P(D + \lambda I_n)Q$, where $P, Q \in M_n(\mathbb{H})$ are invertible matrices, $D = \operatorname{diag}(\mu_1, \mu_2, \dots, \mu_n)$ and λ commutes with the quaternionic matrices. If μ is a standard generalized right eigenvalue of the perturbed quaternionic matrix pencil $\mathbf{L}_1 + \Delta \mathbf{L}_1 \in \mathbb{L}_1(M_n(\mathbb{H}))$, then

$$\min_{\mu_i \in \Lambda_s(\mathbf{L}_1)} |\mu_i - \mu| \le \|P^{-1}\|_2 \|Q^{-1}\|_2 \|E\|_2,$$

where $(\mathbf{L}_1 + \Delta \mathbf{L}_1)\lambda := (A + \Delta A) + \lambda(B + \Delta B)$ and $E := \mu \Psi_{\Delta B} - \Psi_{\Delta A}$

Proof. If $\mu \in \Lambda_s(\mathbf{L}_1)$, then the result follows. Assume that $\mu \notin \Lambda_s(\mathbf{L}_1)$. Since $\mu \in \Lambda_s(\mathbf{L}_1 + \Delta \mathbf{L}_1)$, there exists $0 \neq x \in \mathbb{H}^n$ such that

$$(A + \Delta A)x = (B + \Delta B)x\mu.$$

This system is equivalent to the complex system

$$\Psi_{A+\Delta A}\psi_x = \mu\Psi_{B+\Delta B}\psi_x$$
$$(\Psi_A - \mu\Psi_B)\psi_x = (\mu\Psi_{\Delta B} - \Psi_{\Delta A})\psi_x$$
$$(\Psi_{PDQ} - \mu\Psi_{PI_nQ})\psi_x = (\mu\Psi_{\Delta B} - \Psi_{\Delta A})\psi_x$$
131

$$\Psi_P(\Psi_D - \mu I_{2n})\Psi_Q\psi_x = (\mu\Psi_{\Delta B} - \Psi_{\Delta A})\psi_x$$
$$(\Psi_D - \mu I_{2n})\Psi_Q\Psi_x = \Psi_P^{-1}(\mu\Psi_{\Delta B} - \Psi_{\Delta A})\psi_x$$
$$\Psi_Q\psi_x = (\Psi_D - \mu I_{2n})^{-1}\Psi_P^{-1}(\mu\Psi_{\Delta B} - \Psi_{\Delta A})\Psi_Q^{-1}(\Psi_Q\psi_x).$$

Taking the matrix 2-norm on both sides, we have

$$\|\Psi_{Q}\psi_{x}\|_{2} \leq \|(\Psi_{D} - \mu I_{2n})^{-1}\|_{2} \|\Psi_{P}^{-1}\|_{2} \|(\mu\Psi_{\Delta B} - \Psi_{\Delta A})\|_{2} \|\Psi_{Q}^{-1}\|_{2} \|\Psi_{Q}\psi_{x}\|_{2}$$
$$1 \leq \max_{\mu_{i}\in\Lambda_{s}(\mathbf{L}_{1})} \left\{\frac{1}{|\mu_{i} - \mu|}, \frac{1}{|\overline{\mu_{i}} - \mu|}\right\} \|\Psi_{P}^{-1}\|_{2} \|(\mu\Psi_{\Delta B} - \Psi_{\Delta A})\|_{2} \|\Psi_{Q}^{-1}\|_{2}.$$

By Lemmas 1.13 and 6.1 and putting $E := \mu \Psi_{\Delta B} - \Psi_{\Delta A}$, we have the following expression.

$$1 \leq \frac{1}{\min_{\mu_i \in \Lambda_s(\mathbf{L}_1)} \{ |\mu_i - \mu| \}} \| P^{-1} \|_2 \| Q^{-1} \|_2 \| E \|_2$$
$$\min_{\mu_i \in \Lambda_s(\mathbf{L}_1)} \{ |\mu_i - \mu| \} \leq \| P^{-1} \|_2 \| Q^{-1} \|_2 \| E \|_2. \blacksquare$$

Numerical Examples: Here, we give some numerical examples to illustrate our results.

Example 6.28. Let us consider a quaternionic matrix

$$A = \begin{bmatrix} 2\mathbf{i} & -2\mathbf{j} & \mathbf{j} + \mathbf{k} \\ -\mathbf{k} & 2 & -1 \\ -\mathbf{j} & 1 - \mathbf{i} & 1 \end{bmatrix}.$$

Then the complex adjoint matrix of A is

$$\Psi_A = \begin{bmatrix} 2\mathbf{i} & 0 & 0 & 0-2 & 1+\mathbf{i} \\ 0 & 2 & -1 & -\mathbf{i} & 0 & 0 \\ 0 & 1-\mathbf{i} & 1 & -1 & 0 & 0 \\ 0 & 2 & -1+\mathbf{i} & -2\mathbf{i} & 0 & 0 \\ -\mathbf{i} & 0 & 0 & 0 & 2 & -1 \\ 1 & 0 & 0 & 0 & 1+\mathbf{i} & 1 \end{bmatrix}$$

Since the complex adjoint matrix Ψ_A is diagonalizable, from Theorem 1.12, the matrix A is diagonalizable. The standard right eigenvalues of A are $1, 1 + \mathbf{i}$ and $1 + \mathbf{i}$. Hence from Theorem 1.32, there is an invertible quaternionic matrix Y such that

$$Y = \begin{bmatrix} 1 + \mathbf{i} & \mathbf{i} & \mathbf{i} \\ \mathbf{j} & \mathbf{j} & 0 \\ -\mathbf{k} & -\mathbf{k} & -\mathbf{j} \end{bmatrix}.$$
132

Here $\Lambda_s(A) = \{1, 1 + \mathbf{i}\}$, where $1 + \mathbf{i}$ is with multiplicity 2. Then

$$Y^{-1}AY = \operatorname{diag}(1, 1 + \mathbf{i}, 1 + \mathbf{i}).$$

Suppose a perturbation matrix $\Delta A = \begin{bmatrix} 0 & 0 & 0 \\ 0 & 0 & \epsilon \\ \epsilon & \epsilon & \epsilon \end{bmatrix}$. Take $\epsilon = 10^{-10}$. Then
$$A + \Delta A = \begin{bmatrix} 2\mathbf{i} & -2\mathbf{j} & \mathbf{j} + \mathbf{k} \\ -\mathbf{k} & 2 & -1 + \epsilon \\ -\mathbf{j} + \epsilon & 1 - \mathbf{i} + \epsilon & 1 + \epsilon \end{bmatrix}.$$

Therefore $\|\Delta A\|_2 = \epsilon \sqrt{(1+\sqrt{2})}$ and $\Lambda_s(A+\Delta A) = \{1+1.0001\mathbf{i}, 1.0001+0.9999\mathbf{i}\}$, where $1+1.0001\mathbf{i}$ is with multiplicity 2. The condition number of the quaternionic matrix Y is $K_2(Y) = 10.2193$ and

$$\min_{\lambda_i \in \Lambda_{\mathbf{s}}(A)} \{ |\lambda_i - \mu| \} = \inf_{\eta_i \in \Lambda_{\mathbf{r}}(A), \ \xi \in \Lambda_{\mathbf{r}}(A + \Delta A)} \{ |\eta_i - \xi| \} = 0.0001.$$

Example 6.29. Consider a quaternionic matrix $A = \begin{bmatrix} 1 & -\mathbf{i} & -\mathbf{j} & \mathbf{k} \\ \mathbf{i} & 1 & -2\mathbf{k} & \mathbf{j} \\ \mathbf{j} & 2\mathbf{k} & 7 & -\mathbf{i} \\ -\mathbf{k} & -\mathbf{j} & \mathbf{i} & 1 \end{bmatrix}$. Then the

complex adjoint matrix of A is

$$\Psi_A = \begin{bmatrix} 1 & -\mathbf{i} & 0 & 0 & 0 & 0 & -1 & \mathbf{i} \\ \mathbf{i} & 1 & 0 & 0 & 0 & 0 & 2-\mathbf{i} & 1 \\ 0 & 0 & 7 & -\mathbf{i} & 1 & 2\mathbf{i} & 0 & 0 \\ 0 & 0 & \mathbf{i} & 1 & -\mathbf{i} & -1 & 0 & 0 \\ 0 & 0 & 1 & \mathbf{i} & 1 & \mathbf{i} & 0 & 0 \\ 0 & 0 & -2\mathbf{i} & -1 & -\mathbf{i} & 1 & 0 & 0 \\ -1 & 2\mathbf{i} & 0 & 0 & 0 & 0 & 7 & \mathbf{i} \\ -\mathbf{i} & 1 & 0 & 0 & 0 & 0 & -\mathbf{i} & 1 \end{bmatrix}$$

The set of standard right eigenvalues of A is

$$\Lambda_s(A) = \{-1, 1, 2, 8\}.$$

Since Ψ_A is Hermitian, from Theorem 1.12, A is Hermitian. Also from Theorem 1.29, the matrix A is unitarily diagonalizable, i.e., $U^H A U = \text{diag}(-1, 1, 2, 8)$, where $U \in M_n(\mathbb{H})$

is an unitary matrix. Since A is Hermitian, so $\Lambda_s(A) = \Lambda_r(A)$. Consider a perturbation matrix

$$\Delta A = \begin{bmatrix} 0 & 0 & 0 & 0 \\ 0 & 0 & 0 & \epsilon \\ 0 & 0 & \epsilon & \epsilon \\ \epsilon & \epsilon & \epsilon & \epsilon \end{bmatrix}. \text{ Take } \epsilon = 10^{-3}. \text{ Then } A + \Delta A = \begin{bmatrix} 1 & -\mathbf{i} & -\mathbf{j} & \mathbf{k} \\ \mathbf{i} & 1 & -2\mathbf{k} & \mathbf{j} + \epsilon \\ \mathbf{j} & 2\mathbf{k} & 7 + \epsilon & -\mathbf{i} + \epsilon \\ -\mathbf{k} + \epsilon & -\mathbf{j} + \epsilon & \mathbf{i} + \epsilon & 1 + \epsilon \end{bmatrix}.$$

Therefore $\|\Delta A\|_2 = 0.0024$ and

$$\Lambda_s(A + \Delta A) = \{1.0003 + 0.0001\mathbf{i}, 2.0005 + 0.0005\mathbf{i}, -0.9997 + 0.0003\mathbf{i}, 8.0009\}$$

Since the matrix A is unitarily diagonalizable, $K_2(Y) = 1$. Moreover, it is clear that A is also a central closed as well as a normal matrix.

Example 6.30. Define $t_0 := 2\mathbf{i}, t_1 := \mathbf{j}, t_2 := \mathbf{k}$. Then from [42], the quaternionic Vandermonde matrix is defined as

$$A := \begin{bmatrix} 1 & 1 & 1 \\ t_0 & t_1 & t_2 \\ t_0^2 & t_1^2 & t_2^2 \end{bmatrix} = \begin{bmatrix} 1 & 1 & 1 \\ 2\mathbf{i} & \mathbf{j} & \mathbf{k} \\ -4 & -1 & -1 \end{bmatrix}$$

The complex adjoint matrix of A is

$$\Psi_A = \begin{bmatrix} 1 & 1 & 1 & 0 & 0 & 0 \\ 2\mathbf{i} & 0 & 0 & 0 & 1 & \mathbf{i} \\ -4 & -1 & -1 & 0 & 0 & 0 \\ 0 & 0 & 0 & 1 & 1 & 1 \\ 0 & -1 & \mathbf{i} & -2\mathbf{i} & 0 & 0 \\ 0 & 0 & 0 & -4 & -1 & -1 \end{bmatrix}$$

The eigenvalues of Ψ_A are 0.8014+1.70007i, 0.8014-1.70007i, -0.4552+1.9952i, -0.4552- $1.9952\mathbf{i}, -0.3462 + 1.0469\mathbf{i}, -0.3462 - 1.0469\mathbf{i}$. Hence the complex adjoint matrix Ψ_A is diagonalizable. Then from Theorem 1.12, A is diagonalizable. The set of standard right eigenvalues of A is given as $\Lambda_s(A) = \{0.8014 + 1.70007\mathbf{i}, -0.4552 + 1.9952\mathbf{i}, -0.3462 + 1.9952\mathbf{i}, -0.346\mathbf{i}, -0.34$ 1.0469i}. Then there exists an invertible matrix $Y \in M_n(\mathbb{H})$ such that

$$Y^{-1}AY = \text{diag}(0.8014 + 1.70007\mathbf{i}, -0.4552 + 1.9952\mathbf{i}, -0.3462 + 1.0469\mathbf{i}).$$

Suppose a perturbation matrix

$$\Delta A = \begin{bmatrix} 0 & 0 & 0 \\ 0 & 0 & 0 \\ \epsilon & 0 & \epsilon \end{bmatrix}. \text{ Take } \epsilon = 10^{-3}. \text{ Then } A + \Delta A = \begin{bmatrix} 1 & 1 & 1 \\ 2\mathbf{i} & \mathbf{j} & \mathbf{k} \\ -4 + \epsilon & -1 & -1 + \epsilon \end{bmatrix}.$$

Therefore $\|\Delta A\|_2 = 1.4 \times 10^{-3}$ and

$$\Lambda_s(A + \Delta A) = \{0.8016 + 1.7009\mathbf{i}, -0.4549 + 1.9950\mathbf{i}, -0.3457 + 1.0470\mathbf{i}\}.$$

The condition number of the invertible matrix Y is $K_2(Y) \ge 1$. Also

$$\min_{\lambda_i \in \Lambda_{\mathbf{s}}(A)} \{ |\lambda_i - \mu| \} = \inf_{\substack{\eta_i \in \Lambda_{\mathbf{r}}(A)\\\xi \in \Lambda_{\mathbf{r}}(A + \Delta A)}} \{ |\eta_i - \xi| \} = 0.0002.$$

 $K_2(Y) \|\Delta A\|_2 \ge 0.0014.$

CHAPTER 7

CONCLUSIONS AND SCOPE FOR FUTURE WORK

We have derived localization theorems for the left and right eigenvalues of a quaternionic matrix. In particular, we have presented the Gerschgorin, Ostrowski, and Brauer type theorems for the left and right eigenvalues of a quaternionic matrix. Thereafter we have given a sufficient condition for the stability of a continuous-time quaternionic system. Furthermore, we have derived bounds/location of zeros of quaternionic polynomials.

We have developed various properties of a quaternionic matrix pencil. We have derived localization theorems for generalized right eigenvalues of a quaternionic matrix pencil and their applications.

The definitions of the left and right eigenvalues of quaternionic matrix polynomials have proposed. We have given a sufficient condition for the stability of a discrete-time quaternionic system. We have presented bounds for the absolute values of the left and right eigenvalues of quaternionic matrix polynomials and illustrated for the matrix *p*-norm, where $p = 1, 2, \infty$, and *F* (Frobenius).

Next, we have developed the concept of perturbation bounds for right eigenvalues/generalized right eigenvalues of a quaternionic matrix/quaternionic matrix pencil. In particular, Bauer-Fike type theorems for right eigenvalues/generalized right eigenvalues of a diagonalizable quaternionic matrix/diagonalizable quaternionic matrix pencil have derived. We have provided a relative perturbation bound for right eigenvalues of an invertible diagonalizable quaternionic matrix. We have discussed perturbation bounds for the zeros of quaternionic polynomials.

Finally, we have given the perturbation theory on matrices and polynomials over the skew field of quaternions. Specifically, the Bauer-Fike type theorems for right eigenvalues/generalized right eigenvalues of a diagonalizable quaternionic matrix/diagonalizable quaternionic matrix pencil have derived. In addition, perturbation bounds for right eigenvalues of a quaternionic matrix are discussed via block diagonal decomposition and Jordan canonical form of a quaternionic matrix. The location of right eigenvalues of a quaternionic

matrix and a sufficient condition for the stability of a perturbed quaternionic matrix have given. We have introduced perturbation bounds for zeros of quaternionic polynomials.

In the future work, we extend some of the localization theorems of Chapter 2 for the left and right eigenvalues of quaternionic block matrices. Another open question is investigate modified Gerschgorin and Ostrowski balls for the zeros of quaternionic polynomials by applying the localization theorems (proved in Chapter 2). One can think about the backward error for right eigenvalues and right eigenvector of structured quaternionic matrix polynomials. Bauer-Fike type theorem for quaternionic matrix polynomials is left for future investigation. As we have developed a general framework for quaternionic matrices and quaternionic matrix polynomials. Consequently we can be extended many results from the complex field to the skew field of quaternions.

APPENDIX-A

In this appendix, we find the powers of quaternionic companion matrices. Substituting t = 2 in Theorem 2.35, we have the following expressions:

$$C_{p_{l}}^{2} = \frac{m-2}{2} \begin{bmatrix} 0 & I \\ \hline C & D \end{bmatrix}, \text{ where } C = \begin{bmatrix} C_{p_{l}}(m, 1:2) \\ C_{p_{l}}^{2}(m, 1:2) \end{bmatrix} = \begin{bmatrix} -q_{0} & -q_{1} \\ q_{m-1}q_{0} & q_{m-1}q_{1} - q_{0} \end{bmatrix}$$

and

$$D = \begin{bmatrix} C_{p_l}(m, 3:m) \\ C_{p_l}^2(m, 3:m) \end{bmatrix} = \begin{bmatrix} -q_2 & -q_3 & \dots & -q_{m-1} \\ q_{m-1}q_2 - q_1 & q_{m-1}q_3 - q_1 & \dots & (q_{m-1})^2 - q_{m-2} \end{bmatrix}.$$

$$C_{\tilde{p}_l}^2 = \begin{bmatrix} m-2 \\ 2 \end{bmatrix} \begin{bmatrix} 0 & | I \\ \overline{C} & | D \end{bmatrix}, \text{ where } C = \begin{bmatrix} C_{\tilde{p}_l}(m, 1:2) \\ C_{\tilde{p}_l}^2(m, 1:2) \end{bmatrix} = \begin{bmatrix} -\overline{q_0} & -\overline{q_1} \\ \overline{q_{m-1}} & \overline{q_0} & \overline{q_{m-1}} & \overline{q_1} - \overline{q_0} \end{bmatrix}.$$

and

$$D = \begin{bmatrix} C_{\hat{p}_{l}}(m, 3:m) \\ C_{\hat{p}_{l}}^{2}(m, 3:m) \end{bmatrix} = \begin{bmatrix} -\overline{q_{2}} & -\overline{q_{3}} & \dots & -\overline{q_{m-1}} \\ \overline{q_{m-1}} \ \overline{q_{2}} - \overline{q_{1}} & \overline{q_{m-1}} \ \overline{q_{3}} - \overline{q_{1}} & \dots & (\overline{q_{m-1}})^{2} - \overline{q_{m-2}} \end{bmatrix}$$
$$C_{q_{l}}^{2} = \begin{bmatrix} 2 & m-2 \\ 0 & I \\ C & D \end{bmatrix}, \text{ where } C = \begin{bmatrix} -q_{0}^{-1} & -q_{0}^{-1}q_{m-1} \\ q_{0}^{-1}q_{1}q_{0}^{-1} & q_{0}^{-1}q_{1}q_{0}^{-1}q_{m-1} - q_{0}^{-1} \end{bmatrix}$$

•

and

$$D = \begin{bmatrix} -q_0^{-1}q_{m-2} & \dots & -q_0^{-1}q_1 \\ q_0^{-1}q_1q_0^{-1}q_{m-2} - q_0^{-1}q_{m-1} & \dots & (q_0^{-1}q_1)^2 - q_0^{-1}q_2 \end{bmatrix}.$$

$$C_{\tilde{q}_l}^2 = \begin{bmatrix} 2 & m-2 \\ 0 & I \\ \hline C & D \end{bmatrix}, \text{ where } C = \begin{bmatrix} -\overline{q_0^{-1}} & -\overline{q_0^{-1}q_{m-1}} \\ -\overline{q_0^{-1}q_1} & \overline{q_0^{-1}q_1} & \overline{q_0^{-1}q_1} \\ \hline q_0^{-1}q_1 & \overline{q_0^{-1}q_1} & \overline{q_0^{-1}q_{m-1}} - \overline{q_0^{-1}} \end{bmatrix}.$$

and

$$D = \begin{bmatrix} -\overline{q_0^{-1}q_{m-2}} & \dots & -\overline{q_0^{-1}q_1} \\ \\ \\ \hline q_0^{-1}q_1 & \overline{q_0^{-1}q_{m-2}} - \overline{q_0^{-1}q_{m-1}} & \dots & \left(\overline{q_0^{-1}q_1}\right)^2 - \overline{q_0^{-1}q_2} \end{bmatrix}.$$

By Theorem 2.37 for t = 2, we obtain the following expressions:

$$C_{p_r}^2 = \frac{2}{m-2} \left[\begin{array}{c|c} m-2 & 2 \\ \hline 0 & C \\ \hline I & D \end{array} \right],$$

where

$$C = \begin{bmatrix} C_{p_r}(1:2,m) & C_{p_r}^2(1:2,m) \end{bmatrix} = \begin{bmatrix} -q_0 & q_0 q_{m-1} \\ -q_1 & q_1 q_{m-1} - q_0 \end{bmatrix}$$

and

$$D = \begin{bmatrix} C_{p_r}(3:m,m) & C_{p_r}^2(3:m,m) \end{bmatrix} = \begin{bmatrix} -q_2 & q_2q_{m-1} - q_1 \\ -q_3 & q_3q_{m-1} - q_2 \\ \vdots & \vdots \\ -q_{m-1} & (q_{m-1})^2 - q_{m-2} \end{bmatrix}.$$

$$C_{\tilde{p}_{r}}^{2} = \frac{2}{m-2} \begin{bmatrix} 0 & C \\ \hline I & D \end{bmatrix}, \text{ where } C = \begin{bmatrix} -\overline{q_{0}} & \overline{q_{0}} & \overline{q_{m-1}} \\ -\overline{q_{1}} & \overline{q_{1}} & \overline{q_{m-1}} - \overline{q_{0}} \end{bmatrix} \text{ and}$$
$$D = \begin{bmatrix} -\overline{q_{2}} & \overline{q_{2}} & \overline{q_{m-1}} - \overline{q_{1}} \\ -\overline{q_{3}} & \overline{q_{3}} & \overline{q_{m-1}} - \overline{q_{2}} \\ \vdots & \vdots \\ -\overline{q_{m-1}} & (\overline{q_{m-1}})^{2} - \overline{q_{m-2}} \end{bmatrix}.$$

$$C_{q_r}^2 = \frac{2}{m-2} \begin{bmatrix} 0 & | & C \\ \hline I & | & D \end{bmatrix}, \text{ where } C = \begin{bmatrix} -q_0^{-1} & q_0^{-1} q_1 q_0^{-1} \\ -q_{m-1} q_0^{-1} & q_{m-1} q_0^{-1} q_1 q_0^{-1} \\ -q_{m-1} q_0^{-1} & q_{m-2} q_0^{-1} q_1 q_0^{-1} - q_{m-1} q_0^{-1} \\ \vdots & \vdots \\ -q_1 q_0^{-1} & (q_1 q_0^{-1})^2 - q_2 q_0^{-1} \end{bmatrix}.$$

$$140$$

$$C_{\tilde{q}_{r}}^{2} = {\begin{array}{*{20}c} & & & \\ 2 & & \\ m-2 & \end{array} \left[\begin{array}{*{20}c} & & & \\ 0 & & C \\ \hline & & & \\ \end{array} \right]}, \text{ where } C = \begin{bmatrix} & -\overline{q_{0}^{-1}} & & \overline{q_{0}^{-1}} & \overline{q_{1}q_{0}^{-1}} \\ & & \\ -\overline{q_{m-1}q_{0}^{-1}} & & \overline{q_{1}q_{0}^{-1}} & -\overline{q_{0}^{-1}} \\ \end{bmatrix}} \text{ and } D = \begin{bmatrix} & -\overline{q_{m-2}q_{0}^{-1}} & & \overline{q_{m-2}q_{0}^{-1}} & \overline{q_{1}q_{0}^{-1}} & -\overline{q_{m-1}q_{0}^{-1}} \\ & \vdots & & \vdots & \\ & -\overline{q_{1}q_{0}^{-1}} & & \left(\overline{q_{1}q_{0}^{-1}}\right)^{2} - \overline{q_{2}q_{0}^{-1}} \end{bmatrix}}.$$

APPENDIX-B

In this appendix, we find the powers of the quaternionic block companion matrices C_V and C_S . Hence from Theorem 5.31 for t = 2, we have the following expressions:

$$C_{V}^{2} = \frac{n(m-2)}{2n} \left[\frac{0}{C} \mid \frac{I}{D} \right], \text{ where } C = \begin{bmatrix} C_{V}[m, 1:t] \\ C_{V}^{2}[m, 1:t] \end{bmatrix} = \begin{bmatrix} -V_{0} & -V_{1} \\ V_{m-1}V_{0} & V_{m-1}V_{1} - V_{0} \end{bmatrix}$$

and
$$D = \begin{bmatrix} C_{V}[m, t+1:m] \\ C_{V}^{2}[m, t+1:m] \end{bmatrix} = \begin{bmatrix} -V_{2} & -V_{2} & \dots & V_{m-1} \\ V_{m-1}V_{2} - V_{1} & V_{m-1}V_{3} - V_{2} & \dots & V_{m-1}^{2} \\ \end{bmatrix}.$$

Iso we have

Also we have

$$C_{S}^{2} = \frac{n(m-2)}{2n} \left[\frac{0}{C} \mid \frac{I}{D} \right], \text{ where } C = \begin{bmatrix} C_{S}[m, 1:t] \\ C_{S}^{2}[m, 1:t] \end{bmatrix} = \begin{bmatrix} -S_{m} & -S_{m-1} \\ S_{1}S_{m} & S_{1}S_{m-1} - S_{m} \end{bmatrix}$$

and
$$D = \begin{bmatrix} C_{S}[m, t+1:m] \\ C_{S}^{2}[m, t+1:m] \end{bmatrix} = \begin{bmatrix} -S_{m-2} & -S_{m-3} & \dots & -S_{1} \\ S_{1}S_{m-2} - S_{m-1} & S_{1}S_{m-3} - S_{m-2} & \dots & S_{1}^{2} - S_{2} \end{bmatrix}.$$

ow the powers of C_{M} and C_{G} are given as follows

Now the powers of C_V and C_S are given as follows.

$$C_V^2 := \begin{bmatrix} 0 & 0 & I_n & 0 \\ \vdots & \vdots & \ddots & \vdots \\ 0 & 0 & 0 & \dots & I_n \\ -V_0 & -V_1 & -V_2 & \dots & -V_{m-1} \\ U_0 & U_1 & U_2 & \dots & U_{m-1} \end{bmatrix} \text{ and } C_S^2 := \begin{bmatrix} 0 & 0 & I_n & 0 \\ \vdots & \vdots & \ddots & \vdots \\ 0 & 0 & 0 & \dots & I_n \\ -S_m & -S_{m-1} & -S_{m-2} & \dots & -S_1 \\ \mathcal{L}_m & \mathcal{L}_{m-1} & \mathcal{L}_{m-2} & \dots & \mathcal{L}_1 \end{bmatrix},$$

where $U_j = V_{m-1}V_j - V_{j-1}$ and $\mathcal{L}_{j+1} = S_1S_{j+1} - S_{j+2}$, $j = 0, 1, \dots, m-1$ with $V_{-1} =$ $S_{m+1} = 0$. Define

$$U := \begin{bmatrix} U_0 & U_1 & \dots & U_{m-1} \end{bmatrix}, \ \mathcal{L} := \begin{bmatrix} \mathcal{L}_m & \mathcal{L}_{m-1} & \dots & \mathcal{L}_1 \end{bmatrix}.$$

BIBLIOGRAPHY

- Adler S.L., (1995) Quaternionic Quantum Mechanics and Quantum Fields, Oxford Press, New York.
- [2] Aslaksen H. (1996), Quaternionic determinants, Math. Intell., 18, 57–65.
- [3] Baker A. (1999), Right eigenvalues for quaternionic matrices: a topological approach, Linear Algebra Appl., 286, 303–309.
- [4] Bini D.A., Noferini V., Sharify M. (2013), Locating the eigenvalues of matrix polynomials, SIAM J. Matrix Anal. Appl., 34(4), 1708–1727.
- [5] Brenner J.L. (1951), Matrices of quaternions, Pacific. J. Math., 1, 329–335.
- Bunse-Gerstner A., Byers R., Mehrmann V. (1989), A quaternion QR algorithm, Numer. Math., 55, 83–95.
- [7] Chu E.K.-W. (2003), Perturbation of eigenvalues for matrix polynomials via the Bauer-Fike theorems, SIAM J. Matrix Anal. Appl., 25(2), 551–573.
- [8] Conway J.H., Smith D.A., (2002) On Quaternions and Octonions: Their Geometry, Arithmetic, and Symmetry, A K Peters, Natick.
- [9] Farenick D.R., Pidkowich B.A.F. (2003), The spectral theorem in quaternions, Linear Algebra Appl., 371, 75–102.
- [10] Gohberg I., Lancaster P., Rodman L., (1982) Matrix Polynomials, Academic Press, New York.
- [11] Golub G.H., Van Loan C.F., (1996) Matrix Computations, Third ed. The Johns Hopkins University Press, Baltimore, MD.
- [12] Gordon B., Motzkin T.S. (1965), On the zeros of polynomials over division rings, Trans. Amer. Math. Soc., 116, 218–226.
- [13] Hankins T.L., (1980) Sir William Rowan Hamilton, The Johns Hopkins University Press, Baltimore.

- [14] Hassan M.A., (2004) Inequalities and bounds for the zeros of polynomials using Perron-Frobenius and Gerschgorin theories, Proceedings of American Control Conference, Boston, Massachusetts, 3, 2745–2750.
- [15] Higham N.J., Tisseur F. (2003), Bounds for eigenvalues of matrix polynomials, Linear Algebra Appl., 358, 5–22.
- [16] Horn R.A., Johnson C.R., (1985) Matrix Analysis, Cambridge University Press, Cambridge.
- [17] Horn R.A., Zhang F. (2012), A generalization of the complex Autonne-Takagi factorization to quaternion matrices, Linear Multilinear Algebra, 60, 1239–1244.
- [18] Huang L., So W. (2001), On left eigenvalues of a quaternionic matrix, Linear Algebra Appl., 323, 105–116.
- [19] Janovská D., Opfer G. (2010), A note on the computation of all zeros of simple quaternionic polynomials, SIAM J. Numer. Anal., 48(1), 244–256.
- [20] Janovská D., Opfer G. (2010), The classification and the computation of the zeros of quaternionic, two-sided polynomials, Numer. Math., 115, 81–100.
- [21] Jiang T. (2005), Algebraic methods for diagonalization of a quaternion matrix in quaternionic quantum theory, J. Math. Phys., 46, 052106.
- [22] Junliang W., Limin Z., Xiangping C., Shengjie L. (2008), The estimation of eigenvalues of sum, difference, and tensor product of matrices over quaternion division algebra, Linear Algebra Appl., 428, 3023–3033.
- [23] Kalantari B. (2013), Algorithms for quaternion polynomial root-finding, J. complexity, 29, 302–322.
- [24] Kamberov G., Norman P., Pedit F., Pinkall U., (2002) Quaternions, Spinors, and Surfaces, Contemporary Mathematics, Amer. Math. Soc., 299, Province.
- [25] Kostic V., Cvetkovic L.J., Varga R.S. (2009), Gersgorin-type localizations of generalized eigenvalues, Numer. Linear Algebr. Appl., 16(11-12), 883–898.
- [26] Kuipers J.B., (2002) Quaternions and Rotation Sequences: A Primer with Applications to Orbits, Aerospace, and Virtual Reality, Princeton University Press, Princeton.
- [27] Lee H.-C. (1949), Eigenvalues and canonical forms of matrices with quaternion coefficients, Proc. Roy. Irish Acad. Sect., 52A, 253–260.

- [28] Leo S.D., Ducati G. (2001), Quaternionic differential operators, J. Math. Phys., 42, 2236–2265.
- [29] Leo S.D., Ducati G. (2003), Solving simple quaternionic differential equation, J. Math. Phys., 43, 2224–2233.
- [30] Leo S.D., Ducati G., Leonardi V. (2006), Zeros of unilateral quaternionic polynomials, Electron. J. Linear Algebra, 15, 297–313.
- [31] Leo S.D., Scolarici G. (2000), Right eigenvalue equation in quaternionic quantum mechanics, J. Phys. A, 33, 2971–2995.
- [32] Leo S.D., Scolarici G., Solombrino L. (2002), Quaternionic eigenvalue problem, J. Math. Phys., 43(11), 5815–5829.
- [33] Liping H. (1996), The matrix equation AXB-GXD =E over the quaternion field, Linear Algebra Appl., 234, 197–208.
- [34] Loring T.A. (2012), Factorization of matrices of quaternions, Expo. Math., 30(3), 250–267.
- [35] Mehrmann V., Xu H. (1996), An analysis of the pole plecement problem I. The single-input case, Electron. Trans. Numer. Anal., 4, 89–105.
- [36] Mehrmann V., Xu H. (1997), An analysis of the pole placement problem II. The multi-input case, Electron. Trans. Numer. Anal., 5, 77-97.
- [37] Melman A. (2013), A single oval of Cassini for the zeros of a polynomial, Linear Multilinear Algebra, 61(2) ,183–195.
- [38] Melman A. (2013), Generalization and variations of Pellet's theorem for matrix polynomials, Linear Algebra Appl., 439, 1550–1567.
- [39] Mukundan R., (2002) Quaternions: From classical mechanics to computer graphics, and beyond, In Proceedings of the 7th Asian Technology Conference in Mathematics.
- [40] Nebe G. (1998), Finite quaternionic matrix groups, Represent. Theory, 2, 106–223.
- [41] Niven I. (1941), Equations in quaternions, Amer. Math. Monthly, 48, 654–661.
- [42] Opfer G. (2009), Polynomials and Vandermonde matrices over the field of quaternions, Electron. Trans. Numer. Anal., 36, 9–16.
- [43] Pereira R., (2006) Quaternionic Polynomials and Behavioral Systems, PhD Thesis, University of Aveiro.
- [44] Pereira R., Rocha P. (2008), On the determinant of quaternionic polynomial matrices and its application to system stability, Math. Methods Appl. Sci., 31, 99–122.

- [45] Pogorui A., Shapiro M. (2004), On the structure of the set of zeros of quaternionic polynomials, Complex Var. and Elliptic Funct., 49, 379–389.
- [46] Rodman L. (2007), Canonical forms for mixed symmetric-skewsymmetric quaternion matrix pencils, Linear Algebra Appl., 424, 184–221.
- [47] Rodman L. (2008), Pairs of hermitian and skew hermitian quaternionic matrices: canonical forms and their applications, Linear Algebra Appl., 429, 981–1019.
- [48] Rodman L. (2012), Stability of invariant subspaces of quaternion matrices, Complex Anal. Oper. Theory, 6(5), 1069–1119.
- [49] Rodman L., (2014) Topics in Quaternion Linear Algebra, Princeton University Press, Princeton (NJ).
- [50] Serôdio R., Pereira E., Vitória J. (2001), Computing the zeros of quaternion polynomials, Comput. Math. Appl., 42, 1229–1237.
- [51] Shi X., Wei Y. (2012), A sharp version of Bauer-Fike's theorem, J. Comput. Appl. Math., 236, 3218–3227.
- [52] So W. (2005), Quaternionic left eigenvalue problem, Southeast Asian Bull. Math., 29, 555–565.
- [53] Study E. (1920), Zur Theorie der linearen Gleichungen, Acta Math., 42, 1–61.
- [54] Tisseur F., Higham N.J. (2001), Structured pseudospectra for polynomial eigenvalue problems with applications, SIAM J. Matrix Anal. Appl., 23(1), 187–208.
- [55] Took C.C., Mandic D.P. (2011), Augmented second-order statistics of quaternion random signals, Signal Processing, 91, 214–224.
- [56] Took C.C., Mandic D.P., Zhang F. (2011), On the unitary diagonalisation of a special class of quaternion matrices, Appl. Math. Lett., 24, 1806–1809.
- [57] Varga R.S., (2004) Geršgorin and His Circles, Springer, Berlin.
- [58] Ward J.P., (1997) Quaternions and Cayley Numbers, Mathematics and Its Applications, Kluwer Academic Publishers, Dordrecht, the Netherlands, 403.
- [59] Wolff L.A. (1936), Similarity of matrices in which the elements are real quaternions, Bull. Amer. Math. Soc., 42, 737–743.
- [60] Wu J.L. (2008), Distribution and estimation for eigenvalues of real quaternion matrices, Comput. Math. Appl., 55, 1998–2004.
- [61] Zhang F. (1997), Quaternions and matrices of quaternions, Linear Algebra Appl., 251, 21–57.

- [62] Zhang F. (2007), Geršgorin type theorems for quaternionic matrices, Linear Algebra Appl., 424, 139–155.
- [63] Zou L., Jiang Y., Wu J. (2012), Location for the right eigenvalues of quaternion matrices, J. Appl. Math. Comput., 38, 71–83.